BIO-GENERATION OF POLY/OLIGOSACCHARIDES AND THEIR FERULOYLATED DERIVATIVES FROM COCOA BEAN SHELLS FOR POTENTIAL USE AS SWEETENER PREBIOTICS

Amalie Younes

Department of Food Science and Agricultural Chemistry

McGill University, Montréal

December 2022

A thesis submitted to McGill University in partial fulfillment of the requirements of the degree of Master of Science

© Amalie Younes, 2022

Suggested short title

BIO-GENERATION AND FERULOYLATION OF COCOA BEANS SHELL POLY/OLIGOSACCHARIDES FOR POTENTIAL USE AS SWEETENER PREBIOTICS

ABSTRACT

Cocoa bean shell, the outermost layer of the cocoa bean, is a heavily accumulated biomass product of the chocolate industry. Efforts to valorize this biomass are of high interest, in terms of their nutritional benefits, in agricultural, feedstuff, bioremediation, and food applications. Based on the proximate compositional analysis, cocoa bean shells from two varieties (Nacional (NAC) and CCN) were characterised. Overall, cocoa bean shells were found to be a great source of carbohydrates, specifically dietary fiber (50.1- 61.4% w/w), protein (12.7-19.7% w/w), ash, and polyphenols, namely quercetin, epicatechin, and catechin. There were noticeable differences in the CCN variety where the degree of processing accounted for a variation in the chemical composition. For example, is the protein content has shown a noticeable increase (p < 0.05) from its fermented (13%, CCN) to dried (18.9%, CCN) states, followed by a decrease in its roasted state (12.8%, CCN). Similarly, the fat content increases (p < 0.05) with the processing of cocoa beans from fermentation to drying and roasting, which may be attributed to subjection of elevated temperatures in the drying and roasting processes.

The cell wall material of cocoa bean shells was isolated and characterised as being primarily composed of pectic polysaccharides (80.6-86%), as indicated by the presence of uronic acids (35.1-50.5%), galactose (22.7-29.2%), arabinose (13.6-17.2%), and rhamnose (5.1-6.9%) residues. Additionally, the extraction of cell wall polysaccharides from the cell wall material of cocoa bean shells was explored. The alkaline extractions (0.5 and 4M potassium hydroxide) were successful in the recovery of polysaccharide predominantly enriched with pectic polysaccharide regions (80.6-86% w/w polysaccharide extract), rhamnogalacturonan and arabinogalactan. Increasing the alkali concentration accounts in a shift in the recovery from high to low molecular weight fractions, due to a higher degree of defragmentation; where the 4M extract accounted for two MW main populations (24.0, and 6.3 kDa – NAC; 11.0 and 3.2 kDa - CCN) compared to the three populations found in the less concentrated extracts (3.7, 12.0, and 130 kDa – NAC; 4.5, 9.1, and 32.5 kDa - CCN).

Enzymatic hydrolysis of cocoa bean shell cell wall polysaccharides by arabinase, galactanase and five multi-enzymatic products was investigated for the bio-generation of oligosaccharides. The multi-enzymatic products expressed varied levels of polysaccharide-degrading activities. Viscozyme® L (76.4%, NAC) and DepolTM 670L (20.72%, CCN) attained the highest hydrolysis

extents, while also generating a diverse complex of oligosaccharides with degrees of polymerization between 3 and 10. Most of the oligosaccharides (50.99 - 99.51%) generated by multi-enzymatic biocatalysts had a degree of polymerization between 2 and 5. The prebiotic activities were assessed using cell wall polysaccharides (NAC, CCN) as well as NAC- and CCN-oligosaccharides bio-generated by Viscozyme® L and DepolTM 670L, respectively. For the promotion of the growth of selected probiotic strains (*Lactobacillus rhamnosus GG* and *Bifidobacterium* longum), cell wall polysaccharides and bio-generated oligosaccharides performed similarly or more than inulin and rhamnogalacturonan as it was shown by the prebiotic activity scores. The short chain fatty acid profiles were mostly characterized by high amounts of lactic and acetic acid. However, oligosaccharides also contributed to a significant production of butyric and propionic acids, possibly attributed to their smaller degree of polymerization.

Enzymatic esterification of generated oligo- and polysaccharides from NAC and CCN cocoa bean shells with ferulic acid was investigated. Feruloyl esterases, from *Humicola insolens* and rumen microorganism, were immobilized on iminodiacetic acid-copper modified epoxy-polyacrylic supports. The specific esterifying activity of free and immobilized feruloyl esterases was assessed on raffinose in surfactant-less microemulsions, made of n-hexane/butanone/water (49.4/44.6/6, v/v/v). Although high immobilization yields were attained, the free feruloyl esterases were found to be most active. For the esterification of cocoa bean shell oligo- and polysaccharides, free feruloyl esterases were used. Feruloyl esterase from rumen microorganism led to the highest feruloylation efficiencies with polysaccharides (56.9-60.1%), while the feruloyl esterase from *H. insolens* resulted in the highest feruloylation efficiencies (30.8-50.5%) with oligosaccharides. Based on DPPH, ORAC, and ABTS assays, feruloylated polysaccharides prompted significantly notable increases in the scavenging and antioxidant activity compared to their native forms as well as to ferulic and ascorbic acid.

The extraction techniques explored for the recovery of oligo- and polysaccharides from CCN variety of cocoa bean shells were scaled-up in a pilot plant facility, to compare their efficiencies and extracts recovered, in terms of their monosaccharide profile and molecular weight distribution. One process was based on the isolation of cell wall polysaccharides from cocoa bean shells using 0.5M alkali concentration, while the second process was based on the recovery of bio-generated oligosaccharides using commercially available DepolTM 670L. As expected, the recovery of pilot-scale polysaccharides (3.69%) was lower than those recovered at the lab-scale (29.1%). The

monosaccharide profile of pilot-scale polysaccharides were similar to those recovered at the labscale with the exception of a higher proportion of total neutral sugars, namely galactose (17.4%) and glucose (13.9%). Based on the degree of polymerization, polysaccharides ranged between DP 2-5 (85.2% mg/L), DP 10-15 (10.8% mg/L) and DP > 40 (4.0% mg/L); while oligosaccharides ranged between DP 6-9 (57.6% mg/L), DP 2-5 (39.25 % mg/L), and DP 16-40 (3.2% mg/L). In terms of the MW distribution, both extracts showed four main populations at 0.87, 1.1, 1.6, and 3.3 kDa for oligosaccharides and at 0.83, 2.1, 320, and 2400 kDa for polysaccharides.

Selected chocolate-based formulations were enriched with pilot-scale polysaccharides for the assessment of their consumer acceptability and purchase intent, where a sensory panel was held to compare chocolate formulations with and without polysaccharide enrichment. The design was evaluated by 60 untrained panelists on a 9-point hedonic scale for mean quantitative descriptive scores and mean liking based scores. The relationships between the consumers liking scores and the perceived complexity were assessed to understand the degree of complexity inflicted upon consumers. Overall, consumers generally preferred the enriched chocolate formulations.

RÉSUMÉ

La coque de la fève de cacao, la couche la plus externe de la fève de cacao, est une biomasse fortement accumulée par l'industrie du chocolat. Les efforts pour valoriser cette biomasse sont d'un grand intérêt, en raison de leurs avantages nutritionnels, dans les applications agricoles, alimentaires, de biorémédiation et d'alimentation animale. Sur la base de l'analyse de la composition proximale, les coques de fèves de cacao de deux variétés (Nacional (NAC) et CCN) ont été caractérisées. Dans l'ensemble, les coques de fèves de cacao se sont révélées être une excellente source de glucides, en particulier de fibres alimentaires (50,1 à 61,4 % p/p), de protéines (12,7 à 19,7 % p/p), de cendres et de polyphénols, à savoir la quercétine, l'épicatéchine et la catéchine. Des différences notables ont été constatées dans la variété CCN où le degré de transformation explique une variation de la composition chimique. Par exemple, la teneur en protéines a montré une augmentation notable (p < 0,05) de l'état fermenté (13%, CCN) à l'état séché (18,9%, CCN), suivie d'une diminution à l'état de torréfaction (12,8%, CCN). De même, la teneur en graisse augmente (p < 0,05) avec le traitement des fèves de cacao, de la fermentation au séchage et à la torréfaction, ce qui peut être attribué à la soumission à des températures élevées dans les processus de séchage et de torréfaction.

La paroi cellulaire des coques de fèves de cacao a été isolée et caractérisée comme étant principalement composée de polysaccharides pectiques (80,6-86 %), comme l'indique la présence d'acides uroniques (35,1-50,5 %), du galactose (22,7-29,2 %), d'arabinose (13,6-17,2 %) et de rhamnose (5,1-6,9 %). Les extractions alcalines (hydroxyde de potassium 0,5 et 4M) ont permis de récupérer des polysaccharides principalement enrichis en régions pectiques (80,6-86 % p/p), en rhamnogalacturonan et en arabinogalactan. L'augmentation de la concentration d'alcali entraîne un déplacement de la récupération des fractions de poids moléculaire élevé vers les fractions de poids moléculaire faible, en raison d'un degré plus élevé de défragmentation ; où l'extrait 4M représentait deux populations principales de MW (24,0 et 6,3 kDa - NAC ; 11,0 et 3,2 kDa - CCN) par rapport aux trois populations trouvées dans les extraits moins concentrés (3,7, 12,0 et 130 kDa - NAC ; 4,5, 9,1 et 32,5 kDa - CCN).

L'hydrolyse enzymatique des polysaccharides de la paroi cellulaire de la coque de la fève de cacao par l'arabinase, la galactanase et cinq produits multi-enzymatiques a été étudiée pour la biogénération d'oligosaccharides. Les produits multi-enzymatiques ont exprimé des niveaux variés d'activités de dégradation des polysaccharides. Viscozyme® L (76,4 %, NAC) et Depol[™] 670L (20,72 %, CCN) ont atteint les plus hauts degrés d'hydrolyse, tout en générant un complexe diversifié d'oligosaccharides avec des degrés de polymérisation entre 3 et 10. La plupart des oligosaccharides (50,99 - 99,51%) générés par les biocatalyseurs multi-enzymatiques avaient un degré de polymérisation entre 2 et 5. Les activités prébiotiques ont été évaluées en utilisant des polysaccharides de la paroi cellulaire (NAC, CCN) ainsi que des oligosaccharides NAC- et CCN-générés par Viscozyme® L et Depol[™] 670L, respectivement. Pour la promotion de la croissance des souches probiotiques sélectionnées (*Lactobacillus rhamnosus* GG et *Bifidobacterium longum*), les polysaccharides de la paroi cellulaire et les oligosaccharides biogénérés ont eu des performances similaires ou supérieures à celles de l'inuline et du rhamnogalacturonane, comme le montrent les scores d'activité prébiotique. Les profils des acides gras à chaîne courte étaient principalement caractérisés par des quantités élevées d'acides lactique et acétique. Cependant, les oligosaccharides ont également contribué à une production significative d'acides butyrique et propionique, probablement attribuée à leur plus faible degré de polymérisation.

L'estérification enzymatique d'oligo- et polysaccharides bio-générés à partir de coques de fèves de cacao NAC et CCN avec de l'aide férulique a été étudiée. Des féruloyl estérases, provenant de Humicola insolens et de microorganismes du rumen, ont été immobilisées sur des supports époxypolyacryliques modifiés par de l'acide iminodiacétique et du cuivre. L'activité estérifiante spécifique des féruloyl estérases libres et immobilisées a été évaluée sur le raffinose dans des microémulsions sans tensioactif, composées de n-hexane/butanone/eau (49,4/44,6/6, v/v/v). Bien que des rendements élevés d'immobilisation aient été atteints, les féruloyl estérases libres se sont avérées les plus actives. Pour l'estérification des oligo- et polysaccharides de la coque de fèves de cacao, des féruloyl estérases libres ont été utilisées. La féruloyl estérase provenant d'un microorganisme du rumen a conduit aux efficacités de féruloylation les plus élevées avec les polysaccharides (56,9-60,1%), tandis que la féruloyl estérase provenant de H. insolens a conduit aux efficacités de féruloylation les plus élevées (30,8-50,5%) avec les oligosaccharides. Les capacités antioxydantes des polysaccharides féruloylés ont été évaluées. D'après les tests DPPH, ORAC et ABTS, les polysaccharides féruloylés ont présenté des augmentations significatives et notables de l'activité antioxydante et de piégeage par rapport à leurs formes natives ainsi qu'à l'acide férulique et ascorbique.

Les techniques d'extraction explorées pour la récupération des oligo- et polysaccharides de la variété CCN de coques de fèves de cacao ont été mises à l'échelle dans une installation pilote, afin de comparer leurs efficacités et les extraits récupérés, en termes de profil de monosaccharides et de distribution de poids moléculaire. Un procédé était basé sur l'isolement des polysaccharides de la paroi cellulaire des coques de fèves de cacao en utilisant une concentration alcaline de 0,5 M, tandis que le second procédé était basé sur la récupération des oligosaccharides biogénérés en utilisant le DepolTM 670L disponible dans le commerce. Comme prévu, la récupération des polysaccharides à l'échelle pilote (3,69 %) était inférieure à ceux récupérés à l'échelle du laboratoire (29,1 %). Le profil des monosaccharides des polysaccharides de l'échelle pilote était similaire à ceux récupérés à l'échelle du laboratoire, à l'exception d'une proportion plus élevée de sucres neutres totaux, à savoir le galactose (17,4 %) et le glucose (13,9 %). Selon le degré de polymérisation, les polysaccharides variaient entre DP 2-5 (85,2 % mg/L), DP 10-15 (10,8 % mg/L) et DP > 40 (4,0 % mg/L) ; tandis que les oligosaccharides variaient entre DP 6-9 (57,6 % mg/L), DP 2-5 (39,25 % mg/L) et DP 16-40 (3,2 % mg/L). En termes de distribution des MW, les deux extraits ont montré quatre populations principales à 0,87, 1,1, 1,6, et 3,3 kDa pour les oligosaccharides et à 0,83, 2,1, 320, et 2400 kDa pour les polysaccharides.

Des formulations à base de chocolat sélectionnées ont été enrichies avec des polysaccharides à l'échelle pilote pour l'évaluation de leur acceptabilité par les consommateurs et de leur intention d'achat. Un panel sensoriel a été organisé pour comparer les formulations de chocolat avec et sans enrichissement en polysaccharides. Les formulations ont été évaluées par 60 panélistes non formés utilisant une échelle hédonique à 5 points pour les scores descriptifs quantitatifs moyens et les scores moyens basés sur le goût. Les relations entre les scores d'appréciation des consommateurs et la complexité perçue ont été évaluées afin de comprendre le degré de complexité infligé aux consommateurs. Dans l'ensemble, les consommateurs ont généralement préféré les formulations de chocolat enrichies.

STATEMENT FROM THE THESIS OFFICE

According to the regulations of the Faculty of Graduate Studies and Research of McGill University, Guidelines for Thesis Preparation include:

Candidates have the option of including, as part of the thesis, the text of one or more papers submitted, or to be submitted, for publication, or the clearly duplicated text of one or more published papers. These texts must conform to the "Guidelines for Thesis Preparation" and must be bound together as an integral part of the thesis.

The thesis must be more than a collection of manuscripts. All components must be integrated into a cohesive unit with a logical progression from one chapter to the next. In order to ensure that the thesis has continuity, connecting texts that provide logical bridges between the different papers are mandatory.

The thesis must conform to all other requirements of the "Guidelines for Thesis Preparation" in addition to the manuscripts.

As manuscripts for publication are frequently very concise documents, where appropriate, additional material must be provided in sufficient detail to allow a clear and precise judgement to be made of the importance and originality of the research reported in the thesis.

In general, when co-authored papers are included in a thesis, the candidate must have made a substantial contribution to all papers included in the thesis. In addition, the candidate is required to make an explicit statement in the thesis as to who contributed to such work and to what extent. This statement should appear in a single section entitled "Contribution of Authors" as a preface of the thesis.

When previously published copyright material is presented in a thesis, the candidate must obtain, if necessary, signed waivers from the co-authors and publishers and submit these to the Thesis Office with the final deposition.

PREFACE AND CONTRIBUTION OF AUTHORS

This thesis includes six chapters.

Chapter I provides a general introduction to the valorization of cocoa bean shells, the extraction of their cell wall polysaccharides as well as their current and potential food applications as a functional ingredient.

Chapter II comprises a literature review of the studies relevant to cocoa bean shells' cell wall polysaccharide composition and characterization their methods of cell wall poly- and oligosaccharide extraction, especially chemical and enzymatic approaches.

Chapter III discusses the results pertaining to the extraction, characterization of the cocoa bean shell cell wall polysaccharides as well as the assessment of their prebiotic activity.

Chapter IV discusses the results pertaining to the enzymatic generation of cell wall oligosaccharides via multi-enzymatic products, the assessment of their prebiotic activity and the production of short chain fatty acid metabolites. This chapter will also discuss the results pertaining to the feruloylation of oligo- and polysaccharides using feruloyl esterases as well as the assessment of their antioxidant activities using selected antioxidant assays.

Chapter V discusses the pilot-plant extraction of cocoa bean shell carbohydrate and hydrolyzed carbohydrate extracts through alkaline isolation and multi-enzymatic hydrolysis, respectively. This chapter also discusses the results pertaining to chocolate-based formulation trials along with their sensory evaluation for consumer acceptability and purchase intentions.

Finally, Chapter VI covers an overall summary of the current research results

Amalie Younes was responsible for the experimental work and the preparation of the manuscripts for the publications and thesis.

Dr. Mingqin Li, the second author of Chapter II, reviewed the manuscript prior to submission.

Dr. Lan Liu, the third author of Chapter III, contributed to the research and experimental work related to the characterization of the phenolic compounds by mass spectrometry.

Dr. Amanda Waglay, the second author of Chapter V, contributed towards the facilitation of the pilot-plant scale-up trial related to extraction of cocoa bean shell carbohydrate extracts as well as reviewed the manuscript prior to submission.

Dr. Asma Mdimagh, the third author of Chapter V, contributed towards the statistical analysis of the sensory evaluation datasets of chocolate-based formulations.

Dr. Salwa Karboune, the supervisor of the current M.Sc. Research project, guided the entire research framework and reviewed all the presented chapters in this thesis prior to the submission.

RESEARCH CONTRIBUTIONS

PUBLICATIONS

- Younes, A., Li, M., & Karboune, S. (2022). Cocoa bean shells: a review into the chemical profile, the bioactivity, and the biotransformation to enhance their potential applications in foods. Critical reviews in food science and nutrition, 1–25. https://doi.org/10.1080/10408398.2022.2065659
- 2 Younes, A., Karboune, S., Liu, L., Andreani, E.S., & Dahman, S. Extraction and Characterization of Cocoa Bean Shell Cell Wall Polysaccharides. *Polymers* 2023, 15, 745. <u>https://doi.org/10.3390/polym15030745</u>
- Younes, A., & Karboune, S. (2023). Enzymatic generation of oligosaccharides from cocoa bean shells polysaccharides using multi-enzymatic products and their feruloylation (Publication under review)
- Younes, A., Razmjooyhassankhani, N., Waglay, A., Mdimagh, A., & Karboune, S. (2023).
 Pilot plant extraction of cocoa bean shell oligo- and polysaccharides and its incorporation into chocolate-based formulations as a prebiotic sugar alternative (Publication under review).

ACKNOWLEDGMENTS

Firstly, I would like to express my sincerest gratitude to my supervisor, Dr. Salwa Karboune, for her constant support, guidance, dedication throughout this study.

I would like to extend my appreciation to Dr. Amanda Waglay for her support and guidance during the research work related to the pilot scale-up process for the extraction of cocoa bean shell polysaccharides, which was completed at the Food Research and Development Center in St-Hyacinthe, Qc.

I am grateful to all my dearest friends and colleagues, Amal Sahyoun, Elham Chidar, Kelly Light, Lichen Yang, Mile Shao, Muriel Wongmin, Dr. Najla Ben Akcha, Nandini Taraganth, Rami Bahlawan, Rana Roshanineshat, Rasha Aleed, Dr. Sifeddine Rafik, Vanessa Maakaroun, Yining Dong, for their constant friendship, endless support. All of whom I have been honored to spend countless , yet memorable times both inside and outside the lab. I would also like to express my gratitude to my previous stagières, Sarah Dahman and Nastaran Razmjooyhassankhani, for their constant dedication to their experimental work. I would like to acknowledge Dr. Eugenio Spadoni and Dr. Mingqin Li for their knowledge, tutoring, and working with me at the beginning of my study; Dr. Lan Liu for her research contributions for performing mass spectrometry analysis; and Dr. Asma Mdimagh for her research contributions for sensory statistical analysis.

Last but not least, the completion of this degree would not have been possible without the everlasting and unconditional support, love, and motivation from family, aka, my best friends. No words can explain how grateful I am for my parents, Hanan and Khaled, and my twin siblings, Kenzie and Taimour, who have always encouraged me to follow my dreams despite any challenges. Thank you, Sassie, for contributing to this work with your furry paws...my buddy.

TABLE OF CONTENTS

ABSTRA	ACT				
RÉSUM	ÉVI				
STATE	MENT FROM THE THESIS OFFICEIX				
PREFAC	CE AND CONTRIBUTION OF AUTHORSX				
RESEAI	RCH CONTRIBUTIONSXII				
ACKNO	WLEDGMENTSXIII				
TABLE	OF CONTENTS XIV				
LIST OF	FIGURES XIX				
LIST OF	TABLES XXI				
NOMEN	CLATURE/LIST OF ABBREVIATIONSXXIII				
CHAPT	ER I. GENERAL INTRODUCTION1				
CHAPT	ER II. LITERATURE REVIEW4				
2.1	Introduction4				
2.2	Chemical composition of cacao bean shell6				
2.3	Bioactive/functional chemical composition of cacao bean shells7				
2.3.1	Dietary fibre7				
2.3.1.1	Pectin				
2.3.1.2	Cellulose				
2.3.1.3	Hemicellulose9				
2.3.1.4	Lignin9				
2.4	Proteins and bioactive peptides12				
2.5	Phenolic compounds12				
2.5.1	Phenolic content and protein digestibility13				
2.5.2	Phenolic content and antioxidant ability14				
2.6	Phytosterols15				
2.7	Effect of processing on cocoa bean shell by-products				
2.8	Fermentation21				
2.9	Roasting				
2.10	Extraction methods23				
2.11	Alkaline/acid-assisted extraction23				
2.12	Physical extraction25				

2.12.1	Microwave-assisted extraction25				
2.12.2	Pulsed electric field26				
2.12.3	Sub-critical water extraction27				
2.12.4	Cavitation extraction techniques27				
2.12.4.1	Ultrasound-assisted extraction27				
2.12.4.2	Hydrodynamic cavitation28				
2.12.5	Pressurized-liquid extraction				
2.13	Enzymatic-based approaches				
2.14 function	Cell wall feruloylated oligo- and polysaccharides: biotransformation and alities				
2.14.1 oligosacc	Hydrolysis of cell wall polysaccharides for the generation of feruloylated				
2.15	Feruloylation of cell wall oligo- and polysaccharides32				
2.16	Functional properties of cell wall feruloylated oligosaccharides				
2.16.1	Antioxidant activity				
2.16.2	Prebiotic potential				
2.16.3	Immunomodulatory activity				
2.17	Conclusion				
CONNE	CTING STATEMENT I40				
CHAPT SHELL	ER III. EXTRACTION AND CHARACTERIZATION OF COCOA BEAN CELL WALL POLYSACCHARIDES41				
3.1	Abstract				
3.2	Introduction				
3.3	Materials and Methods43				
3.3.1	Materials43				
3.3.2	Preparation of cocoa bean shells43				
3.3.3	Proximate compositional analysis of cocoa bean shells44				
3.3.4	Isolation of cell wall materials from cocoa bean shells44				
3.3.5	Alkaline extraction of cocoa bean shell cell wall material polysaccharides45				
3.3.6	Structural characterisation of carbohydrates45				
3.3.6.1	Determination of total neutral sugars and uronic acid contents45				
3.3.6.2	Monosaccharide profile45				
3.3.6.3	Molecular weight distribution45				

3.3.7	Extraction and determination of polyphenols content46				
3.3.8	Determination of phenolic compounds via LC-MS46				
3.3.9	Prebiotic activity assay47				
3.3.10	Determination of short chain fatty acid catabolites47				
3.3.11	Statistical analysis				
3.4	Results and Discussion				
3.4.1	Characterization of cocoa bean shells and its cell wall material48				
3.4.2	Polyphenolic profile of cocoa bean shells and its cell wall material52				
3.4.3	Extraction of cell wall polysaccharides and their characterisation53				
3.4.4	Prebiotic activity				
3.5	Conclusion				
Referen	ces63				
CONNE	CTING STATEMENT II69				
CHAPT COCOA FERUL	ER IV. ENZYMATIC GENERATION OF OLIGOSACCHARIDES FROM A BEAN SHELLS USING MULTI-ENZYMATIC PRODUCTS AND THE OYLATION OF OLIGO/POLYSACCHARIDES				
4.1	Abstract				
4.2	Introduction71				
4.3	Materials and Methods72				
4.3.1	Materials72				
4.3.2.	Alkaline extraction of CBS polysaccharides73				
4.3.3.	Bio-generation of oligosaccharides from CBS polysaccharides73				
4.3.4.	Analytical methods for the quantification of oligo- and polysaccharide extracts .74				
4.3.4.1.	Determination of total neutral sugars and uronic acid contents74				
4.3.4.2.	Molecular weight distribution profile74				
4.3.5.	Preparation of feruloyl esterase and feruloylation of oligo- and polysaccharides 74				
4.3.5.1.	Immobilization of feruloyl esterase74				
4.3.5.2.	Esterification activity of free and immobilized feruloyl esterase75				
4.3.5.3.	Feruloylation of oligo- and polysaccharides76				
4.3.6.	Functional properties76				
4.3.6.1.	Prebiotic activity assay76				
4.3.6.2.	Determination of short chain fatty acid catabolites77				
4.3.6.3.	Total phenolic content77				

4.3.6.4.	Antioxidant Activity78				
4.3.6.4.1.	. 2,2-Diphenyl-1-picrylhydrazyl (DPPH) Radical Scavenging Assay				
4.3.6.4.2.	2. Oxygen Radical Absorption Capacity (ORAC) Assay				
4.3.6.4.3.	2,2′ -azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) decolorization assay79				
4.3.7.	Statistical analysis				
4.4	Results and Discussion				
4.4.1	Bio-generation of oligosaccharides from cocoa bean shells				
4.4.1.1	Composition of cocoa bean shell polysaccharides80				
4.4.2	Enzymatic generation of oligosaccharides from cocoa bean shells polysaccharides. 				
4.4.3	Feruloylation of oligo and polysaccharides from cocoa bean shells				
4.4.3.1	Feruloyl esterase immobilization86				
4.4.3.2	Feruloylation cocoa bean shell oligo- and polysaccharide extracts				
4.4.4	Functional Properties				
4.4.4.1	Prebiotic Activity				
4.4.4.2	Antioxidant Activity				
4.5	Conclusion				
Referenc	es98				
Supplem	entary Data105				
CONNE	CTING STATEMENT III106				
CHAPTI FROM C BASED I	ER V. PILOT PLANT EXTRACTION OF OLIGO/POLYSACCHARIDES COCOA BEAN SHELLS AND THEIR INCORPORATION INTO CHOCOLATE FORMULATIONS				
5.1	Abstract				
5.2	Introduction107				
5.3	Materials and Methods108				
5.3.1	Materials108				
5.3.2	Lab-scale extraction of cocoa bean shell polysaccharides109				
5.3.3	Pilot-plant extraction of cocoa bean shell oligo- and polysaccharides109				
5.3.4 extracts	Analytical methods for the quantification and characterization of carbohydrate				
5.3.4.1	Determination of total neutral sugars and uronic acid contents110				
5.3.4.2	Monosaccharide profile110				
5.3.4.3	Molecular weight profile110				

5.3.5	Preparation of chocolate	111		
5.3.6	Sensory evaluation of chocolate formulations			
5.3.7	Statistical Interpretation			
5.4	Results and Discussion1			
5.4.1 isolation	Recovery of carbohydrate extracts using laboratory and pilot-plant scale up	.113		
5.4.2	Chocolate-based formulations enriched with carbohydrate extract	.120		
5.4.2.1	Statistical interpretation of chocolate-based formulations	120		
5.4.2.2	Correlation of sensory attributes120			
5.5 Conc	lusion	130		
Reference	ces	132		
CHAPT	ER VI. GENERAL CONCLUSION AND FUTURE WORK	136		
Reference	ces	.139		

LIST OF FIGURES

Figure 3.2. Prebiotic activity scores of *Lactobacillus rhamnosus* GG (R0343 AR) on polysaccharide extracts, inulin and rhamnogalacturonan, measured at 6,24, 48, 72 and 96h of incubation. Values represented as an average \pm standard deviation. Poly NAC: polysaccharide NAC variety, Poly CCN: polysaccharide CCN variety, RhamnoG: Rhamnogalacturonan, respectively. For each incubation time, bars with different letters represent scores significantly different at P < 0.05.

Figure 4.2. A) Prebiotic activity scores of *Lactobacillus rhamnosus GG* (Harmonium R0343 AR) on oligosaccharide extracts prepared from multi-enzymatic hydrolysis, inulin and rhamnogalacturonan, measured at 6,24, 48, 72 hr of incubation. B) Prebiotic activity scores of *Bifidobacterium longum* (ATCC ® 15707) on oligosaccharide extracts prepared from multi-enzymatic hydrolysis, inulin and rhamnogalacturonan, measured at 6, 24, 48, 72 hr of incubation. Values represented as an average \pm standard deviation. Oligo NAC: oligosaccharide NAC variety, Oligo CCN: oligosaccharide CCN variety, RhamnoG: Rhamnogalacturonan, respectively. For each incubation time, bars with different letters represent scores significantly different at P < 0.05

LIST OF TABLES

Table 2.1 Proximate Analysis of Cocoa beans and its by-product, CBS10
Table 2.2. Fatty acids, Vitamins and Crude Minerals Composition of CBS 11
Table 2.3. Protein and dietary fiber comparison between CBS and coffee silver-skin
Table 2.4. Phytochemical contents of cocoa bean shells in terms of phenolic compounds, phytosterols, and alkaloids 18
Table 2.5 Antoxidant activity of cocoa bean shells at various conditions
Table 2.6. Applications of cocoa bean shells among several industries
Table 3.1. Proximate composition of Cocoa Bean Shells (CBS) and saccharide profile of their corresponding Cell Wall Materials (CWM)
Table 3.2. Alkaline extraction of polysaccharides from the cell wall material (CWM) of cocoa bean shells (CBS)
Table 3.3. Concentration (10 ⁻³ mol/L) of short chain fatty acids released in the 0.5M alkaline treated cell-wall polysaccharide samples (NAC, CCN) as compared to glucose, inulin and rhamnogalacturonan standards which were obtained from <i>Lactobacillus rhamnosus</i> GG (R0343 AR) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation
Table 3.4. Concentration (10 ⁻³ mol/L) of short chain fatty acids released in the 0.5M alkaline treated cell-wall polysaccharide samples (NAC, CCN) as compared to glucose, inulin and rhamnogalacturonan standards which were obtained from <i>Bifidobacterium longum</i> (ATCC 15707®) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation
Table 4.1. Characterization of cocoa bean shell polysaccharide extracts
Table 4.2. Efficiency of bi and multi-enzymatic systems used for the enzymatic generation of oligosaccharides from the NAC cell wall polysaccharides after 7hr reaction time
Table 4.3. Efficiency of bi and multi-enzymatic systems used for the enzymatic generation of oligosaccharides from the CCN cell wall polysaccharides after 7hr reaction time
Table 4.4. Comparison of the hydrolytic and esterification activities of feruloyl esterases in theirfree and immobilized (modified IDA-Cu-Sepabeads EC-ER®) forms
Table 4.5. Feruloylation extent (%) of cocoa bean shell oligo- and polysaccharides (FA-OCCN, FA-ONAC, FA-PCCN, FA-PNAC enzymatically generated oligosaccharides using free feruloyl esterases (2U/l) from rumen microorganism and H. insolens at 144hr

Table 4.6. Concentration (10⁻³ mol/l) of short chain fatty acids released in the cell-wall oligosaccharides samples (NAC, CCN) prepared from multi-enzymatic hydrolysis as compared to glucose, inulin and rhamnogalacturonan (RhamnoG) standards which were obtained from *Lactobacillus rhamnosus* GG (R0343 AR) and *Bifidobacterium longum* (ATCC® 15707) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation....................94

NOMENCLATURE/LIST OF ABBREVIATIONS

AJS ESI	Agilent jet stream electrospray ionization			
AnFaeA	Feruloyl esterase from Aspergillus niger			
ANOVA	Analysis of variance			
AOAC	Association of Official Analytical Chemists			
AXE	Acetyl xylan esterase			
CBS	Cocoa bean shells			
CDTA	Cyclohexanediaminetetraacetic acid			
CFU	Colony forming units			
СНО	Carbohydrate			
CWM	Cell wall material			
CWP	Cell wall polysaccharides			
Da	Dalton			
DPPH	2,2-diphenyl-1-picrylhydrazyl			
EDTA	Ethylenediaminetetraacetic acid			
ESI-MS	Electrospray ionization mass spectrometry			
EtOH	Ethanol			
FAE	ferulic acid esterase			
FRAP	Ferric reducing antioxidant power			
GAE	Gallic acid equivalent			
HC	Hydrodynamic cavitation			
HC1	Hydrochloric acid			
HPAEC	High performance anion exchange chromatography			
HPLC	High performance liquid chromatography			
HPSEC	High performance size exclusion chromatography			
HSD	Honest significant difference			
КОН	Potassium hydroxide			
LC-MS	Liquid chromatography mass spectrometry			
LDL	Low-density lipoprotein			
m/z	mass to charge ratio			
MAE	Microwave-assisted extraction			

MANOVA	Multi-analysis of variance		
MES-NAOH	2-Morpholinoethanesulfonic acid - sodium hydroxide		
mМ	Millimolar		
MRS	de Mann regosa		
MS	Mass spectrometry		
MW	Molecular weight		
Na ₂ CO ₃	Sodium carbonate		
NaBH ₄	Sodium borohydride		
NaClO ₂	Sodium chlorite		
NaOH	Sodium hydroxide		
NcFaeD	Feruloyl esterase from Neurospora crassa		
PA	Prebiotic activity score		
PAW	Phenol: Acetic acid :water		
PDCAAS	Protein digestibility-corrected amino acid score		
PEF	Pulsed electric field		
PLE	Pressurized liquid extraction		
PS	Polysaccharide		
Q-TOF-MS	Quadrupole time of flight mass spectrometry		
qPCR	Quantitative polymerase chain reaction		
RGI	Rhamnogalacturonan I		
SCFA	Short chain fatty acids		
StFaeC	Type C FAE from Sporotrichum thermophile		
SWE	Subcritical water extraction		
TBARS	thiobarbituric acid reactive substance assay		
TEAC	Trolox equivalent antioxidant capacity		
TFA	Trifluoroacetic acid		
TNF-α	Tumor necrosis factor alpha		
TsFaeC	Feruloyl esterase from Talaromyces stipitatus		
UAE	Ultrasound-assisted extraction		

CHAPTER I. GENERAL INTRODUCTION

Theobroma cacao L., also known as the cacao tree, is natively grown in tropical climate countries on the equatorial belt such as Ivory Coast, Ghana, Nigeria, Cameroon, Mexico, Brazil, Ecuador, and Indonesia. It has been cultivated for use ever since the Aztec and Mayan era in South America (Motamayer et al., 2008). Cocoa beans are seeds of the *T. cacao* L. in which undergo a sequence of post-harvesting practices, namely fermentation, drying and roasting to obtain cocoa, the main ingredient used to produce chocolate. However, throughout these post-harvesting conditions, cocoa biomass is generated. In fact, cocoa bean production generates an estimated 15% cocoa biomass (International Cocoa Organization, 2017). One of these by-products are the cocoa bean shells, the outermost layer of the bean, removed in the dehulling or deshelling stage to retrieve the cocoa beans (Younes et al., 2022). Efforts to valorize cocoa bean shells have been explored in several applications, such as in agricultural, feedstuff, and bioremediation practices (Yajima et al., 2016; Magistrelli et al., 2016; A. Adeyina et al., 2010; Soeharsono et. al., 2017).

Indeed, the valorization of biomass for their rich dietary fiber composition is of high interest. Cocoa bean shells account for a high dietary fiber (13.86 - 60.6 %, w/w), and varies depending on the applications of roasting and drying (Agus et al., 2018; Bamba et al., 2015; Fakhlaei et al., 2019; Martínez et al., 2012; Okiyama et al., 2017; Rojo-Poveda et al., 2019, 2020; Younes et al., 2022), and polyphenol composition (1.45-94.95 mg GAE/g) (Delgado-Ospina et al., 2020; Lessa et al., 2018; Martínez et al., 2012; Mellinas et al., 2020; Nsor-Atindana et al., 2012; Rojo-Poveda et al., 2020; Utami et al., 2016; Younes et al., 2022). Cocoa bean shell cell-wall polysaccharides, a sub-class of dietary fiber, comprise of 40–45% pectin, 34% cellulose, and 21–26% hemicellulose (Redgwell et al., 2003). Pectic regions are mostly attributed the presence of rhamnogalactouronan and arabinogalactan. Hemi-cellulosic fractions comprise of 5-7% xyloglucan, 7-10% galactoglucomannan, and 4% glucuronoarabinoxylan of non-cellulosic cell wall polysaccharides in cocoa bean shells (Redgwell et al., 2003; Vojvodić et al., 2016; Acosta et al., 2018). However, the origin, variety and processing conditions can affect the overall composition of cocoa bean shells. As far as we are aware, this has not been assessed so far. The assessment of cocoa bean shells' nutritional and cell wall compositions between two varieties at three different processing conditions, namely fermentation, drying, and roasting, are to be evaluated in the present study.

Considering the structural complexity of cell wall polysaccharides, the single enzymatic hydrolysis for the retrieval of isolated oligosaccharides can be challenging. Alternatively, commercially available multi-enzymatic products, have been previously explored for their favorable and efficient isolation enzymatically generated oligomers and oligosaccharides (Waglay et al., 2019; Spadoni Andreani et al., 2021; Khodaei & Karboune, 2016). The prebiotic potential of cell wall oligo- and polysaccharides derived from cocoa bean shells has yet to be reported. Therefore, a comprehensive assessment of the prebiotic capabilities of cocoa bean shells would complement their techno-functional properties for potential use in food formulations.

With a generic structure of one or two ferulic acid moieties esterified to a mono-, oligo- and polysaccharide chains, the generation of feruloylated oligo- and polysaccharides through feruloyl esterase (EC 3.1.1.73)-catalysed esterification reaction has been reported (Tamayo-Cabezas & Karboune, 2019; Spadoni Andreani et al., 2021). Feruloylated oligo- and polysaccharides have attracted scientific attention due to the prebiotic, antimicrobial, anti-inflammatory and antioxidant functions inherent to their structural features (Gálvez Ranilla et al., 2017; Wang et al., 2020). For instance, *in vitro* research showed that feruloylated arabinoxylooligosaccharides are fermented by cultured human colon microbiota (Snelders et al., 2014), while feruloylated oligosaccharides isolated from corn bran showed higher protective effect than ferulic acid against oxidative stress in H₂O₂-injured pheochromocytoma cells (Yao et al., 2014). Currently, the assessment of feruloylated oligo- and polysaccharides from cocoa bean shells has not been conducted.

The overall objective of the present research is to assess the nutritional and cell wall composition of cocoa bean shells as well as investigate the isolation of cell wall polysaccharides from cocoa bean shells by alkaline treatment, the enzymatic generation of oligosaccharides using commercial multi-enzymatic products and the feruloylation of these carbohydrates. The assessment of prebiotic and antioxidant activities of the generated cocoa bean shells carbohydrates (oligo/polysaccharides) and of their contribution to chocolate formulations as prebiotic sweeteners were also carried out in order to gain insight of their properties as functional food ingredients. The specific objectives are as follows:

 Proximate compositional analysis and the evaluation of the polyphenolic profile of cocoa bean shells

- Isolation of cell wall materials/polysaccharides from cocoa bean shells through alkaline treatments and the characterization of their monosaccharide profile and molecular weight distribution.
- Enzymatic generation of oligosaccharides from cell wall polysaccharides using selected commercial multi-enzymatic products and the evaluation of their monosaccharide profile and molecular weight distribution.
- Investigation of the prebiotic potential and the production of short chain fatty acids of cell wall polysaccharides and enzymatic generated oligosaccharides
- 5) Study of the feruloylation efficiency of free and immobilized feruloyl esterases and the assessment of their antioxidant capacities through various assays.
- 6) Pilot-plant extraction trial for the isolation of cell wall oligo- and polysaccharides for use as a functional food ingredient in chocolate-based products.
- Elucidation of the relationship between the sensory attributes and the perceived complexity of panelists of selected chocolate-based formulations enriched with cell wall polysaccharides.

CHAPTER II. LITERATURE REVIEW

2.1 Introduction

Theobroma cacao L., also known as the cacao fruit, is a heavily utilized commodity in the confectionary industry. Cacao beans are cultivated in an acclimatized belt within 20 degrees latitude of the equator where cacao plants are subjected to relatively high temperatures, such as in the Ivory Coast, Ghana, Nigeria, Brazil, and Ecuador (Aprotosoaie et al., 2016). Several varieties of *T. cacao* exist in nature; however only three varieties, Criolla, Forastero and Trinitario, are widely cultivated, with the Forastero variety being the most extensively planted around the world (Panak Balentić et al., 2018).

During production, raw cacao beans are subjected to an extensive set of processing methods and conditions, such as roasting to render an end-product known as cocoa. It should be noted that the composition of cacao beans and its by-products vary depending on the clone variety, climate, harvest location, fruit maturity as well as post-harvesting conditions (Agus et al., 2018). In 2017, 4.7 million tons of cocoa beans were produced globally, where 700,000 tons of cocoa waste were generated in return (International Cocoa Organization, 2017). Cacao by-products are discarded as waste which negatively influence the environment and economic value of cocoa beans. Cocoa bean shells (CBS), also referred to as bean husk, hull, or skin, are the outer layer of the cocoa bean, which is removed in the de-hulling step to retrieve cacao nibs. CBS are considered as one of the heavily discarded by-products of cacao which constitutes around 17-20% of the total cacao bean weight.

CBS valorization is being explored in agricultural and bioremediation fields, such as animal feed, fertilizer, use as a bio adsorbent as well as a source of biofuel (Table 2.6). However, in the case of animal applications, serious caution is advised given that the presence of polyphenols, such as theobromine, are considered harmful to animals. Nonetheless, Adamafio et al., (2013) investigated the in-situ purification of cocoa pod husks with *Candida krusei* to reduce theobromine and other phenolic compounds for use in animal feed. The authors indicated a significant reduction in theobromine (62.5%) as in the total phenolic content (41.5%). Considering that animals digest polyphenols at a slower rate, in-situ approach treatments shows promising approach towards enhancing polyphenolic digestibility in animals.

In addition, researchers have also explored its use in several food applications. Adamafio et al., (2013) incorporated CBS powder as a clean label functional ingredient in baked goods to increase their fiber content. Other CBS food explorations include its use as a fat replacer in functional cakes and chocolate muffins (Martínez-Cervera et al., 2011), a stabilizer and color enhancer in pork sausages (Choi et al., 2019), as well as a sugar and fat replacer in chocolate (Barišić et al., 2019).

There is some concern pertaining to the stability of CBS when incorporated into food systems due to the subjection of fermentation, roasting and other processes that involve exposure to temperature changes. As a result, researchers have proposed the idea of CBS extract encapsulation to enhance its stability (Jokić et al. 2020). Nevertheless, it is of great importance to ensure CBS bioavailability upon consumption as the rich content of phytochemicals can promote anticarcinogenic and anti-diabetic activity, protection against neuro-degenerative diseases, as well as enable immunomodulatory activity (Rojo-Poveda et al., 2020). From an economic perspective, the chocolate industry can greatly benefit from the valorization of the bean by-product, which would ultimately reduce their production, manufacturing, and transportation costs. CBS can be used as a low-cost, clean label, functional ingredient to substitute cocoa as well as serve as a cocoa flavoring agent (Rojo-Poveda et al., 2020). Furthermore, CBS is a rich source of dietary fibre due to the presence of various cell-wall polysaccharides (CWP) such as, cellulose, pectin, 1,3 and 1,4 β -glucans, and arabinoxylans. During processing, these non-starch functional components will migrate from within the bean to the outer regions of the shell (Harris & Smith, 2006; Anderson et al., 2009; Zhu, 2018).

For years, researchers have explored the interaction of CWP with polyphenols and/or proteins in food-food ingredients. These studies primarily explore the synergistic effects between these biopolymers, in terms of their antioxidant and prebiotic activities, as well as their potential in disease prevention. Studies have also investigated the ionic and hydrophobic interactions of polyphenol-CWP and feruloylation as possible approaches for modulating the health promoting properties of cell wall polysaccharide type functional ingredients (Padayachee et al., 2013; Tamayo-Cabezas & Karboune, 2019). There are several factors to consider that may impact the interaction between polyphenols, proteins and CWP which include 1) polyphenol structure, 2) CWP structure, 3) environmental factors, i.e. temperature and pH, 4) polysaccharide-polyphenolprotein interactions, and 5) bioavailability (Zhu, 2018). To the best of our knowledge, chemical and enzymatic extraction methods on CBS and the modification of CBS oligosaccharides for enhanced functionality have not been investigated. The scope of this review will therefore highlight the chemical and bioactive compositions of CBS, bioactive chemical changes by means of processing as well as methods of extraction and isolation of the cell wall polysaccharides from CBS and of their biotransformation, in particular for the generation of feruloylated oligo/polysaccharides.

2.2 Chemical composition of cacao bean shell

Proximate composition analyses of CBS have been previously explored for its use in food, animal feed and other applications (Rojo-Poveda et al., 2020; Panak Balentić et al., 2018). CBS chemical composition was found to vary according to several production factors, such as genotype, soil, climate, harvest conditions, quality. As well as several processing factors such as fermentation, roasting and winnowing conditions (Agus et al., 2018). Table 2.1 summarizes CBS proximate analysis under various conditions, such as unprocessed, fermented, roasted, etc. Several studies explored certain macro and micro-components in CBS that promote its bioactive functionality to human health.

The carbohydrate content in cocoa beans ranges between 3.62-55.85% w/w (Table 2.1) which is related to the flavor development and profiles of cocoa bean products upon fermentation, drying, and roasting (Agus et al., 2018). As seen in Table 2.1, roasted CBS were found to have a higher proportion of carbohydrates compared to unroasted CBS and roasted cocoa beans. This could be attributed to the degradation of mucilaginous pulp and the migration of carbohydrates towards outer shell layer during processing (Agus et al., 2018). In contrast, protein, a macronutrient that contributes to cocoa's flavor profile, was observed to decrease upon roasting (El-Saied et al., 1981) suggests that approximately 10% of the protein content, is present in its unbound form while the remainder, in the form of alpha-amino nitrogen, is bound to oxidized polyphenols, which upon roasting transform into polyphenol-quinones. Additionally, the fat content will fluctuate depending on the degree of roasting applied on the material, as cocoa butter is inclined to migrate from its cotyledon layer to the cocoa shell (El-Saied et al., 1981). Comparing the fat content in CBS to cocoa butter, El-Saied et al. (1981) reported that CBS accounted for a greater percentage of fat content, lower melting point, higher saponification value, as well as lower iodine value. Furthermore, CBS was also found to be richer in short-chain unsaturated fatty acids compared to

cocoa butter (Table 2.2). In terms of the vitamin composition, CBS are found to contain vitamins B1, B2, B6, D, and E. Knapp and Coward (1935) predicted that vitamin D is produced as a result of the activation of ergosterol, a vitamin D precursor, upon exposure to sun drying. On the other hand, Oracz et al. (2014) concluded that the roasting process is known to reduce the tocopherol content, a class of compounds promoting vitamin E activity (Table 2.2). Alternatively, the ash composition in CBS contains a great source of calcium, phosphorus, potassium and more (Fakhalei et al., 2019; Agus et al., 2018; Martinez et al., 2012; Mellinas et al., 2020; Okiyama et al., 2017). Minerals in CBS continuously accumulate with increasing cocoa bean processing time, further reinforcing its use as low-cost fertilizer.

2.3 Bioactive/functional chemical composition of cacao bean shells

2.3.1 Dietary fibre

Dietary fibre is the edible part of plants or analogous carbohydrate that is resistant to digestion and absorption in the small intestine, with complete or partial fermentation in the large intestine (American Association of Cereal Chemists, 2001). In CBS, the dietary fibre content accounts for 13.86-60.6 (w/w %), and varies depending on the applications of roasting and drying (Fakhlaei et al., 2019; Martínez et al., 2012; Agus et al., 2018; Okiyama et al., 2017). In roasted CBS, dietary fibre is likely present in the form of complexes with protein from Maillard reaction (Redgwell et al., 2003). Cell-wall polysaccharides (CWP) are a sub-class of dietary fibres that are heavily constituted in CBS, roughly comprising of 40-45% pectin, 34% cellulose, and 21-26% hemicellulose (Redgwell et al., 2003). As shown in Table 2.3, the CBS monosaccharide profile of CWP material comprises of glucose (63%), galactose (12%), mannose (7.5%), arabinose (6.6%) rhamnose (3%), and xylose (4.6%) (Vojvodić et al., 2016; Acosta et al., 2018; Grillo et al., 2019). Comparing the monosaccharide profiles, CBS have a relatively lower glucose content than cocoa beans. However, CBS contains more rhamnose, galactose, mannose, xylose, and uronic acid content than cocoa beans, indicating less cellulosic polysaccharides and greater pectic and noncellulosic CWP. According to Redgwell et al. (2003), increased rhamnose content in CBS was a result of increased rhamnosylation of the pectin backbone compared to its content in cocoa beans. Furthermore, the neutral sugar residues of the non-cellulosic region were predominantly mannose, whereas cocoa beans were predominantly arabinose and galactose (Redgwell et al., 2003). As seen in Table 2.3,, the monosaccharide profile of CBS is comparable to coffee silver-skin, a by-product of the coffee industry used as a potential dietary fibre supplement, with enriched xylose in CBS

(Mussatto et al., 2011; Ballesteros et al., 2014; Vojvodić et al., 2016; Acosta et al., 2018; Grillo et al., 2019; Gemechu, 2020). Sánchez et al. (2010) suggests that fibre from the cocoa cell wall was shown to exhibit health-promoting properties, by means of decreasing total cholesterol, triglyceride, plasma glucose and insulin levels in Zucker rats. This indicates the ability of soluble dietary fibers to counter balance cardiovascular diseases and other metabolic syndromes (Sánchez et al. 2010).

2.3.1.1 Pectin

Pectin is a polysaccharide comprised of (1-4)-linked α -D-galactopyranosyluronic acid units that is primarily used as a gelling agent in food formulations (Biliaderis & Izydorczyk, 2007). Alternatively, pectin has been explored in terms of its ability to interact with polyphenols for use as a microencapsulation agent to promote bio-functionality in foods (Noh et al., 2018). In CBS, pectin constitutes around 7.62-15.59 % (dry weight) depending on the method of extraction (Rojo-Poveda et al., 2020). Pectin is known for its colloidal properties that help support the structure and texture of the products. The techno-functional properties of pectin can be modulated by interactions with other biopolymers, varying degrees of esterification, oxidation and enzymatic degradation (Le Bourvellec & Renard, 2005; Le Bourvellec et al., 2005; Simonsen et al., 2009; Watrelot et al., 2013; Zhu, 2018). Additionally, cross-linking has been shown to greatly impact the interaction of pectin with polyphenols, where Le Bourvellec et al. (2005) observed that the apparent affinity constant of pectin and the procyanidin molecular weight were directly proportional. A higher pectic apparent affinity would facilitate a stronger gel network as well as lead to the formation of hydrophobic pockets to enhance complexation and encapsulation with procyanidins.

2.3.1.2 Cellulose

 β -1,4 D-glucan linked units comprise the basic structure of cellulose, a primary component in plant cell walls (Aspinal, 1980). Depending on the method of isolation and extraction, cellulose constitutes around 15.1-34% w/w (Table 2.3) in CBS (Acosta et al., 2018; Grillo et al., 2019; Gómez Hoyos et al., 2020). Similarly, to pectin, processing, including cross-linking and harsh drying, has shown to impact the structural integrity of cellulose. This can result in decreased

porosity, increased apparent saturation and affinity levels, which enhances pectin's binding affinity with polyphenols (Le Bourvellec et al., 2005).

2.3.1.3 Hemicellulose

Hemicellulose constitutes 11.1% (w/w) of CWP in CBS (Gómez Hoyos et al., 2020), residing with cellulose and pectin, primarily comprised of 5-7% xyloglucan, 7-10% galactoglucomannan, and 4% glucuronoarabinoxylan of non-cellulosic cell wall polysaccharides (Redgwell et al., 2003; Vojvodić et al., 2016; Acosta et al., 2018). Xyloglucan is present in CBS in its fucosylated form, similarly to other cell wall materials of dicotyledons (Bacic et al., 1988; Redgwell & Hansen, 2000). Galactoglucomannan in CBS was found to consist of a β -1,4 glycosyl (13%) and mannosyl (42%) that are substituted at O-6 to provide 2-linked galactose side chain residues as the internal sugar. Other sugar residues were found to bind with 2-linked galactose as the terminal sugar, such as galactose, arabinose or xylose (Redgwell & Hansen, 2000). Relative to cocoa beans, CBS galactoglucomannan appears more dispersed due to the degree of molecular branching and therefore, is more abundant (Redgwell & Hansen, 2000). Glucuronoarabinoxylan consists of β -1,4 xylosyl residues, where every seventh xylosyl residue is substituted at O-4 with glucuronosyl, arabinosyl, and xylosyl residues (Redgwell & Hansen, 2000).

2.3.1.4 Lignin

Lignin is composed of monomer linkages of p-coumaryl, guiacyl, and sinapyl-propane, which have been studied to possess antioxidant properties, which enhance their application in food systems (Redgwell et al., 2003). The lignin content is generally estimated by the insoluble fraction obtained by removal of ash concentrates via acid hydrolysis of the plant tissues, termed as "Klason lignin" (Chen, 2015). As seen in Table 2.3, lignin varies between 15.1-35 % (w/w) (Gomez Hoyos et. al, 2020; Acosta et. al, 2018). Lignin can also form tannin-protein complexes, where Redgwell et al. (2003) concluded it accounts for roughly 30% (dry weight) in non-defatted roasted CBS.

Table 2.1	Proximate	Analysis of	f Cocoa beans	and its by	product, CBS
		~		2	

	Component	% w/w dry material	Reference
Cocoa Bean*	Crude Protein	6.3-10.6 / 7.0-29.0	Sotelo & Alvarez, 1991; Agus et al., 2018; Adamafio, 2013; Olubamiwa et al., 2006; Martín-Cabrejas et al., 1994; Pèrez Sira 2015
	Carbohydrate	3.62-55.85	Sotelo & Alvarez, 1991; Agus et al., 2018; Adamafio, 2013; Olubamiwa et al., 2006; Martín-Cabrejas et al., 1994;
			Pèrez Sira 2015
	Dietary Fibre	3.13-17.8 / 6.0-19.4	Sotelo & Alvarez, 1991; Grillo et al. 2019; Agus, et al., 2018; Olubamiwa et al., 2006; Martín-Cabrejas et al., 1994; Pèrez Sira, 2015
	Crude Fat	5.46-13.93/ 6.0-17 <u>.</u> 3	Sotelo & Alvarez , 1991; Grillo et al. 2019; Agus et al., 2018; Olubamiwa et al., 2006; Martín-Cabrejas et al., 1994; Pèrez Sira , 2015
	Ash	1.41-11.67	Sotelo & Alvarez, 1991; Grillo et al. 2019; Agus et al., 2018; Olubamiwa et al., 2006; Martín-Cabrejas et al., 1994; Pèrez Sira, 2015
Unroasted CBS	Crude Protein	18	Fakhlaei et al., 2019
	Carbohydrate	44.63	Agus et al., 2018
	Dietary Fibre	13.86	Agus et al., 2018
	Crude Fat	4.09-4.87	Fakhlaei et al., 2019; Agus et al., 2018
	Ash	9.93-10.22	Fakhlaei et al., 2019; Agus et al., 2018
Roasted CBS	Crude Protein	16.35	Fakhlaei et al., 2019; Agus et al., 2018
	Carbohydrate	55.85	Agus et al., 2018
	Dietary Fibre	16.06-32.01	Fakhlaei et al., 2019; Agus et al., 2018
	Crude Fat	3.09-6.46	Fakhlaei et al., 2019; Agus et al., 2018
	Ash	10.44-11.67	Fakhlaei et al., 2019; Agus et al., 2018
Dried CBS	Crude Protein	6.2-18.6	Martínez et al., 2012; Mellinas et al., 2020; Okiyama et al., 2017
	Carbohydrate	17.8-23.17	Martínez et al., 2012
	Dietary Fibre	50.4-60.6	Martínez et al. 2012; Okiyama et al., 2017
	Soluble Fibre	14.53-16.24	Martínez et al., 2012
	Insoluble Fibre	35.64-42.17	Martínez et al., 2012
	Crude Fat	2.02-2.05	Martínez et al. 2012; Okiyama et al.,2017
	Ash	6-11.42	Martínez et al., 2012; Mellinas et al., 2020; Okiyama et al.,2017

Component		% w/w dry material	Reference
Crude Fat		1.8-6.87	Vojvodić et al., 2016; Nsor-Atindana et al., 2012; Gómez Hoyos et al., 2020
Fatty Acids	Fatty Acid 16:0	15.6-19.6	Lessa et al., 2018
	Fatty acid 17:0	0.32-1.12	Lessa et al., 2018
	Fatty acid 18:0	24.23-41.67	Lessa et al., 2018
	Fatty acid 20:0	1.24-1.29	Lessa et al., 2018
	Fatty acid 22:0	1.1-3.64	Lessa et al., 2018
	Fatty acid 24:0	0.6-0.83	Lessa et al., 2018
	MUFA 18:1n9c	28.7-35.8	Lessa et al., 2018
	PUFA 18:2n6	2.6-5.4	Lessa et al., 2018
	PUFA 18:3n6	2.3-8.26	Lessa et al., 2018
	PUFA 20:5n-3	1.13-1.73	Lessa et al., 2018
Vitamins	Vitamin B1	0.00007-0.00031	Rojo-Poveda et al., 2020
	Vitamin B2	0.00009-0.00031	Rojo-Poveda et al., 2020
	Vitamin B6	traces	Rojo-Poveda et al., 2020
	Vitamin D	traces - 0.000053	Rojo-Poveda et al., 2020
	Vitamin E	0.000102	Rojo-Poveda et al., 2020
Crude Minerals		9.1	Vojvodić et al., 2016
	Calcium	0.23-0.44	Rojo-Poveda et al., 2020
	Chromium	0.67-4.86	Rojo-Poveda et al., 2020
	Cobalt	0.1	Rojo-Poveda et al., 2020
	Copper	2.35-6.62	Rojo-Poveda et al., 2020
	Magnesium	0.48-1.29	Rojo-Poveda et al., 2020
	Manganese	4.53	Rojo-Poveda et al., 2020
	Phosphorus	0.58-1.00	Rojo-Poveda et al., 2020
	Potassium	1.25-1.82	Rojo-Poveda et al., 2020
	Selenium	0.21	Rojo-Poveda et al., 2020
	Sodium	16.00-192.20	Rojo-Poveda et al., 2020
	Zinc	2.75-19.00	Rojo-Poveda et al., 2020

Table 2.2. Fatty acids, Vitamins and Crude Minerals Composition of CBS

2.4 Proteins and bioactive peptides

Table 2.3 shows that CBS crude protein constitutes 0.7-28.1 % (w/w) depending on the degree of processing, where the protein content decreases with increased processing (Fakhlaei et al., 2019). Albumins and globulins were the major fractions (79%, w/w) of total protein from CBS (Bonvehí & Coll, 1999). The essential amino acids in cocoa husk represent 44.75% of total amino acids, predominately lysine (28.4%), tyrosine (18.7%), phenylalanine (14.1%), and histidine (7.7%) in roasted CBS (Bamba et al., 2015). In comparison, Iriondo-DeHond et al. (2019) reported coffee silver-skin constitutes 11-19 % (w/w) where its essential amino acid profile is predominantly phenylalanine (14.6%), valine (12.7%), lysine (12.6%), and threonine (12.6%) (Table 2.3). Additionally, Bonvehí and Coll (1999) reported an amino acid score of 73 for CBS, comparable to rice (73) and soybean protein (80), as well as a low energetic value of 21.9J/kg further supporting its incorporation in food formulations.

Bioactive peptide fractions, namely albumin (11.5%) and vicilin-like (3.9%), were identified in cocoa beans by means of spontaneous fermentation. To date, the presence of bioactive peptides in CBS have not been explored, however, current studies using cocoa beans suggest that oligopeptides are released through three main pathways, which include 1) cocoa bean autolysis (Sarmadi et al., 2012; Sarmadi et al., 2011; Domínguez-Pérez et al., 2020), 2) exogenous enzymatic hydrolysis (Tovar-Pérez et al., 2019; Martorell et al., 2013; Domínguez-Pérez et al., 2020) and 3) after cocoa bean fermentation (Preza et al., 2010; Marseglia et al., 2019; Marseglia et al., 2014; Ryan et al., 2017; Domínguez-Pérez et al., 2020). Typically, oligopeptides contribute to the flavor profile of cocoa, serving as flavor precursors. Additionally, Domínguez-Pérez et al. (2020) suggest they may provide health benefits as potential antioxidant, anti-diabetic, and antiobesogenic agents.

2.5 Phenolic compounds

Phenolic acids are plant secondary metabolites that are released upon exposure of the plant material to environmental stress. They exist as lignin oligomers/polymers or form hetero-conjugates with CWP (Mandal et al., 2010). Similarly, to the cocoa bean itself, CBS contain a great source of phenolic compounds, aiding to its potential use as a functional ingredient. Studies (Table 2.4) indicate the total flavonoid and phenolic contents in CBS to be within 1.65-40.72 mg
catechin equivalent per gram of dry CBS (Rojo-Poveda et al., 2020) and 17.21-41.82 mg gallic acid equivalent per 100 gram of dry CBS, respectively (Lessa et al., 2018). Sub-classes of flavonoids such as anthocyanins and catechins were identified and explored in terms of their content and functionality in CBS.

CBS contains caffeic (0.004-0.89 mg/g), chlorogenic (0.76 mg/g), ferulic (0.51 mg/g), gallic (0.015-0.48 mg/g) and p-coumaric (0.017-0.04 mg/g), phytic (6g/kg), protocatechuic (0.98-1.32 mg/g) and uronic (82.7 -115.2 mg galacturonic acid/g) acids were identified among several studies (Okiyama et al., 2017; Jokić et al., 2020; Barišić et al., 2019; Mellinas et al., 2020). These phenolic compounds are valued for their potential health-promoting effects, such as inhibiting LDL oxidation, inhibiting pro-inflammatory cytokines as well as promoting radical scavenging, antimicrobial and anti-diabetic activities (Meyer et al., 1998; Călinoiu & Vodnar, 2018; Kumar & Goel, 2019). Flavanols have been identified in CBS as another major class of polyphenols, including 0.03-6.16 mg/g catechin, 0.059-17.7 mg/g epicatechin, and 0.007-0.07 mg/g epicatechin gallate. Flavanols have also shown optimal antioxidant activity by means of "modulating inflammatory mediators", decreasing LDL oxidation levels, as well as providing antiatherogenic properties (Goya et al., 2016; Martín & Ramos, 2017). Similarly, procyanidin A (83 mg/mL) and B types (B1: 0.55-0.85 mg/g; B2: 0.23-1.38 mg/g) were found in CBS extracts (Okiyama et al., 2018; Rossin et al., 2019; Rojo-Poveda et al., 2020). Other flavonoids present in CBS include quercetin, kaempferol, myricetin, and isorhamnetin, at concentration of 4.24 mg/g of dry matter, condensed tannins at 2.36-12.9 mg catechin equivalent per gram of dry matter and anthocyanins at 0.4-0.5 mg per gram of dry matter (Table 2.4).

2.5.1 Phenolic content and protein digestibility

Alternatively, excess polyphenol content may interfere with protein digestibility from CBS. Efforts were made to explore both *in vitro* and *in vivo* protein digestibility of foods from various poly-phenol rich extracts from cocoa powder, black and green tea, coffee, faba bean hulls, and more (Eggum et al., 1983; Jansman et al., 1994; Bohn, 2014; Cirkovic Velickovic & Stanic-Vucinic, 2018). The general consensus obtained from *in vitro* studies concluded that both covalent and non-covalent interaction between polyphenols and proteins decreased protein digestibility. Charlton et al. (2002) suggested that the key factors influencing protein affinity to polyphenols were the amino acid composition and their structures. A green coffee extract rich in chlorogenic

acid was found to slightly decrease protein digestion upon supplementation in bread (Budryn et al., 2013; Cirkovic Velickovic & Stanic-Vucinic, 2018). However, coffee polyphenol extracts supplemented with β-lactoglobulin increased protein digestion (Stojadinovic et al., 2013; Cirkovic Velickovic & Stanic-Vucinic, 2018). Świeca et al. (2013) observed decreased protein digestion upon supplementation of dry onion skin, as the primary source of phenols, in bread.

In vivo studies primarily focus on the protein digestibility-corrected amino acid score (PDCAAS) to determine the protein value in human nutrition (Cirkovic Velickovic & Stanic-Vucinic, 2018). Jansman et al. (1994) supplemented fava bean hull extracts, abundant in condensed tannins, in a casein-rich diet. The outcome resulted in decreased protein digestibility in rats. Similarly, green coffee also showed a negative correlation with protein digestibility upon supplementation in soy-bean meal and barely diets in rats. Alternatively, this was not the case for cocoa supplementation, as findings suggested there was no impact on protein digestibility (Eggum, et al., 1983).

2.5.2 Phenolic content and antioxidant ability

Due to the rich polyphenolic content in CBS, the antioxidant ability of this by-product should be explored in order to determine its potential use as a functional ingredient. Table 2.5 summarizes the antioxidant capacity of CBS at various conditions using different assay methods. The 2,2diphenyl-1-picrylhydrazyl (DPPH) activity was measured at 78.5-79.2% for unprocessed CBS, 19.4% for unfermented CBS, and 7.881.3% for fermented CBS (Utami et al., 2016; Agus et al., 2018; Lessa et al., 2018). The large variation in DPPH activity for fermented CBS is due to several factors, such as fermentation time, pH, and roasting time. The ferric reducing antioxidant power (FRAP) activity was estimated for dried CBS samples between 47-84 mM Trolox/g, while undetermined conditions of CBS were estimated at 0.67-72.32 mM Trolox/g, depending on the method of CBS extraction prior (Lecumberri et al., 2007; Martínez et al., 2012; Okiyama et al., 2018; Mellinas et al., 2020). The ABTS assay, also known as Trolox equivalent antioxidant capacity (TEAC) assay, was estimated between 2.56-7.73 mM Trolox/g, with no mention of the degree of processing applied on CBS (Lecumberri et al., 2007; Martínez et al., 2012). Lastly, thiobarbituric acid reactive substance assay (TBARS), was assessed for dried CBS between 0.17-0.53 mM Trolox/g (Choi et al., 2019). It is important to note that the antioxidant capacity of CBS will also vary drastically depending on the condition and mode of processing. Generally, processed CBS samples accumulate a higher antioxidant activity compared to unprocessed samples (Agus, et al., 2018).

2.6 Phytosterols

Referred to as plant lipids, phytosterols identified in CBS were composed of stigmasterol (4.8-15.5%), campsterol (0.12%), cholesterol (1.72%) and β -sitosterol (0.0078-0.21%). Phytosterols in CBS were shown to be less abundant as compared to those in cocoa beans in both roasted and unroasted forms (El-Saied et al., 1981; Agus et al., 2018). Nonetheless, they are viewed as a critical class of compounds serving anti-inflammatory, anti-bacterial, anti-fungal, anti-tumoral activities (Sánchez-Machado et al., 2004). Additionally, phytosterols were found to reduce cholesterol absorption in the intestine, further reducing blood cholesterol levels in humans (Agus et al., 2018; Cantril, 2008).

Component		% w/w dry material (relative %)	Reference
Cocoa bean shells			
Crude Protein		0.7 - 28.1	Okiyama et al., 2018; Agus et al., 2018; Vojvodić et al., 2016; Bamba et al., 2015; Nsor-Atindana et al., 2012; Aregheore, 2002
Amino Acids	Lysine	1.45 (28.4%)	Bamba et al., 2015
	Arginine	0.48 (9.4%)	Bamba et al., 2015
	Histidine	0.39 (7.7%)	Bamba et al., 2015
	Phenylalanine	0.72 (14.1%)	Bamba et al., 2015
	Tyrosine	0.95 (18.7%)	Bamba et al., 2015
	Leucine	0.35 (6.9%)	Bamba et al., 2015
	Isoleucine	0.27 (5.3%)	Bamba et al., 2015
	Valine	0.22 (4.3%)	Bamba et al., 2015
	Threonine	0.19 (3.73%)	Bamba et al., 2015
	Methionine	0.07 (1.4%)	Bamba et al., 2015
Dietary Fibres*		13.86 - 60.6	Agus et al., 2018; Fakhlaei et al., 2019; Rojo-Poveda et al., 2020; Rojo-Poveda et al., 2019; Martínez et al., 2012; Okiyama et al., 2017
	Total	25.9	Grillo et al., 2019
	Monosaccharides Glucose	16 5 (63%)	Grillo et al., 2019
	Galactose	3.1 (12%)	Grillo et al., 2019
	Mannose	2.6 (7.5%)	Grillo et al., 2019
	Rhamnose	0.8 (3%)	Grillo et al., 2019
	Xvlose	1.2 (4.6%)	Grillo et al., 2019
	Arabinose	1.7 (6.6%)	Grillo et al., 2019
	Cellulose	15.1-34	Grillo et al., 2019; Gómez Hoyos et al.,
	Hemicellulose	11.1**	2020; Acosta et al., 2018 Gómez Hoyos et al., 2020
	Lignin	15.6 -35	Gómez Hoyos et al. 2020; Acosta et al., 2018
	Holocellulose	49	Acosta et al., 2018
	Pectin	7.62 - 15.59	Rojo-Poveda et al., 2020
	Uronic acid	3.48 - 7.13	Lecumberri et al., 2007
	Galacturonic acid	6.9	Vojvodić et al., 2016
Coffee Silver-skin			
Crude Protein		11-19	
Amino Acids	Lysine	0.177 (12.6%)	Iriondo-DeHond et al., 2019
	Arginine	0.114 (8.1%)	Iriondo-DeHond et al., 2019

Table 2.3. Protein and dietary fiber comparison between CBS and coffee silver-skin

0.092 (6.5%)

Iriondo-DeHond et al., 2019

	Histidine		
	Phenylalanine	0.205 (14.6%)	Iriondo-DeHond et al., 2019
	Tyrosine	0.148 (10.5%)	Iriondo-DeHond et al., 2019
	Leucine	0.154 (10.9%)	Iriondo-DeHond et al., 2019
	Isoleucine	0.104 (7.4%)	Iriondo-DeHond et al., 2019
	Valine	0.179 (12.7%)	Iriondo-DeHond et al., 2019
	Threonine	0.178 (12.6%)	Iriondo-DeHond et al., 2019
	Methionine	0.057 (4.0%)	Iriondo-DeHond et al., 2019
Dietary Fibers		54-74	Ballesteros et al., 2014; Gemechu 2020; Borrelli et al., 2004; Costa et al., 2018; Iriondo-DeHond et al., 2019; Jiménez- Zamora et al., 2015; Toschi et al., 2014)
	Total	not reported	
	Monosaccharides Glucose	17.8	
	Galactose	3.8	Mussatto et al., 2011; Ballesteros, et al., 2014; Gemechu, 2020
	Mannose	2.6	Mussatto et al., 2011; Ballesteros et al., 2014; Gemechu, 2020
	Rhamnose	not reported	
	Xylose	4.7-7.61	Mussatto et al., 2011; Ballesteros et al., 2014; Gemechy 2020
	Arabinose	2.0 - 7.61	Mussatto et al., 2011; Ballesteros et al., 2014; Gemechu, 2020
	Cellulose	18 -23.8	Ballesteros et al., 2014; Gemechu , 2020
	Hemicellulose	13 -16.7	Ballesteros et al., 2014; Gemechu , 2020
	Lignin	28.6	Ballesteros et al., 2014; Gemechu, 2020
	Holocellulose	not reported	
	Pectin	not reported	
	Uronic acid	10	Furusawa et al., 2011
	Galacturonic acid	not reported	

*specific to roasted and dried CBS ** reported as % w/w of total cell wall polysaccharides in CBS

Con	nponent	Extraction Method	Value*	Reference
Total Flavor	oids content	Acetone, Ethanol, Methanol, Water	1.65-40.72 mg CE/g	Rojo-Poveda et al., 2020; Nsor-Atindana et al., 2012
Total Phenolic content		Methanol-Acetone/Water/Soxhlet/Microwave Assisted/Ethanol/Hexane- Acetone/Methanol/Acetone/Hydroethanol/Acetone- Water-Acetic Acid	1.45-94.95 mg GAE/ g	Nsor-Atindana et al., 2012; Mellinas et al., 2020; Utami et al., 2016; Lessa et al., 2018; Martínez et al., 2012; Delgado-Ospina et al., 2020; Rojo- Poveda et al., 2020
Total Tannii	n content	Ethanol-Water/ Ethanol-Methanol/Ethanol Acetone	85-359.7 mg CE / L	Rojo-Poveda et al., 2020; Rossin et al., 2019
Total Conde	nsed Tannins	Water/Ethanol/Methanol/Acetone	2.36-12.9 mg CE/g	Nsor-Atindana et al., 2012
Carotenoids		Hydro-ethanol	0.2-0.8 mg b- carotene/100g	Lessa et al., 2018
Phenolic Acids	Caffeic acid	Methanol/Subcritical Water/Microwave Assisted Extraction	0.0004-0.002136 g/100g	Mellinas et al., 2020; Jokić et al., 2020; Barišić et al., 2019
	Chlorogenic acid	Water/Microwave Assisted Extraction	0.001824 g/100g	Mellinas et al., 2020
	Ferulic acid	Water/Microwave Assisted Extraction	0.001224 g/100g	Mellinas et al.,2020
	Gallic acid	Subcritical Water	0.0037-0.0048 g/100g	Jokić et al., 2020
	p-coumaric acid	Methanol/Subcritical Water	0.0002 g/100g	Jokić et al., 2020
	Phytic acid	N.D.	0.6 g/100g	Okiyama et al., 2017; Arlorio et al., 2005
	Protocatechuic acid	Microwave Assisted	0.002353-0.003168 g/100g	Mellinas et al., 2020
	Galacturonic	n.a (raw material)	6.9 g/100g	Vojvodić et al., 2016
	Uronic acid	Microwave Assisted/Soxhlet	82.7-115.2 mg GlcA/g	Mellinas et al., 2020
Anthocyanins		In natura/Hydro-ethanol	0.4-0.5 mg QE/100g	Lessa et al., 2018
Flavanols		Water	0.424g/100g	Soares et al., 2020
	Procyanidin A- type pentoside	Ethanol/Acetone	0.166 g/100g	Rossin et al., 2019

Table 2.4. Phytochemical contents of cocoa bean shells in terms of phenolic compounds, phytosterols, and alkaloids

	Procyanidin B1	Water	55-70 g/100g	Rojo-Poveda et al., 2020; Papillo et al., 2019
	Procyanidin B2	Accelerated solvent extraction/Ethanol-Methanol	0.0138-0.03728 g/100g	Rojo-Poveda et al., 2020; Papillo et al., 2019; Rossin et al., 2019; Okivama et al., 2018
	Catechin	Methanol/Hexane/Subcritical Water	0.0088-0.024g/100g	Rojo-Poveda et al., 2020; Okiyama et al., 2018; Papillo et al., 2019; Jokić et al., 2020; Barišić et al., 2019
	Epicatechin	Methanol/Subcritical Water/Pulsed Electric Field/Water/Hexane	0.00022-0.0568 g/100g	Barbosa-Pereira et al., 2018; Rojo-Poveda et al., 2019; Delgado-Ospina et al., 2020; Jokić et al., 2020; Barišić et al., 2019
	Epicatechin gallate	Methanol/Subcritical Water	0.00027-0.0007 g/100g	Jokić et al., 2020; Barišić et al., 2019
Phytosterol	Stigmasterol	N.D.	4.803 -15.54 g/100g	Agus et al., 2018
	Campsterol	N.D.	0.1224 g/100g	Romancyzk & McClelland , 2004
	Cholesterol	N.D.	1.712 g/100g	Agus et al., 2018
	b-sitosterol	N.D.	0.0078-0.211 g/100g	Agus et al., 2018
Alkaloids	Caffeine	N.D.	0.04-0.42 g/100g	Rojo-Poveda et al., 2019
	Theobromine	N.D.	0.39-1.83 g/100g	Rojo-Poveda et al., 2019
	Theophylline	N.D.	0.0058-0.0188 g/100g	

*values will drastically vary depending on the method of extraction and degree of processing.

Antioxidant Activity Assay	Condition	Value**	Method of CBS extraction	Reference
DPPH activity (%)	In natura	78.5-79.2	N.D.	Lessa et al., 2018
	unfermented	19.38	N.D.	Utami et al., 2016
	fermented*	7.77-81.3	N.D.	Lessa et al., 2018; Utami et al., 2016
	unroasted	14.97	N.D.	Agus et al., 2018
	roasted*	59.3	N.D.	Agus et al., 2018
FRAP activity (mM Trolox/g)	N.D.	0.67-72.32	Ethanol/Methanol- Acetone/Soxhlet/Microwave assisted extraction/N.D.	Lecumberri et al., 2007; Martínez et al., 2012; Mellinas et al., 2020
	Dried	47-84	Pressurized liquid extraction	Okiyama et al., 2018
TEAC (mM Trolox/g)	N.D.	2.56-7.73	Ethanol/Methanol- Acetone/N.D.	Lecumberri et al., 2007; Martínez et al., 2012;

Table 2.5	Antoxidant	activity of	of cocoa	bean	shells a	t various	conditions
-----------	------------	-------------	----------	------	----------	-----------	------------

*duration of fermentation and roasting and pH are variable

**values will drastically vary as a result of the methods of extraction

2.7 Effect of processing on cocoa bean shell by-products

The quality of CBS starting materials is determined by the cacao variety and post-harvesting processing conditions of the cacao beans. Upon harvest, cacao beans undergo fermentation, drying and roasting to favor the retrieval of cocoa nibs, as well as serve as the main modes for cocoa flavor development. These processes, especially fermentation and roasting, impact the overall quality, flavor, and composition of cocoa beans and their by-products, such as CBS.

2.8 Fermentation

Initially, cacao beans are covered in a mucilaginous pulp, namely rich in glucose and saccharose (Anvoh et al., 2009). Spontaneous fermentation is induced by subjecting the pulpcovered beans into an anaerobic environment, where endogenous microorganisms such as yeast, lactic acid bacteria and acetic acid bacteria, easily degrade the pulp exposing the bean (Schwan & Wheals, 2004; Agus et al., 2018). Ultimately, fermentation and drying processes consume the carbohydrate content of cocoa beans, while increasing the crude protein content (Cirkovic Velickovic & Stanic-Vucinic, 2018; Agus et al., 2018; Aremu et al., 1995). Fermentation also promotes the migration of small phenolic compounds from the bean to the shell (Roelofsen, 1958; Kim & Keeney, 1984; Utami et al., 2016). Utami et al. (2016) suggested that partial fermentation (24 hours) resulted in the highest total phenolic content and antioxidant activity in CBS. On the other hand, longer fermentation times surpassing 24 hours led to decreased phenolic content and antioxidant activity due to oxidation caused by polyphenol oxidase, and complexation between phenolic compounds and proteins/peptides (Brito et al., 2001; Camu et al., 2008; Utami et al., 2016). Effects of fermentation on the bioactivity of cocoa proteins have also been studied (Flint et al., 1998; Preza et al., 2010; Addison & Aguilar, 2011; Ryan et al., 2017; Marseglia et al., 2019). Studies suggested that fermentation of cocoa bean enhances in vitro proteolytic hydrolysis of cocoa protein, whereas subsequent processes, such as roasting and chocolate formulation, may reverse such effects (Marseglia et al., 2019; Carbonaro et al., 2000; Giami et al., 2001; Pushparaj & Urooj, 2011; El Hag et al., 2013; Cirkovic Velickovic & Stanic-Vucinic, 2018). In addition, fermentation was shown to positively correlate with the quantity and varieties of peptides of cocoa beans, favor the release of angiotensin converting enzyme inhibitory di- or tri-peptides and increase the antidiabetic, anti-tumor properties of the cocoa products/by-products (Marseglia et al., 2019; Kumari et al., 2018; Preza et al., 2010; Flint et al., 1998; Addison & Aguilar, 2011; Ryan et al., 2017).

These were found to be mediated by the amino acid profile of the fermented cocoa products, as opposed to their antioxidant properties (Preza et al., 2010; Flint et al., 1998; Addison & Aguilar, 2011; Ryan et al., 2017).

Solid-state fermentation is viewed as a method to valorize waste-products, such as CBS, which can be carried out using fungi such as *Aspergillus* (Santos et al., 2011), *Rhizopus* (Santos et al., 2017) and *Penicillium* (Lessa et al., 2018). Lessa et al. (2018) observed that solid-state fermentation of CBS using *Penicillium roqueforti* led to an increase in the phenolic, carotenoid and saponin content, which contributed to the antioxidant properties of CBS (Mioso et al., 2015; Lessa et al., 2018). In addition, Lessa et al. (2018) reported that anthocyanins accounted for 0.4-0.5 mg quercetin equivalent per 100 gram CBS both pre- and post-solid-state fermentation, where its complexing capabilities can be enhanced with increasing pH to bind with pectic and cellulosic materials (Lin et al., 2016; Zhu, 2018). Moreover, a 25% increase in mono-unsaturated fatty acid, specifically in terms of oleic, linoleic and gamma-linolenic acids was observed in CBS upon fermentation (Lessa et al., 2018).

2.9 Roasting

Following fermentation and drying, roasting is initiated and is regarded as the key processing step involved in generating the characteristic and distinctive cocoa flavors and aromas. Cocoa beans are subjected to high temperatures where enzymatic and non-enzymatic, like Maillard reactions are induced due to the low moisture content present in cocoa beans. As a result, compositional changes in the bean and shell vary depending on the roasting temperature, between 120-150°C and typically 20-40 minutes (Pèrez Sira, 2015). Oracz et al. (2014) implied that the fat, tocopherol, phytosterol contents and antioxidant capacity of cocoa beans, from Brazil, Ecuador and Indonesia, are reduced with increased roasting. Another study, conducted by Agus et al. (2018), observed an increase in fat, crude fiber, crude protein, carbohydrate, phenolic and phytosterol content in roasted CBS samples at 140°C for 40 minutes. The discrepancy reported in the literature is due to variations in the roasting condition, including temperature and relative humidity, as well as the type and variety of the cultivar (Oracz et al., 2014). Several authors concluded comparable findings in terms of the protein content upon roasting (Carbonaro et al., 2000; Giami et al., 2001; Pushparaj & Urooj, 2011; El Hag et al., 2013; Cirkovic Velickovic & Stanic-Vucinic, 2018). In fact, the roasting process resulted in decreased protein digestibility in

cocoa beans, however, this can be avoided by establishing a pre-soaking treatment in a sodium bicarbonate solution (Zia-Ur-Rehman & Shah, 2001; Cirkovic Velickovic & Stanic-Vucinic, 2018). Typically, the roasting process is known to decrease the phytochemical content. However, Bitzer et al. (2015) observed that the high-molecular weight and degree of polymerization of pro-anthocyanidins have also shown to prevent loss of gut barrier function and epithelial inflammation. Similarly, Stanley et al. (2018) observed that a roasting condition of 150°C increased catechin and high-molecular weight pro-anthocyanidin concentrations, inducing significant inhibition of pancreatic lipase.

2.10 Extraction methods

In addition to the means of processing applied on cocoa beans and their by-products, various methods of extraction of the bioactive components will also have a significant impact on the composition and functional properties. These chemical, physical (through microwave, pulsed-electric field, subcritical water, ultrasound, hydrodynamic cavitation and pressurized liquid) and enzymatic methods.

2.11 Alkaline/acid-assisted extraction

Chemical treatment, via acid or alkaline solutions, is commonly used for the extraction and characterization of plant cell wall polysaccharides or phenolic compounds occurs by means of disrupting the plant cell wall. Acidic extraction, using tartaric, malic, citric acids and others, is a more commonly performed technique due to its ability to yield commercial pectin with a high degree of esterification; however, this approach may lead to side-chain degradation (Harris & Smith, 2006; Karboune & Khodaei, 2016; Levigne et al., 2002). Alkaline treatment ensures the solubilization of lignin by means of ester-linkage cleavage between the lignin-polysaccharide complexes and the release of phenolic acids (Kondo et al., 1992). Alkaline hydrolysis has been applied on cell wall materials from corn cobs, brewer's spent grain, sugarcane bagasse, and cocoa bean shells for the generation of polysaccharides and phenolic acids (Li et al., 2020; Mussatto et al., 2007; Redgwell et al., 2003; Torre et al., 2008). In the alkaline extraction of corn cobs, Torre et al. (2008) obtained an optimal solubilization of ferulic acid and *p*-coumaric acid at 1171 mg/L and 2156 mg/L, respectively, using 0.5N NaOH treatment for 6 hours, while retaining most of the hemicellulosic and cellulosic fractions. In another study, brewer's spent grain was first subjected to acid hydrolysis, using 72% sulfuric acid, for the removal of hemicelluloses, followed by 2%

NaOH treatment at 120°C for 90 minutes to achieve a release of 145.3 mg/L of ferulic acids and 138.8 mg/L of p-coumaric acids (Mussatto et al., 2007). Furthermore, both studies suggested a positive correlation between phenolic acid release and the NaOH concentration, temperature, and reaction time (Torre et al., 2018; Mussatto et al., 2007). Redgwell et al (2003) investigated a sequential alkaline hydrolysis treatment of cocoa bean shell cell wall material using 0.05M CDTA, 0.05M Na₂CO₃, and 4M KOH paired with 20mM NaBH₄. In the CDTA fraction, a high proportion of soluble pectic polysaccharides rich in uronic acid (87.2%mol) were observed; indicating the presence of a homogalacturonan-type structure. In the Na₂CO₃ fraction, the authors observed high proportions of rhamnose (13.%mol), galactose (14.9%) and uronic acid (66.6%mol), indicating the presence of rhamnogalacturonan. The authors also observed the presence of xylose (23% mol), mannose (9.9%mol), glucose (21.8%mol), and uronic acid (26.2%mol), suggesting the 4M KOH fraction containing a combination of hemicellulose and pectic polysaccharides (Redgwell et al., 2003). Chemical extraction via alkaline/acid hydrolysis was observed to yield higher efficiency in the extraction of phenolic acids than enzymatic approaches using ferulic acid esterases (FAEs) isolated from A. niger, xylanase from T. viride, or commercial enzyme preparations (Bartolomé et al., 1997; Bartolomé et al., 2003; Bartolome & Gomez-Cordoves, 1999; Mussatto et al., 2007). Furthermore, sequential chemical treatments can be applied for the generation of well-defined cell wall polysaccharides for characterization and quantification purposes (Pan et al., 1998; Xu et al., 2005; Li et al., 2020). Mateos-Aparicio et al. (2010) assessed the cell wall material composition of okara, bean pod, and pea pod samples upon sequential treatment using CDTA/ammonium oxalate, 0.05M NaOH, 1M KOH, 4M KOH and NaClO₂/acetic acid. Their findings suggested that okara and bean pod samples are mostly comprised of pectic-rich regions, specifically slightly methylated rhamnogalacturonan substituted with arabinans, galactans, and/or arabinogalactans. On the other hand, the cell wall compositions of pea pods are predominately insoluble fibre fractions, namely cellulose, xylogalacturonan and xylan (Mateos-Aparicio et al., 2010). Moreover, hot water, 2% EDTA and 4M NaOH at 30°C followed by an increase in temperature to 70°C were applied sequentially to hydrolyze coffee (*Coffea arabica L*) leaves to study the effect of salt stress on the cell wall composition. Results showed that salt stress during plant growth resulted in significant increase in negatively charged cell wall polysaccharides and lignin content (de Lima et al., 2014).

2.12 Physical extraction

2.12.1 Microwave-assisted extraction

Microwave-assisted extraction (MAE) was utilized for extracting bioactive compounds from plant cell wall materials such as citrus peels, peanut skins, grape seeds, spent filter coffee, walnut leaves, defatted canola seed cake, brown seaweed, solanum nigrum, cranberry pomace and CBS (Hayat et al., 2009; Ballard et al., 2010; Li et al., 2011; Pavlović et al., 2013; Kaderides et al., 2019; Mellinas et al., 2020; Spadoni Andreani & Karboune, 2020). Microwave energy assists the release of bioactive compounds from plant materials by inducing molecular friction, which triggers the dipolar rotation of polar solvents and the conductive migration of dissolved ions (Eskilsson & Björklund, 2000; Srogi, 2006; Rodriguez-Jasso et al., 2011; Pandit et al., 2015; Mirzadeh et al., 2020). The advantages of MAE include less solvent consumption, shorter extraction times, while improving extraction yields compared to conventional extraction methods, like Soxhlet techniques (Proestos & Komaitis, 2008). The parameters that highly influence the overall extraction yield include the extraction time, solvent type and proportion, as well as microwave power. These parameters are modulated depending on the target compounds of extraction as well as the nature of plant material. For example, Mellinas et al. (2020) observed that the MAE operating conditions of 5 minutes, pH 12, 97°C, and a solid: liquid ratio of 0.04 g/mL resulted in improved polysaccharide (370.5 mg glucose/g) and polyphenol (35.9 GAE/g) extraction yields from cocoa bean shells. Chen et al. (2005) suggested that high extraction yields of Solanum nigrum were achieved at MAE conditions of a 15-minute extraction time, 455W microwave radiation powder, and a 1:20 process ratio. Interestingly, the study observed that extraction times exceeding 15 minutes resulted in decreased polysaccharide extraction yields (Chen et al., 2005). Proestos and Komaitis (2008) noted that acetone is a better extraction solvent than methanol due to its low absorptivity against microwave energy. However, the authors observed comparably high phenolic acid extraction yields using acetone/water (60:40, v/v) and methanol/water (60:40, v/v), ranging between 11.2-23.8 mg gallic acid/g dry sample and 10.3-21.2 mg gallic acid/g dry sample from a variety of plants, namely Styrax officinalis, Origaanum dictamnus, Rosmarinus officinalis, Origanum majorama, Teurcrium polium, Vitex aganus-cactus (Proestos and Komaitis, 2008). Using MAE (855W power, 30 seconds extraction time) on peanut skins (1.5g), a high recovery of phenolic compounds was achieved, measuring 143.6mg gallic acid equivalent/g skins. In addition, ORAC levels significantly increased with a longer extraction time (150 seconds) and higher temperature (Ballard et al., 2010). Kaderides et al. (2019) concluded similar findings with total phenolic content extraction yields from pomegranate peels (199.4 mg GAE/g dry peel), with optimal MAE conditions at 600W, 0.0167 g/mL solvent (50 and 70% aqueous ethanol and methanol) for 4 minutes.

Microwave-assisted extraction (MAE) can also be paired with an alkaline solvent to improve the release of higher molecular weight oligosaccharides by means of loosening the CWP bonds (Khodaei et al., 2016; Müller-Maatsch et al., 2016; Spadoni Andreani & Karboune, 2020). Spadoni Andreani and Karboune (2020) showed that a low microwave power (85 W/g) paired with a high alkaline molarity (0.5M KOH) yielded 20% of pectic oligosaccharides, of homogalacturonan and rhamnogalacturonan origin, from cranberry pomace, which have a higher degree of polymerization (7-10) than those obtained from enzymatic extractions. Alternatively, Davis et al. (2021) recommend the use of high microwave power would favor the release of high molecular weight polysaccharides (200-700kDa). Nevertheless, MAE can be coupled with other analytical techniques such as chromatography to have a more complete picture on the bioactive compounds of plant cell wall materials (Mirzadeh et al., 2020; Xia et al., 2011).

2.12.2 Pulsed electric field

Pulsed electric field (PEF) treatment is an emerging non-thermal extraction technique involving the application of moderate to high voltage pulses for a short period of time. PEF is currently commercially available for food processing and has the potential to revalorize agroindustrial by-products, such as CBS, for the development of novel products or ingredients (Poojary et al., 2017). Barbosa-Pereira et al. (2018) explored the extraction yield of CBS polyphenols using PEF treatment and observed increases in the extraction of phenolics and epicatechin up to 1.8-19.5% and 30% from CBS compared to supercritical carbon dioxide extraction (Arlorio et al., 2005) and solid-liquid extraction (Martínez et al., 2012). Furthermore, the authors highlighted that higher recovery and radical scavenging activity of flavonoids from CBS were achieved, compared to other thermal treatments. In agreement with Barbosa-Pereira et al. (2018), other studies reported PEF as an efficient method to extract certain polyphenols, specifically for anthocyanins (Luengo et al., 2013; Medina-Meza & Barbosa-Cánovas, 2015; Puértolas & Barba, 2016). The factors influencing the PEF extraction yield include the extraction time, solvent concentration, as well as the material's compound characteristics (Puértolas & Barba, 2016; Barbosa-Pereira et al., 2018).

2.12.3 Sub-critical water extraction

Subcritical water extraction (SWE) is another environmentally friendly extraction method. This technique uses water as the solvent along with the application of high pressure and temperature to extract both polar and non-polar compounds (Muñoz-Almagro et al., 2019). In theory, SWE would provide a high recovery for active compounds, such as phenols. SWE (121°C, 103.4 bar, 30 minutes) with and without supercritical fluid extraction resulted in higher pectic yields and purity (10.9%) from cacao pod husks compared to conventional extraction, using 4% (w/v) citric acid (8.3%) (Muñoz-Almagro et al., 2019). This was explained by high reactivity of the ions generated as a consequence of water dissociation under subcritical conditions (Muñoz-Almagro et al., 2019). SWE was further coupled with micro-encapsulation and spray-drying techniques to ensure the high yield and stability of the bioactive phytochemicals in the CBS extracts (Jokić et al., 2020). Using response surface methodology, Getachew et al. (2018) and Tomšik et al. (2017) investigated the optimal SWE operating conditions to mediate a high polysaccharide and phenolic extraction yield from spent coffee grounds and wild garlic samples. Getachew et al. (2018) suggested SWE conditions set at 178.85°C, 20 bar for 5 minutes would provide a high polysaccharide, namely galactomannans, recovery, (18.25%), while Tomšik et al. (2017) suggested that optimal phenolic recovery (4.11 g gallic acid equivalent/100 g) would be achieved at 179.0°C for 10 minutes with a 1.09% HCl acidifier.

2.12.4 Cavitation extraction techniques

2.12.4.1 Ultrasound-assisted extraction

Following the principles of acoustic cavitation, ultrasound-assisted extraction (UAE) will ensure cell wall disruption by applying low-frequency, high-intensity ultrasonic waves (5-400W), providing the release of phenolic compounds, oligo- and polysaccharides from the plant material (Yusof et al., 2019; Pavlović et al., 2019; Grillo et al., 2019; Leong et al., 2011). Grillo et al. (2019) compared the extraction efficiency of UAE on CBS starting materials pre-treated with a ternary mixture of hexane/ethanol/water (30:49:21). With UAE (15 min, 150W, 19.9kHz, 40°C) yielded higher proportions of hexane (2.5%) and hydro-alcoholic phases (13.9%). Similarly, Yusof et al. (2019) reported UAE (55°C, 45min) coupled with an 80% ethanol pre-treatment yielded a maximum level of flavonoids (7.41mg rutin equivalent / g) in CBS extracts. In terms of the extraction efficiency of alkaloids, Pavlović et al. (2019) proposed UAE (30 min, 35kHz, 25-60°C)

using 50% aqueous ethanol is ideal for caffeine recovery (994.60 mg/kg) while providing a 91.87% DPPH radical scavenging activity from CBS extracts.

2.12.4.2 Hydrodynamic cavitation

Hydrodynamic cavitation (HC) is a low-cost, yet advanced extraction method in situations where industrial scale up is desired. Also paired with a pre-treatment, this technique generates cavitation by producing high pressure turbulences that is reduced below liquid vapor pressure, further, disrupting the cell wall matrix, increasing mass transfer and providing better contact between reactants (Rinaldi et al., 2017). Similar to UAE, HC extraction has been explored for by-product valorisation, such as in CBS (Grillo et al., 2019). Using hexane/ethanol/water (30:49:21) for CBS pre-treatment, HC (3000 rpm, 11 min, 25°C) yielded greater proportions of hydrophilic extracts (14.6% vs 13.9%) and lipid fractions (10.1% vs 2.5%) than UAE. Furthermore, HC provided a two-fold increase in theobromine recovery than UAE (16.02% vs 9.14% w/w) (Grillo et al., 2019). Comparably, Lohani et al. (2016) explored the extraction of sorghum flour and apple pomace using hydrodynamic cavitation, in which increases in their total phenolic content (39.5% and 42%) and antioxidant activity (38.6% and 97%) were observed. HC has shown its ability to extract protein from soybeans at much greater yields compared to acoustic cavitation, through UAE (Preece et al., 2017).

2.12.5 Pressurized-liquid extraction

Heavily utilized for its successful polyphenol recovery, pressurized-liquid extraction (PLE) involves the use of liquid solvents, namely ethanol or water, at a high temperature and pressures for a short period of time (Mustafa & Turner, 2011; Colivet et al., 2016; Okiyama et al., 2018; Fernández-Ponce et al., 2016). Okiyama et al. (2018) explored the effects of PLE ($60-90^{\circ}$ C, 10.35MPa, 5-50 minutes, EtOH), on the total flavonol and alkaloid content in CBS extracts. Their findings suggested that the catechin ($26-135 \mu g$ flavonols/g CBS) and epicatechin ($1.36-3.26 \mu g$ flavonols/g CBS) concentrations increased with increasing temperature and static time. Similar observations were made with the caffeine and theobromine concentrations where equilibrium was achieved after a 30-minute static time (Okiyama et al., 2018). Jacques et al. (2008) suggests that elevated temperatures promote the release of high molecular weight compounds, however, the static time may negatively influence the extraction yield. This was observed for CBS' procyanidin B2 concentrations as they decreased with increasing static time ($0.973 \mu g$ flavonol/g CBS at 30

minutes vs 0.73µg flavonols/ g CBS at 50 minutes) (Okiyama et al., 2018). Another study conducted by Mazzutti et al. (2018) compared PLE (70°C, 10MPa, 20 minutes, EtOH) to Soxhlet extraction (EtOH) to recover polyphenols from CBS. Both, techniques provided similar extraction yields (PLE: 7.2% vs Soxhlet: 6.8%), however, PLE yielded a lower total phenolic content (0.6 mg gallic acid equivalent/g vs 49.4 mg gallic acid equivalent/g) than Soxhlet, which was a result of using a non-defatted CBS sample as well as a short PLE static time (Mazzutti et al., 2018). Nevertheless, PLE ensures less solvent consumption, reduced extraction time, improved solation of target compounds, namely phenolic compounds, lignin, and carotenoids (Mazzutti et al., 2018; Okiyama et al., 2018; Mustafa & Turner, 2011).

2.13 Enzymatic-based approaches

Plant by-product valorization using enzymatic-based approaches involve disrupting the cell wall matrix by means of breaking down lignocellulosic material while supporting the release and recovery of selected bioactive compounds. In plant cell wall structures, the lignin is tightly bound around cellulose fractions via phenolic-glycoside or di-ferulic covalent linkages that forms a physical barrier to prevent glycoside hydrolase activity. Gartaula et al. (2019) used ferulic acid esterase from *Clostridium thermocellum* to release trans-ferulic acid, 5-5' diferulic acid and 8-O-4' diferulic acid from popped endosperm and wheat cell walls. The synergistic effects of FAEs and different glycosyl hydrolases were often evaluated in terms of cell wall disruption and the release of phenolic compounds. The works of de Oliveira et al. (2016) observed the use of ferulic acid esterase (40 µg/mL), from Aspergillus clavatus, in combination with xylanase (25 U/mL) which contributed to a significant release of ferulic acid (0.55 mg/g) using the enzyme cocktail compared to FAE (0.1 mg/g) and xylanase (0.03 mg/g) alone, as a result of xylanases producing short chain feruloylated xylooligosaccharides with less steric hindrance (Vardakou et al., 2004; de Oliveira et al., 2016). Enzymatic hydrolysis of natural sugar cane fractions was investigated by Várnai et al. (2014) using xylanases, ß-xylosidase, acetyl xylan esterase, and three feruloyl esterases (AnFaeA, TsFaeC, and NcFaeD, respectively) in cellulase mixtures. Acetyl xylan esterase (AXE) improved the yield of cellulose and xylan hydrolysis by removing substitutions and enhancing the accessibility of cellulose/xylan structure (Tenkanen et al., 1997; Várnai et al., 2011). In terms of the FAE-cellulase mixtures, the TsFaeC-cellulase mixture enabled the release of *p*-coumaric acid (20.9% in sugar cane pith) while the AnFaeA and NcFaeD mixtures effectively released ferulic acid (18.3% from sugar cane pith). Both studies observed the synergistic effects of FAE and cellulase where the extraction product profile of phenolic acids is related to type of FAEs (Gottschalk et al., 2010; Várnai et al., 2014). Alternatively, Gottschalk et al. (2010) assessed the enzymatic efficiency of fungal biomass *Trichoderma reesei* and *Aspergillus awamori* in the hydrolysis of steam pre-treated sugarcane bagasse. In *T. reesei* biomass, carboxymethyl-cellulase and exoglucanase activities were predominant, while xylanase and β-glucosidase activities were predominant in that of *A. awamori*. Their blend resulted in synergistic enhancement of exoglucanase and carboxymethyl-cellulase activities by 73-110% and promoted the hydrolysis of celluloses and xylans to over 80%. The enhanced effectiveness of the cellulase and xylanase blend was attributed to the cleavage between xylan and lignin, further creating more hydrolyzing sites in glucan/xylan (Gottschalk et al., 2010).

Treatments with glycosyl hydrolase at controlled conditions were developed for the generation of polysaccharides with potential functionalities from cell wall materials. In order to effectively release polysaccharides from CBS, enzymatic treatment was carried out with a cocktail comprised of Pentopan®MonoBG, Viscozyme® L and Pectinex® 5XL and the total sugar release was attained at 47.1-50.7g/kg (dry matter). The authors recommended enzyme concentrations of 0.6%, 0.8% and 1% (w/w) would enable optimal polysaccharide release from CBS (Alemawor et al., 2009). The extraction of pectin from lime peels using Laminex C2K from Penicillum funiculosum (cellulase, xylanase, and arabinoxylanse) was reported to be around 23% (w/w) (Dominiak et al., 2014). Other studies have suggested enzymatic extraction at pH conditions of 4-7 with assistance using organic solvents, acids or hot water extraction for improved efficiency in extracting high molecular weight cell wall polysaccharides (Wu et al., 2013; Kilpeläinen et al., 2014; Reisinger et al., 2014; Martínez-Abad et al., 2016). Using a multi-enzymatic approach to yield a variety of low and high oligosaccharides and oligomers, Khodaei and Karboune (2018), observed that equal proportions of Depol[™] 670L and Gamanase 1.5L reacted with potato rhamnogalacturonan I (RGI), both chosen for its high RG I specificity, attained yields of low (0.1-13.9%) to high molecular-weight oligosaccharides (0.0-37.5%) as well as oligomers (0.0-75.7%).

2.14 Cell wall feruloylated oligo- and polysaccharides: biotransformation and functionalities

Feruloylated oligosaccharides, a hydroxycinnamic acid derivative, exist in nature within the plant cell wall. Typically, their recovery from the plant cell wall is limited due to structural inconsistencies as well as the prevalence of ferulic acid in the phenolated moiety (Tamayo-Cabezas & Karboune, 2019). In order to better exploit the plant cell wall extracts, different approaches have been developed to achieve desired functionality profiles via the modulation of the structural and physicochemical properties of the main compounds. Potential modifications of cell wall feruloylated oligo- and polysaccharides are highlighted in this section to support the development of multi-functional and bioactive ingredients derived from CBS.

2.14.1 Hydrolysis of cell wall polysaccharides for the generation of feruloylated oligosaccharides

Acid hydrolysis can serve as a technique to recover feruloylated oligosaccharides. Previously, successful extraction of feruloylated oligosaccharides were recovered from maize bran (Allerdings et al., 2005; Saulnier et al., 1995), sugar beet pulp (Ralet et al., 1994; Ishii, 1997) and rice bran (Li et al., 2008) using a mild acid hydrolytic treatment with trifluoroacetic acid (TFA). Furthermore, a food-grade treatment, using oxalic acid, was also investigated for the extraction of feruloylated oligosaccharides from maize bran, which increased the production of feruloylated oligosaccharides as 52.2% of ferulic acid was released (Ou & Sun, 2014; Huang & Ou, 2012). Response-surface modeling suggested that optimal acid hydrolysis conditions for rice bran were at 193mM TFA, 1.36 hours hydrolysis time at 100 °C, in which 916.12µg feruloylated oligosaccharides/g were recovered, accounting for 54.08% of the total acyl ferulic content (Li et al., 2008). Alternatively, Ralet et al. (1994) suggested that high temperatures or a two-step TFA hydrolytic treatment would support a more complete hydrolytic approach, recovering 90-95% of feruloylated oligosaccharides. The product from acid hydrolysis derived from sugar beet pulp was characterized by its enriched arabinose-type sidechains, attributing to high susceptibility of the regions with arabinose and their α -(1 \rightarrow 5) linkages towards acid hydrolysis than other regions (Ralet et al. 1994). The ferulic acid-glycoside linkages from cell wall materials of maize bran were identified to be 5-O-(trans-feruloyl)-L-arabinose f, O-B-D-xylose $p-(1 \rightarrow 2)$ -[5-O-(trans-feruloyl)-Larabinose-f/ and O-L-Gal- $p(1 \rightarrow 4)$ -O-D-xyl- $p-(1 \rightarrow 2)$ -[5-O-(trans feruloyl)-L-Ara f/, by nuclear

magnetic resonance spectroscopy (Saulnier et al.,1995). The feruloyl groups were found to be linked to the arabino-furanosyl residues of the main core α -(1 \rightarrow 5)-linked arabinan chains and to the galacto-pyranosyl residues of the main core of β -(1 \rightarrow 4)-linked type 1 galactan chains of the pectic polysaccharide of cell wall materials (Colquhoun et al., 1994; Fry, 1982; Ralet et al., 1994). As compared to acid hydrolysis, enzymatic hydrolysis has proven to recover higher yields in feruloylated oligo- and polysaccharides (Ralet et al., 1994).

2.15 Feruloylation of cell wall oligo- and polysaccharides

Efforts to enhance the degree of feruloylation of cell wall oligo and polysaccharides have been achieved using an enzymatic approach. Successful transesterification of methyl ferulates to Larabinose was mediated using type C FAEs from Sporotrichum thermophile (StFaeC), feruloyl esterase paired with xylanase (Topakas et al., 2005; Vafiadi et al., 2007). The authors suggested the modified glycosides are biocompatible, biodegradable as well as exhibit anti-microbial activity (Topakas et al., 2005). Electrospray ionization mass spectrometry (ESI-MS) was used to characterized the chemoenzymatic synthesized feruloylated arabino-oligosaccharide; the results suggested that feruloylation occurred on the hydroxyl group of thenon-reducing ring of each Larabino-oligosaccharide (Vafiadi et al. 2007). Several studies have reported increases in bioconversion yields using surfactant-less microemulsions for enzymatic synthesis of feruloylated (Couto et al., 2010; Coutu et al., 2011; Topakas et al., 2005; Vafiadi et al., 2006). Couto et al. (2011) used a n-hexane/1,4-dioxane/MES-NaOH buffer mixture, where a 26.8% feruloylated galactobiose bioconversion yield was achieved. Similarly, a n-hexane/2-butanone/MES-NaOH buffer mixture, resulted in feruloylated sugars of raffinose, arabinobiose, xylobiose bioconversions yield were achieved at 11.0%, 7.9% and 9.0% (Couto et al., 2011). The microemulsion composition will impact the overall esterification efficiency of FAE, where it is highly dependent on the structure of the glycoside substrate (Couto et al., 2011). Moreover, Couto et al. (2011) suggested ideal bioconversion yields would be achieved at 35°C, with a 3:1 ferulic acid to raffinose molar ratio, 3% water content (v/v) as well as an enzyme concentration of 345 FAE units.

Furthermore, enzyme immobilization can serve as commercial application of the biosynthesis of feruloylated oligosaccharides, through optimizing enzyme activity at specific macroenvironment and ensuring enzyme stability and reusability (Thörn et al., 2011; He et al., 2015; Tamayo-Cabezas & Karboune, 2019). Using FAE, from *Humicola insolens*, (Tamayo-

Cabezas & Karboune, 2020) suggest that the pH, buffer molarity, and the enzyme-to-support ratio served as the key parameters to modulate the immobilization activity yield to mediate the bioconversion of feruloylated oligosaccharides. By means of modulating the optimal conditions, the authors reported that as much as 92% of the feruloylation efficiency was retained upon immobilization of FAE, from *H. insolens* (Tamayo-Cabezas & Karboune, 2020). Similarly, Spadoni Andreani et al. (2021) observed as much as a six-fold increase in the feruloylation efficiency of cranberry polysaccharides using immobilized FAE, from *H. insolens*; as compared to free FAE. Additionally, this yielded feruloylated polysaccharides with a greater phenolic content (Spadoni Andreani et al., 2021). Other works explored similar effects of immobilization on the feruloylated oligosaccharide bioconversion yield (Couto et al., 2011; He et al., 2015; Tamayo-Cabezas & Karboune, 2019; Thörn et al., 2011; Topakas et al., 2005).

2.16 Functional properties of cell wall feruloylated oligosaccharides

Feruloylated oligosaccharides are functionally bioactive compounds that have shown to promote antioxidant (Couto et al., 2011; Fang et al., 2012), anticarcinogenic, antimicrobial and anti-inflammatory activities (Ou & Kwok, 2004; Couto et al., 2010), as well as promoting microflora proliferation and pathogen suppression in the gastrointestinal tract (Ou et al., 2016).

2.16.1 Antioxidant activity

Ferulic acid glycoside esters were shown to be stronger antioxidants than their free forms, as indicated by radical scavenging activity against DPPH (Kylli et al., 2008). Couto et al. (2010) suggested this was attributed to the enhanced ability of ferulic acid to donate hydrogen to DPPH and to stabilize the phenoxy radical upon its conformational modification with di- and oligosaccharides. Similar findings were observed with feruloylated oligosaccharides from maize bran, wheat bran, wheat flour, and barley; providing improved radical scavenging activities compared to free ferulic acid (Ohta et al., 1994; Katapodis et al., 2003; Szwajgier et al., 2005; Zhang et al., 2005; Shen et al., 2012; Lin et al., 2014). In contrast, feruloylated xylooligosaccharides and arabino-xylooligosaccharides showed less radical scavenging specificity towards DPPH compared to free ferulic acid (Couto et al., 2011; Katapodis et al., 2003). However, feruloylated xylo-oligosaccharides were viewed as " the most potent scavenger" as it retained a radical scavenging yield of 94% (Couto et al., 2011). The study also reported that the antioxidant activity was greater for acylated ferulic acid with hexoses than pentoses (Couto et al., 2011). Ou

and Sun (2014) suggests that the antioxidant activity of feruloylated oligosaccharides is related to the structure, and ferulic acid content. Silva et al. (2000) proposed that the steric hindrance of the glycosidic substituents on the rotation degree of the phenyl moiety can impact the scavenging activity.

Chigorimbo-Murefu et al. (2009) stated that lipase catalyzed feruloylated arbutin ferulates (23%) obtained from lipase-catalyzed reactions showed enhanced capability in inhibiting LDL oxidation as compared to ferulic acid (16%) and arbutin (13%) alone. Ou and Sun (2014) investigated the feruloylated oligosaccharides antioxidant activity in human erythrocytes which suggested that feruloylated oligosaccharides prevented hydrogen peroxide cell damage (Ou and Sun, 2014). In addition, no signs of cytotoxicity or genotoxicity were observed after this intervention against hydrogen peroxide exposure (Wang et al., 2008; Zhang et al., 2005; Shen et al., 2012). Other biological property of feruloylated oligosaccharides associated with their antioxidant activity were reported, including increased the cell viability, superoxide dismutase activity, decreasing the release of lactate dehydrogenase and malondialdehyde and inhibiting glycation reaction (Boušová et al., 2005; Shen et al., 2012; Silván et al., 2011; Vlassara & Palace, 2002; Wang et al., 2009; Wu et al., 2009).

2.16.2 Prebiotic potential

Similarly, to non-digestible oligosaccharides, feruloylated oligosaccharides may serve as prebiotics to enhance the intestinal microbiome. Feruloylated oligosaccharides extracted from wheat bran have shown to stimulate the growth of *Bifidobacterium bifidum* (Yuan et al., 2005). However, these carbohydrates showed lower fermentability in the colon compared to xylooligosaccharides. In order to improve the rate of fermentation, Ou and Sun (2014) suggested ferulic acid to be released from oligosaccharides via microbial ferulic acid esterases to increase their bioavailability.

2.16.3 Immunomodulatory activity

Since feruloylated oligosaccharides were deemed as viable antioxidants, one may investigate their impact on immune responses. *In vitro* analysis with feruloylated oligosaccharides extracted from rice bran suggested the suppression of pro-inflammatory cytokines such as tumor necrosis factor alpha (TNF- α), interleukin-1 β , interleukin-6, nitric oxide, and prostaglandin E2 (Fang et

al., 2012). As a result, Fang et al. (2012) concluded these findings are the first signs of the immunomodulatory effects associated with feruloylated oligosaccharides. Therefore, one could correlate the anti-inflammatory response of feruloylated oligosaccharides to that of β -glucan, arabinogalactan, pectin, arabinan, heteroglycan, inulin, agaro-oligosaccharides, chitosan oligosaccharides as well as ferulic acid (Hirabayashi et al., 1995; Sakai et al., 1999; Chen & Yan, 2005; Yamada & Kiyohara, 2007; Yoon et al., 2007)

2.17 Conclusion

CBS are de-hulled from cocoa beans during processing, later discarded as waste or used in animal feed and fertilizer. Nonetheless, chemical and bioactive composition analysis studies conclude an abundant source of dietary fiber and phenolic compounds, providing functional properties that can serve use in food formulations. Many techniques that are developed for the valorization of plant cell wall materials can be adapted towards cocoa bean shells to support its exploitation. This review explored the effects of cocoa processing, especially fermentation and roasting, where observed differences were found in terms of chemical and bioactive compositions. Physical, chemical and enzymatic methods in bioactive component extraction from the cell wall matrix of CBS were discussed. Depending on the desired target compound, either individual approaches or their combined uses can be explored. Future research should focus on the isolation, bio-generation and potential formulation of feruloylated oligo- and polysaccharides derived from CBS. These bio-products will be analyzed for the functional properties that can be used in food formulations such as a sugar alternative in chocolate manufacturing. Although non-specific to CBS, studies were compiled to investigate efforts done to modify or extract feruloylated oligosaccharides from the cell wall material. Provided that CBS has both a rich phenolic and cell wall polysaccharide content, the potential to generate feruloylated oligosaccharides is promising where one can further investigate its bioactive properties of cell wall polysaccharides, such as its prebiotic, antioxidant, and immunomodulatory activity. With this, future research should assess the use of CBS-derived feruloylated oligosaccharides in food formulations and their technofunctional property as well as investigate health-related functional properties.

Table 2.6. Applications of cocoa bean shells among several industries	
---	--

Field	Use	Application	Outcome	Reference
Agriculture		Barn bedding material	Lower bacterial counts and plasma cortisol levels as well as decreased ammonia concentrations from 6-21% were observed using CBS bedding for lactating Holstein cows to promote udder health	Yajima et al., 2016
	Swine intestinal microbiota	Prebiotic	7.5% CBS substitution with conventional cereal-based diets led to a decrease in <i>Firmicutes</i> and an increase in <i>Bacteroidetes</i> in pigs. Also, reduced fecal populations of <i>Lactobacillus-Enterococcus</i> and <i>Clostridium histolyticum</i> were observed suggesting CBS' potential for stabilizing swine intestinal microbiota.	Magistrelli et al., 2016
Feedstuff	Cockerel Feed	Protein and fibre supplement	a 10% CBS supplementation in Cockerel feed led to a decrease in packed cell volume, haemoglobin and white blood cell counts as well as increased Broiler weights	Adeyina et al., 2011
	Goat Feed	Fibre supplement	CBS supplementation with Dried Brewer's grain led to a 5.3kg increase among goats over 56 days	Aregheore, 2002
	Laying hens feed	Feed supplement	results in an increase in spleen, kidney and ovary weights as well as an increase in egg production per	Emiola et al., 2011
	Laying hens feed	Feed supplement	Enzyme-treated CBS supplementation (10%) in replacement of maize was effectively incorporation in hens' diets without adverse effects on the performance and egg quality characteristics.	Olumide et al., 2014
	Rabbit feed	Feed supplement	Hot water treated CBS (200g/kg) in rabbit feed increased feed intake and led to significant weight gain and optimal performance.	Adeyina et al., 2010
	Cattle feed	Feed supplement	60% rice bran with 40% CBS as a protein supplement resulted in optimal daily weight gain and feed efficiency among post-weaning cattle than rice bran supplementation alone.	Soeharsono et al.,2017
Food	Biscuits	Microencapsulated polyphenol extract	Maltodextrin (stabilizing agent)-microencapsulated CBS polyphenol (80:20) provided a stable powder without affecting the total polyphenol content during the baking process.	Papillo et al., 2019

Gluten-free bread	Fibre-enrichment and textural enhancement during storage	Fibrous material led to a coarser structure, reduced specific volume and poor moisture absorption. Also, led to a desirable color as well as a crumb and crust texture (hardness) CBS polyphenol extracts contributed to the high entire ident extracts and shake shared data in history	Rinaldi et al., 2020
Functional Beverage	Antioxidant and anti-diabetic agents	capacity in the home-made functional beverage while achieving consumer desirability in terms of appearance and odor.	Rojo-Poveda et al., 2019
Pork Sausages	Color enhancer, stabilizer, textural enhancer	Pork sausages supplemented with CBS (0.75-1%) showed ideal emuslion stability and increased apparent viscosity, color enhancement (less redness), and flavor acceptability. Also, CBS improved refrigerated storage properties through prolonging the onset of lipid oxidation.	
Cookies	Fiber enriched flour substitute	Supplementing wheat flour with 20% of CBS flour with led to high acceptability in terms of texture, taste, crunchiness, hardness, and after-taste 20% soluble cocce fibre replaced with oil showed	Handojo et al., 2019
Chocolate muffins	Fat replacer	similar textural (cohesiveness and chewiness) properties while imparting a nice color compared to the control,	Martínez-Cervera et al., 2011
Cooked Beef	Antioxidant enrichment	CBS extracts reduced lipid oxidation as a result of its high antioxidant and radical scavenging activity as compared to synthetic antioxidants (BHT and β - tocopherol) for cooked beef stored at 4°C for 14 days.	Ismail & Yee, 2006
Extra virgin olive oil jam	Antioxidant enrichment	Encapsulated CBS extract prolonged onset of rancidity	Hernández-Hernández et al.,2019
Dairy drink	Antioxidant enrichment	Dairy drink using CBS, coffee silverskin and orange peels attained a desirable antioxidant activity (against DPPH) by 82.20% and a total phenolic content of 5.74 gallic acid equivalent mg/g while retaining consumer acceptability.	Quijano-Aviles et al., 2016
Yoghurt Drink	Polyphenol and antioxidant enrichment	Encapsulated CBS phenolic extract <i>in-vitro</i> bioaccessbility was 5-fold higher in the liposomal powder where catechin and ferulic acid bioaccessibility accounted for 50 and 80% during 15 days of storage	Altin et al., 2018
Extruded snacks	Protein and fibre fortifier	Extruded snacks using corn flips and CBS achieved harder textures with increased retrogradation	Jozinović et al., 2019

			tendencies while increasing the proportions of resistant starch, polyphenols, and antioxidant activity	
	Pound cake	Fat replacer	50% vegetable oil replacement with leached or raw CBS in pound cake formulation improved the chemical, physical, and sensory properties compared to the control	Öztürk & Ova, 2018
Packaging		Antioxidant bioelastomer	Active packaging material (CBS + PDMS) exhibited good moisture barrier properties and optimal antioxidant activity against DPPH and ABTS Biocomposite films of polylactic acid and CBS	Tran et al., 2017
	Biocomposite packaging films		produced showing increases in Young's modulus by 80% as well as exhibiting moisture-barrier, antioxidant and swelling properties that allow these films to be biodegradable in aquatic environments	Papadopoulou et al., 2019
	Biocomposite packaging films	Biomass ash partial	Polycaprolactone and CBS (30%) composites produced and exhibited enhanced rigidity as well as optimal composite biodegradability CBS biomass ash led to decreased water absorption	Puglia et al., 2016
Industrial	Concrete	concrete replacement	porosity and capillary indexes, resulting in increased concrete mechanical strength and durability.	Fontes et al., 2019
		Bioadsorbent	Grafted CBS with diazonium salts experienced increased chemical and mechanical resistance that would facilitate as a bioadsorbent to entrap pollutants such as heavy metals, gases, or industrial dyes.	Fioresi et al., 2017
		Bioadsorbent	Plasma-treated CBS effectively removed Azur II and Reactive Red dyes in the basic and acidic conditions after 40 and 240 min serving as a multi-purpose sorbent for the removal of coexisting pollutants from aqueous solutions	Takam et al., 2017
		Bioadsorbent	Developed active carbons with high meso- and microporosity	Pérez-Cadenas et al., 2018
		Bioadsorbent	Developed binderless carbon monoliths with a high micropore volume and good mechanical performance	Plaza-Recobert et al., 2017
	Biofuel	Liquid smoke and charcoal production	CBS pyrolysis (400-550°C) achieved a liquid smoke (18-23%) and charcoal (39%) while showing faster heating rates for the production of ash and water contents of 16.5-19% and 6.5-8.5%	Handojo et al., 2020

Biofuel	Biogas production	A 14% increase in the biomethane yield was achieved using pre-treated CBS as a source of lignocellulosic material.	Mancini et al., 2016
Biofuel	Bioethanol production	An 8.46% bioethanol yield was achieved using hydrolyzed CBS functioning as biomass alternative	Awolu & Oyetuji, 2015
Automotive brake pads	Composite filler additive	Observed increases in the wear rate, tensile and compressive strength as well as increases in the coefficient of friction using CBS as an asbestos filler replacement	Olabisi et al., 2016
	Bioadsorbent	Using CBS as an activated-carbon precursor where Methylene Blue dye was adsorbed and controlled by film diffusion	Ahmad et al., 2012
	Bioadsorbent	CBS (20g) with coffee grind residues (10g) effectively lead concentrations in contaminated waters 98.53%	Diaz et al., 2018

CONNECTING STATEMENT I

This review is an accepted manuscript of an article published by Taylor & Francis in CRITICAL REVIEWS IN FOOD SCIENCE AND NUTRITION on April 25th, 2022, available at https://www.tandfonline.com/doi/full/10.1080/10408398.2022.2065659

In the literature review, presented in Chapter II, the structural characterization, composition and functional properties of cocoa bean shells are discussed. The characterization of cocoa bean shells composition and the corresponding cell wall material, together with the assessment of the isolation of polysaccharides using alkaline treatment are reported in Chapter III. Proximate analysis, dietary fiber, and polyphenol quantification were performed on cocoa bean shells and their cell wall material. The monosaccharide profile of the cell wall material was determined by high performance anion exchange chromatography. Alkaline treatment, using two concentrations, was applied to cocoa bean shells cell wall material, for the isolation of their polysaccharides, which were characterized and compared in terms of the molecular weight and composition.

CHAPTER III. EXTRACTION AND CHARACTERIZATION OF COCOA BEAN SHELL CELL WALL POLYSACCHARIDES

3.1 Abstract

Cocoa bean shells are the outermost layer of the cocoa bean and a heavily accumulated byproduct of the cocoa industry. In this study, cocoa bean shells were characterized in terms of their chemical composition where they were found to be a great source of carbohydrates, specifically dietary fiber, protein, ash, and polyphenols, namely quercetin, epicatechin, and catechin. Cell wall polysaccharides were isolated through sequential extraction (ethanol, chloroform:methanol, and phenol:acetic acid:water) followed by alkaline extraction (0.5M or 4M KOH) for their characterization. Polysaccharide extracts were found to be primarily rich in pectic polysaccharides (80.6-86%) namely rhamnogalacturonan and arabinogalactan as well as hemi-cellulosic polysaccharides (13.9-19.4%). Overall, 0.5M KOH polysaccharides were favored having provided a diverse profile of neutral sugars and uronic acids. When tested for the promotion of the growth of selected probiotic strains, CBS cell wall polysaccharides performed similarly or more than inulin and rhamnogalacturonan based on the prebiotic activity scores. The short chain fatty acid profiles were characterized by high amounts of lactic acid, followed by acetic and propionic acid.

3.2 Introduction

The generation of Cacao (*Theobroma cacao*) by-products are estimated at 700,000 tons annually, which negatively contribute to the environmental and economic value of cocoa processing industry. Among these by-products, cocoa bean shells (CBS) are the most abundant and are the outermost layer of the cacao bean that encases cacao nibs. Efforts to valorise CBS have been explored in agricultural and bioremediation fields such as its use in animal feeds and fertilizers (Adeyina et al., 2010; Soeharsono et al., 2017). In food applications, CBS has been used as a functional ingredient for fiber enrichment and fat replacement in baked goods and as a stabilizer and color enhancer in pork sausages (Bernaert & Ruysscher, 2016; Choi et al., 2019; Martínez-Cervera et al., 2011). Although there has been an increasing industrial interest in effective valorization of CBS into functional food ingredients, only the pilot-scale extraction of polyphenols from CBS has been explored (Grillo et al., 2019). The extraction of cell wall polysaccharides from CBS has yet to be reported. To the authors' knowledge, only few studies have characterized the cell wall polysaccharide profile of CBS in terms of monosaccharide profile, molecular weight distribution and linkage types (Lecumberri et al., 2007; Redgwell et al., 2003; Redgwell & Hansen, 2000; Vojvodić et al., 2016). According to Redgwell et al. (2003), CBS holds a reliable source of dietary fibers as its cell-wall polysaccharide profile comprises of 45% pectic polysaccharides, 35% cellulose as well as 20% hemicelluloses, namely gluconoarabinoxylan, xyloglucan, and galactoglucomannans. As far as the authors are aware, the extent of the variability in the cell wall polysaccharide profile of CBS, depending on the cacao variety and the processing conditions, has not been studied.

Several approaches have been reported for the isolation of cell wall polysaccharides from cell wall, including chelating agent, acid, microwave-assisted alkaline, and ultrasound-assisted alkaline extractions (Aguiló-Aguayo et al., 2017; Khodaei & Karboune, 2013; Redgwell et al., 2003; Spadoni Andreani & Karboune, 2020). The selection of an appropriate extracting agent or approach can help target the isolation of polysaccharides with well-defined structures. As an example, acidic extraction, using hydrochloric acid, targets the isolation of homogalacturonan (Khodaei & Karboune, 2013). Alternatively, alkaline extraction of cell wall polysaccharides, using sodium or potassium hydroxides paired with sodium borohydride, has proven to be effective for isolating and understanding the distribution of pectic and hemi-cellulosic polysaccharides within the cell wall materials of potato, cranberries, bamboo, maize and more (Khodaei & Karboune, 2013; Li et al., 2015; Spadoni Andreani et al., 2021; Zhu et al., 2022). This approach enables the release of rhamnogalacturonan I through β-elimination and oxidative peeling of homogalacturonan (Bonnina et al., 2001; Khodaei & Karboune, 2013; Zykwinska et al., 2006).

There is an increasing interest in the cell wall-rich by-products as their polysaccharide content can act as a beneficial source of dietary fibers and prebiotics that can modulate the gut microbiota and hence promote the intestinal health (Spadoni Andreani et al., 2021). Cell wall polysaccharides of by-products derived from pearl millet fibre, tangerine peels, apple pomace, and coconut residues have been previously reported to support the effective growth of beneficial gut microorganisms (Farooq et al., 2017; Islamova et al., 2017; Mohd Nor et al., 2017; Spadoni Andreani et al., 2021). However, no study on the fermentability of CBS cell wall polysaccharides by beneficial lactic acid bacteria has been reported so far.

The aim of the present study was to assess the nutritional composition of CBS at various processing conditions. To the authors' knowledge, this has not been previously investigated for

cocoa beans. In addition, the present study offers novel insight pertaining to the isolation and characterization of CBS cell wall materials from two cacao varieties obtained after selected processing conditions (fermentation, drying, roasting). The cell wall polysaccharides were, thereafter, extracted by alkaline treatments, structurally characterised, and subjected to fermentation by selected lactic acid bacteria. The investigation of the compositional, structural, and functional properties of CBS cell wall polysaccharides would provide novel insight useful to the development of a highly effective approach that valorizes cocoa bean shells as a source of carbohydrate-based functional ingredients in food applications.

3.3 Materials and Methods

3.3.1 Materials

Fresh cacao pods (CCN 51) were supplied by Smart Natural Circle (Montreal, QC, Canada). Commercial samples of CBS (Nacional/Arriba (NAC) and CCN 51) were supplied by Cocoa processing industry (Guayaquil, Ecuador). The bacterial strains, *Lactobacillus rhamnosus GG* (RO343 AR) and *Bifidobacterium Longum* (ATCC 15707[™]) strains were supplied from Harmonium International (Mirabel, QC, Canada) and ATCC©. 1-Kestose, nytose, and 1Ffructofuranoysl-nystose were obtained from Wake Pure Chemical (Osaka, Japan). The analytical grade reagents were from Sigma- Aldrich Co., St. Louis, MO, USA and salts were obtained from Fisher Scientific (Fair Lawn, NJ).

3.3.2 Preparation of cocoa bean shells

Fresh cacao pods (CCN 51) were subjected to lab-scale fermentation and drying processes to retrieve the cacao bean shells (CBS) as described by Romanens et al. (2018). The inner contents of cacao pods, containing the mucilage-pulp and beans, were placed in tightly sealed containers (4L) to provide an anaerobic environment in an incubator (Excella E24 incubator, New Brunswick Scientific, Canada), for 117 hours, where the temperature was adjusted in 6-24hr intervals, 28°C at 0hr, 30°C at 6hr, 32°C at 21hr, 35°C at 30hr, 38°C at 45hr, 42°C at 54hr, and 45°C at 69hr. The pulp-bean mass was mixed every 24 hours to ensure proper removal of the pulp. After fermentation, the beans were subjected to drying in an oven for 72 hr at 45°C until a final water content of 6-7%. The beans were de-shelled manually. Commercial samples of cocoa bean shells were from two varieties (Nacional/Arriba (NAC) and CCN 51). Both freshly prepared and

commercial CBS samples were blended using a Model 7011C commercial blender (Conair corporation, Stamford, CT, USA) and analysed for their proximate compositions.

3.3.3 Proximate compositional analysis of cocoa bean shells

Moisture content of CBS was determined using a variation of AOAC method 927.05 by drying for 18 hr at 70 °C in vacuum oven. Total fat content of blended CBS was determined using a modified AOAC method 945.16. In this last method, Soxhlet extraction of CBS (1 g with 150 ml) was carried out using petroleum ether as solvent for 6 hr. Ash content was measured by AOAC method 942.05, charring 2 g of blended CBS for 30 min, followed by incineration in a muffle furnace (5 hr at 525 °C). Crude protein content was determined on 0.1 g of blended CBS by Dumas method (Saint-Denis and Goupy 2004) with a FP 628 analyser (LECO Corporation, St Joseph, MI, USA). Carbohydrate content (% dry weight) of blended CBS will be determined as:

100 - (crude fat + ash + crude protein)

Dietary fiber content was determined on 1 g of blended CBS using the total dietary fiber assay kit (Sigma-Aldrich Co., USA) and a modified AOAC method 985.29 (Latimer and AOAC International 2016), in which α -amylase, protease and amyloglucosidase activities were used to digest the starting material. The remaining fiber was precipitated with ethanol, washed with ethanol and acetone, filtered and, once dry, its protein and ash content were determined and subtracted from the dry weight. All analyses were performed in triplicate.

3.3.4 Isolation of cell wall materials from cocoa bean shells

The cell wall material (CWM) was sequentially isolated as described by Redgwell et al. (2003). Using de-fatted CBS (4g), the isolation proceeded as follows: 1) 80% ethanol (2hr, 50 rpm), 2) chloroform: methanol (1:1, 1.5 hr, 50 rpm), 3) chloroform: methanol (1:1, 1.5 hr, 50 rpm), 4) phenol: acetic acid: water (PAW, 2:1:1, overnight), 5) phenol: acetic acid: water (PAW, 2:1:1, 2hr). Each treatment was followed by centrifugation (7,000xg, 15mins) and vacuum filtration prior to the following step. The recovered CWM was dialyzed (5-8kDa cut-off) against distilled water and freeze-dried. The isolation of CWM was performed in triplicate.

3.3.5 Alkaline extraction of cocoa bean shell cell wall material polysaccharides

Extraction of CBS polysaccharides was performed as described by Khodaei and Karboune (2013) where CWM material (2% w/v) was suspended in KOH solutions (0.5 and 4M) containing 0.02M NaBH₄. Mixtures were incubated (60°C, 24 hr), thereafter centrifuged (10,000 x g for 15 mins) and recovered via filtration (0.45 μ m). Recovered polysaccharides were neutralized with HCl, dialyzed (5-8 kDa cut-off) and freeze-dried (-50°C, 0.003 mbar).

3.3.6 Structural characterisation of carbohydrates

3.3.6.1 Determination of total neutral sugars and uronic acid contents

Uronic acid content was measured by sulphamate/m-hydroxydiphenyl assay (Blumenkrantz & Asboe-Hansen, 1973). The phenol–sulphuric acid colorimetric assay was used for the determination of neutral sugar content (DuBois et al., 1956).

3.3.6.2 Monosaccharide profile

CWM and CBS polysaccharides were subjected to acid hydrolysis according to the method of Khodaei and Karboune (2013). Briefly, 1 ml of a mixture of 37% HCl/methanol (1:4 v/v) was added to the sample (200 μ l of liquid extract, or 5% (w/v) of solid in water). The mixtures were incubated at 70 °C for 24 hr and dried by heating in a convection incubator (Excella E24 incubator, New Brunswick Scientific, Canada). To complete hydrolysis, 3 ml of water and 1 ml of trifluoroacetic acid were added and heated at 100 °C for 1 hr. After evaporating the trifluoroacetic acid, the samples were neutralized with NaOH and centrifuged (8,000 rpm, 5 min). The supernatants (0.02 ml) were analyzed by high performance anion exchange chromatography (HPAEC) on a Dionex ICS 3000 system equipped with pulsed amperometric detection using a Carbopac PA-20 column (Dionex Co., Sunnyvale, CA, USA) at a temperature of 30 °C. Mobile phase was 20mM NaOH at flow rate of 0.4 ml/min. Rhamnose, arabinose, glucose, xylose, galactose, and mannose were used at varying concentrations (2.5 – 50 μ M) as standards.

3.3.6.3 Molecular weight distribution

A high-performance size-exclusion chromatography (HPSEC) system (Model 1525 binary HPLC pump, equipped with a Model 2414 refractive index detector, Waters Co., Milford, MA, USA) was used to estimate the molecular weight distribution of the carbohydrates. Analysis was determined using a TSK G5000 PWXL (Tosoh Co., Yamaguchi, Japan) column, with dextrans

(50– 670 kDa), soybean rhamnogalacturonan (0.125-1 g/l) and galactoglucomannan (0.125-1g/l) standards for molecular weight and concentration calibration. The temperature of the system was 30 °C, the eluent was 0.1 M NaCl, and the flow rate was set at 0.4 ml/min.

3.3.7 Extraction and determination of polyphenols content

The polyphenolic compounds were extracted using accelerated solvent extraction (Dionex ASE 350, Dionex, USA) from CBS and CWM. 0.5g samples were placed in 22ml cells containing Ottawa sand and Diatomaceous Earth. Samples were subjected to three consecutive repeated extractions using 80% methanol (40° C,1600 psi) to exhaust the material. The total polyphenolic content was determined using the folin-ciocalteu assay as described by Singleton, Orthofer, and Lamuela-Raventós (1999). A gallic acid calibration curve (0.001to 0.1mM) was constructed (R = 0.95). The total polyphenolic content was expressed as gram of gallic acid equivalent per gram of sample. All measurements were performed in triplicate.

3.3.8 Determination of phenolic compounds via LC-MS

CBS and CWM were analyzed by LC-MS using an Agilent 1290 Infinity II LC system coupled to the 6560-ion mobility Q-TOF-MS (Agilent Technologies, Santa Clara, USA). The LC separation was conducted on a Poreshell120 EC-C18 analytical column (Agilent Technologies; 2.7 μ m × 3 mm × 100 mm) connected with a Poreshell120 EC-C18 guard column (Agilent Technologies; 2.7 μ m × 3 mm × 5 mm). The mobile phase A was HPLC water with 0.1% formic acid and the mobile phase B was acetonitrile with 0.1% formic acid. HPLC parameters were as follows: injection volume was 1 μ l, the flow rate was 0.3 ml/min, and the column temperature was set to 30° C. The mobile phase profile used for the run in negative ion mode was 2% B (0 to 1.0 min), 2% - 20% B (1.0 to 4.0 min), 20%-100% B (4.0 to 8.0 min), 100% B (4.0 to 8.0 min), hold at 100% B (8.0-13.0 min), decrease to 2% B (13.0.0 to 13.5 min), and hold 2% B (13.5 to 14 min). The mass spectrometer was equipped with a Dual AJS ESI ion source operating in negative ionization mode. MS conditions were as follows: for ESI-, the drying gas temperature was 200 °C, drying gas flow rate was 12 l/min, sheath gas temperature was 250°C, sheath gas flow rate was 12 l/min, the pressure on the nebulizer was 35 psi, the capillary voltage was 4000 V, the fragmentor voltage was 240 V, and the nozzle voltage was 1000 V. Full scan MS data were recorded between mass-to-charge ratios (m/z) 100 and 1100 at a scan rate of 2 spectra/s, and were collected at both centroid and profile mode. Reference ions (m/z at 112.9856 and 1033.9881 for ESI-) were used for automatic mass recalibration of each acquired spectrum. Data treatment was conducted using Quantitative Analysis B.07.01 from Agilent MassHunter Workstation Software.

3.3.9 Prebiotic activity assay

The fermentability of CBS polysaccharides (0.5M KOH extracts, roasted CCN and NAC) by B. longum (ATCC 15707TM) and L. rhamnosus GG (Harmonium AR) was investigated and compared to that of inulin and rhamnogalacturonan, using a modification of the method reported by Yang et al. (2013). The selected strains, B. longum and L. rhamonosus GG, were maintained in an anaerobic chamber at 37°C. Pre-reduced reinforced clostridial broth was used for the reanimation of B. longum from a freeze-dried stock, and MRS broth was used for L. rhamnosus GG. An aliquot (0.5ml) of the primary cultures was then added into tubes containing 5ml of prereduced, carbohydrate-free MRS broth, supplemented with 0.4% glucose. Bacterial growth was measured using colony counting on Tryptic soy agar supplemented with 5% sheep blood for B. longum and pre-reduced MRS agar for L. rhamnosus GG. Tubes containing 5mL of pre-reduced carbohydrate-free MRS broth supplemented with 0.2% glucose, inulin, rhamnogalacturonan, and cell wall polysaccharides (CCN and NAC varieties) were inoculated with 0.1ml of secondary culture. Colonies were counted on Tryptic soy agar supplemented with 5% sheep blood for B. longum and pre-reduced MRS agar for L. rhamnosus GG after 6, 24, 48, 72 and 96 hr incubation intervals. Microbial counts were measured as colony forming units (CFU)/ml, and the resulting averaged data were transformed to log CFU/ml. A prebiotic activity score (PA) was calculated for each combination of bacterium and polysaccharide as follows:

$$PA = \frac{\left(Log \frac{CFU}{ml} \text{ on polysaccharide at } x \text{ } hr - Log \frac{CFU}{ml} \text{ on polysaccharide at } 0hr\right)}{Log \frac{CFU}{ml} \text{ on glucose at } x \text{ } hr - Log \frac{CFU}{ml} \text{ on glucose at } 0hr}$$

3.3.10 Determination of short chain fatty acid catabolites

An aliquot of cultures (0.5ml) was taken at 24 and 48hr. Bacteria were precipitated by centrifugation at 1400xg for 5min at 37°C in a Minispin Plus centrifuge (Eppendorf AG, Hamburg, Germany). The supernatants were filtered using 0.22 μ m polytetrafluorethylene syringe filters (Thermo Fischer Scientific) and stored at -80°C for a maximum of 2 weeks. Freshly thawed samples (0.02mL) were analyzed using the HPLC Model 1525 system (Waters) equipped with a refractive index detector and a Zorbax SB-C18 column (4.6x 250mm) (Agilent Technologies).

Elution was done with 0.005N sulfuric acid at a constant flow rate of 0.6ml/min. Standard curves were constructed using lactic, acetic, propionic, and butyric acids as standards (R = 0.95).

3.3.11 Statistical analysis

Statistical analyses were performed using XLSTAT software (Addinsoft, New York, NY, USA) in Microsoft Excel (Microsoft, Redmond, WA, USA). One-way analysis of variance (ANOVA) and the Tukey's honest significant difference (HSD) test were performed to detect significant differences (P < 0.05).

3.4 Results and Discussion

3.4.1 Characterization of cocoa bean shells and its cell wall material

The chemical composition of fermented (CCN), dried (CCN) and roasted (CCN and NAC) cocoa bean shells was determined based on AOAC standard protocols. Overall, the composition of CBSs comprised of 12.7-19.7% protein, 0.3-7.6% fat, 6.1-7.1% ash, 65.5-80.7% carbohydrates, and 50.1- 61.4% fibers (Table 3.1). These findings are within the ranges reported in the literature (Fakhlaei et al., 2019; Agus et al., 2018; Martínez et al., 2012; Mellinas et al., 2020; Okiyama et al., 2017), which found CBS to comprise of 6.2-18.6% protein, 2.02-6.46% fat, 6-11.67% ash, 17.8-55.85% carbohydrates, and 13.86-60.6% fibers . However, it should be noted that the broad range in the composition is due to variations in their origin, variety, processing condition, and assessment method.

The protein content has shown a noticeable increase (p < 0.05) from its fermented (13%, CCN) to dried (18.9%, CCN) states, followed by a decrease in its roasted state (12.8%, CCN). According to Agus, Mohamad, and Hussain (2018), the observed increase in the protein content from fermented to dried conditions is attributed to the increased microbial mass which produce nitrogen during growth". The decrease in protein content, following roasting, may be a result of the elevated temperatures involved in the roasting process (120-150°C). This mediates the Maillard reaction, which involves the interaction between the carbonyl group of a reducing sugar and a free amino acid from protein, thereby decreasing the protein content (Adeyeye et al., 2010; Agus et al., 2018). Interestingly, Okiyama et al. (2017), reported that only 1% of the protein content in cocca bean shells exists in its free form. The remainder is strongly bound to oxidized polyphenols, further converting into polyphenol-quinones. Similarly, the fat content increases (p < 0.05) with the processing of cocoa beans from fermentation to drying and roasting. The variation in the fat content
can be explained by the subjection of elevated temperatures involved in the drying and roasting processes (70-150°C), further inducing the migration of cocoa butter, from its cotyledon layer to the cocoa bean shells (El-Saied et al., 1981; Younes et al., 2022). The fatty acid composition in CBS has been reported to predominantly contain palmitic, oleic and linoleic acids (El-Saied et al., 1981). The ash content (p < 0.05) is relatively stable amongst the four samples, fermented, dried, and roasted CBS. Overall, cocoa bean shells have shown to contain higher ash contents than other fruit by-products, namely passion fruit seeds (1.34%), and apple pomace (0.5%) (Chau & Huang, 2004; Martínez et al., 2012; Sudha et al., 2007).

On average, the fiber content in cocoa bean shells accounts for around 50% of the entire material. Our findings are consistent with the literature (Martin-Cabrejas et al. 1994; Grillo et. al 2019), in which all four CBS samples contain 50.1-64.1% fiber. Using the sequential isolation approach, involving the use of chloroform and methanol, ethanol, followed by phenol, acetic acid, and water (2:1:1), to retrieve the cell wall material, we observed an overall cell wall material yield between 40.6 - 48.1%; higher than previously reported by Redgwell et. al (2003). The overall ratio between neutral sugar and uronic acid between the fermented and dried shells (0.82 - 0.97) compared to the roasted shells (1.4-1.8), indicate a higher proportion of neutral sugars upon subjection to roasting. Xiao et al. (2018) observed a reduction in the cell wall polysaccharide composition of apple chips, specifically fucose, rhamnose, arabinose, and xylose, with increased drying temperatures (60-90°C); further supporting that temperature conditions significantly influence the neutral sugar and uronic acid profile of foodstuff.

Based on the findings, as determined by anionic exchange chromatography (Table 3.1), all four sets of CBS CWM, regardless of their processing conditions, were predominantly rich in pectic polysaccharides (80.6-86%), comprised of uronic acids (35.1-50.5%), galactose (22.7-29.2%), arabinose (13.6-17.2%), and rhamnose (5.1-6.9%) residues. The remainder consists of hemicellulosic polysaccharides (13.9-19.4%), comprised of glucose (3.3-7.1%), fucose (2.4-3.6%), xylose and mannose (0.9-1.9%) residues. These are consistent with the findings of Redgwell et al. (2003) and Grillo et al. (2019) which suggested that CBS contain a greater proportion of pectic and non-cellulosic cell wall polysaccharides than cellulosic polysaccharides. Upon extracting the cell wall material, the protein content (12.9-16.6%) remained unchanged upon extraction, likely

indicating the complexes between dietary fiber and protein from the Maillard reaction, especially in the roasted samples (Redgwell et al. 2003).

Proximate Analyses of CBS								
Content (%, w/dry w)								
Components	Fermented CCN51	Dried CCN 51	Roasted CCN 51	Roasted NAC				
Protein	13±2 ^b	18.9±1.02ª	12.8±1.1 ^b	19.8±1.9ª				
Fat	$0.3{\pm}0.04^{\circ}$	$2.3{\pm}0.4^{b}$	2.1 ± 0.4^{b}	7.6±0.4ª				
Ash	$6.1{\pm}0.07^{b}$	7.1±0.13ª	$6.5{\pm}0.5^{ab}$	7.1±0.3ª				
СНО	80.6±2.1ª	$71.7{\pm}1.6^{b}$	78.7±1.1ª	65.5±1.6°				
Fiber	61.4±8.3ª	50.2±7.3ª	50.1±0.3ª	54.3±2.8ª				
Polyphenol content*	0.52±0.08 ª	0.63±0.13 ª	0.63±0.05 ª	0.44±0.02 ^a				
Composition of CWM	l of CBS							
CWM yield	40.6±12.6	41.6±2.6	49.4±0.04	48.1±3.2				
Polyphenol content*	0.05±0.002°	0.14±0.01ª	$0.1{\pm}0.01^{b}$	0.11 ± 0.01^{b}				
Protein content*	16.6±0.8ª	17.6 ± 1.6^{a}	17.3±1.4ª	12.9±1.1 ª				
Neutral Sugar content*	31.4±5.3°	29.9 ± 7.6^{bc}	41.9±5.8ª	40.9 ± 8.0^{ab}				
Uronic acid content*	32.5±7.6 ª	36.5±14.7 ª	22.8±2 ª	29.8±2.7ª				
Monosaccharide profile	e (%, g/100g rel. proportion)							
Fucose	2.6±0.03 ª	3.6±2.4ª	2.4±0.05ª	2.8±1.6ª				
Rhamnose	$6.2{\pm}0.9^{\mathrm{ab}}$	5.1±1.1 ^b	$6.9{\pm}0.4^{ab}$	6.4±0.1ª				
Arabinose	13.8±2.1ª	13.6±1.26ª	17.2±1.1ª	14.2±0.9ª				
Galactose	22.7±2.5ª	23.7±3.9ª	29.2±1.4ª	27.2 ± 0.7^{a}				
Glucose	3.3±0.1 ^b	$5.2{\pm}0.7^{b}$	$7.1{\pm}0.8^{a}$	5.9±0.3ª				
Xylose+ Mannose	0.9±0.3ª	1.3±0.12 ^a	1.9±0.15ª	1.5±0.2ª				
Uronic acids	50.5±5.9ª	47.5±1.1ª	35.1±2.0ª	42.1±2.2ª				
Hemi-cellulosic PS*	13.9±0.8ª	19.4±6.1ª	17.8±1.9ª	17.5±2.9ª				
Pectic PS*	$86.0{\pm}0.8^{a}$	80.6±6.1ª	82.2±1.9ª	82.5±2.9ª				
Polyphenol Profile of	CBS and CWM (ng/Kg CBS	S / CWM CBS)						
Gallic acid	0.2±0.01 ^{ab} /0.3±0.03 ^a	0.2±0.06ª/0.2±0.02ab	$0.1 \pm 0.02^{bc}/n.d.**^{c}$	n.d.**°/n.d.**°				
Protocatechuic acid	7.2±1.9°/9.3±0.03°	23.2±3.9 ^b /34.6±5.2 ^a	$7.0{\pm}0.5^{cd}$ /5.3 ${\pm}0.3^{cd}$	$1.3\pm0.1^{d}/7.5\pm1.1^{c}$				
Catechin	$4.1{\pm}0.4^{cd}$ /6.4 ${\pm}0.1^{cd}$	24.3±6.7ª/20.2±0.9ªb	$0.3{\pm}0.06^d{/}0.4{\pm}0.04^d$	$13.0 \pm 1.3^{bc}/2.1 \pm 0.4^{d}$				
Epicatechin	37.4±0.1 ^{bc} /65.8±1.7 ^{ab}	122.8±7.5ª/124.4±26.9ª	$6.1 \pm 0.5^{bc}/10.1 \pm 0.7^{bc}$	0.9±0.1°/19.8±2.6 ^{bc}				
Caffeic acid	$0.1{\pm}0.03^{b}/0.4{\pm}0.1^{b}$	$0.3{\pm}0.1^{b}/0.4{\pm}0.1^{b}$	$0.1 \pm 0.003^{b}/9.6 \pm 0.9^{b}$	$1.1\pm0.1^{a}/0.3\pm0.1^{b}$				
p-Coumaric acid	$0.05{\pm}0.01^{b}/~0.05{\pm}0.0005^{b}$	$0.1{\pm}0.01~^{ab}~/0.1{\pm}0.02^{ab}$	n.d.** ^b / n.d.** ^b	$0.1 \pm 0.01^{ab} / 0.2 \pm 0.04^{a}$				
Ferulic acid	0.2±0.01ª/0.2±0.04ª	0.1±0.04ª/0.05±0.0001ª	$0.4^{a}/0.3{\pm}0.2^{a}$	0.2±0.1ª/1.2±0.1ª				
Quercetin	$2.1\pm0.1^{bc}/4.6\pm0.5^{a}$	3.1±0.3 ^{ab} /6.0±1.1 ^a	$0.4{\pm}0.04^{d}/0.5{\pm}0.03^{d}$	$0.8{\pm}0.1^{cd}/n.d.**d$				
Caffeine	261.6±1.2ª/257.6±7.9ª	139.4±3.9ª/169.5±13.1ª	4.5±0.2ª/19.3±0.4ª	17.7±0.1ª/22.3±5.0ª				

Table 3.1. Proximate composition of Cocoa Bean Shells (CBS) and saccharide profile of their corresponding Cell Wall Materials (CWM)

Values reported as average \pm standard deviation *Units expressed as % weight per weight CWM

**not detected

 $^{\rm a}$ Within the same row, means with different letters are significantly different at $P \le 0.05$

3.4.2 Polyphenolic profile of cocoa bean shells and its cell wall material

Polyphenolic compounds are naturally present within the cotyledons of the cocoa seeds; however, compounds can migrate towards the shell through diffusion, further enriching the byproduct (Andres-Lacueva et al., 2008; Forsyth et al., 1958; Okiyama et al., 2017). The CBS total polyphenol content comprised between 0.44-0.63% (Table 3.1) similar to the content found in cocoa beans (Hernández-Hernández et al., 2018; Rojo-Poveda et al., 2020). In addition, after isolating its CWM, the total polyphenol content was obtained around 0.05-0.14%; directly influenced by the use of solvents in the sequential isolation process used to retrieve the CWM. In general, the polyphenol content is expected to increase with further processing. Our findings showed no significant differences in the polyphenol content of CBS at different processing conditions of fermentation, drying and roasting. However, overall, the roasted CBS samples (NAC and CCN) hold a greater total polyphenol content as compared to their fermented and dried states. Alternatively, dried CBS polyphenols may be more tightly bound to cell wall materials than fermented and roasted CBS. Subjection of heat from the drying and roasting process can mediate not only the Maillard reaction, but also promote condensation and polymerization. Given our findings showed a great amount of crude fiber (non-digestible carbohydrates and lignin), one may suggest that lignin is pyrolyzed slowly with the onset of roasting conditions allowing for the release of polyphenols entrapped within its network, otherwise known as the polyphenol complex (Agus et al., 2018; Brebu & Vasile, 2010; Hečimović et al., 2011).

Based on our findings, as determined by LC-MS (p < 0.05) in Table 3.1, the most abundant class of polyphenols found in CBS and its CWM are the flavan-3-ols, which include epicatechin (122.8-0.9 ng/kg material or extract) and catechin (24.3-0.3 ng/kg material or extract; this is consistent with the literature (Miller et al., 2006; Payne et al., 2010). Other polyphenols were detected in both CBS and the corresponding CWM, which include flavonoids, such as quercetin, as well as phenolic acids, such as gallic, protocatechuic, p-coumaric, ferulic, and caffeic acids. Nevertheless, cocoa and its by-products are known for its richness in methylxanthines, such as theobromine and caffeine (Okiyama et al., 2017). Our findings detected considerable amounts of caffeine among all samples of CBS and its CWM; with the highest amounts found among the fermented and dried types (139.4-261.6 ng/kg material). Interesting, the caffeine content was significantly lower after the roasting process. According to Hečimovic et. al (2011), caffeine and

other polyphenols were found in greater amounts in their raw and lightly roasted forms (160°C); further indicating the effect of processing conditions on the polyphenolic profile.

3.4.3 Extraction of cell wall polysaccharides and their characterisation

To extract and characterize the CBS cell wall polysaccharides, alkaline extraction was performed using two concentrations of KOH, 0.5M and 4M. The polysaccharide yield (Table 3.2) ranged between 23.7-35.4 % (w/w) using 0.5M KOH depending on the starting materials (fermented, dried, roasted) and variety. Interestingly, as the alkaline solution increased, the roasted samples (CCN and NAC) saw an increase in their polysaccharide yields (32.9-53.6%, w/w). Such findings are consistent with the findings assessed on pectic polysaccharides derived from potato peel and cranberry pomace (Khodaei & Karboune, 2013; Spadoni Andreani et al., 2021). In addition, the results show the proportion of neutral sugars decreased as the alkaline solution was increased, whereas the molar proportion of uronic acid remained relatively unchanged. As a result, the uronic/neutral sugar ratio increased, possibly indicating the onset of debranching or defragmentation of neutral sugar regions.

The monosaccharide composition (Table 3.2) reveals the main sugars in the extracted CBS cell wall polysaccharides were uronic acids (42.9-58.2%), arabinose (7.8-16.4%), and galactose (8.1-22.4%) for the 0.5M alkaline treatment, signifying the presence of arabinan, galactan and rhamnogalacturonan. As for the 4M alkaline treatment, the main sugars were uronic acids (56.4-66.7%), galactose (9.9-15.9%), xylose and mannose (2.6-15.9%), indicating a higher proportion of hemicellulose as compared to the latter. We can estimate the proportion of pectic polysaccharides by determining the pectic neutral sugars content, calculated as the ratio of rhamnose, arabinose and galactose to total neutral sugars. The results suggest the polysaccharide profile of CBS alkaline extracts are predominantly pectic polysaccharides (56.4-88.8%) with the remainder being hemi-cellulosic polysaccharides, as signified by the presence of xylose, mannose, and glucose. Using a 4M KOH extraction treatment, Redgwell et al. (2003) reported the presence of a mixture of hemicelluloses and pectic polysaccharides based on the levels of uronic acid (26.2%) xylose (23.5%), glucose (21.8%), and mannose (9.9%). In addition, the study reported rhamnogalacturonan as the main pectic polysaccharide, as well as three hemi-cellulosic polysaccharides, namely galactoglucomannan, xylogalacturonan, and glucoarabinoxylan (Redgwell et al., 2003). These findings are supported by the fact that pectic neutral sugars are of a lower proportion compared to 0.5M alkaline extracts, comprising of 55.6-76.1% (Table 3.2). Considering all samples (using both treatments) showed high proportions of pectic polysaccharides, the ratio of arabinose and galactose to rhamnose provides an estimate of the level of branching of rhamnogalacturonan I. As seen in Table 3.2, the 4M extracts showed the highest ratios, indicating the presence of more abundant arabinan and galactan side chains. Furthermore, by increasing the alkaline solution concentration, the degree of branching increased by two to fivefolds, depending on the processing condition. Based on the rhamnose molar proportions with the 0.5M alkaline treatment, it can be assumed that rhamnogalacturonan I is more abundant in Fermented and Roasted CCN 51 samples. As the alkaline solution increased, the molar proportion had decreased to 2.3-3.7%. All in all, the NAC variety attained a higher polysaccharide yield (35.4-53.6%) among the two alkaline treatments compared to the CCN variety (19.5-32.9%). In addition, the polysaccharide profiles of both varieties showed similarities, with the exception that the CCN variety expressed a richer profile of galactose (10.7-15.9% μ M) and xylose and mannose (4.2-10.6% μ M) while the NAC variety expressed a richer content of arabinose (13.4% μ M).

Roasted samples (NAC and CCN varieties) were analyzed for their molecular weight (MW) distribution. The results showed a significant impact of increasing the alkaline solution from 0.5 to 4M KOH. In the 0.5M extracts, three main MW populations were found at 3.7, 12, and 130, kDa (NAC variety) and 4.5, 9.1, and 32.5 kDa (CCN variety) (Figure 3.1). Increasing the alkaline solution resulted in a shift from high to low molecular weight fractions, where the 4M extract main populations were found at 24, and 6.3 kDa (NAC variety) and 11, and 3.2 kDa (CCN variety) (Figure 3.1). This is a result of increasing the alkaline solution which allowed for a higher degree of debranching or defragmentation of the pectic polysaccharides present in CBS. As per the author's knowledge, no studies have been reported on the MW populations of CBS. However, Redgwell and Hansen (2000) previously reported that a high MW population (2000 kDa) was detected in cell wall polysaccharides extracted from unfermented and fermented cocoa beans. The MW populations of cranberry pomace (1.5-1500 kDa), potato peel protein (1-600 kDa), and olives (260-400 kDa) have also been reported, showing a profile mixture of both low and high MW fractions (Jiménez et al., 1994; Khodaei & Karboune, 2013; Spadoni Andreani et al., 2021). The presence of low MW fractions may be due to the disassociation of non-covalently bound oligosaccharides during the dialysis purification steps (Spadoni Andreani et al., 2021).

Table 3.2. Alkaline extraction of polysaccharides from the cell wall material (CWM) of cocoa bean shells (CBS)

	Fermented CCN51	Dried CCN 51	Roasted CCN 51	Roasted NAC
0.5M KOH				
PS yield (%, w/w)	23.7±0.01	24.4±0.05	29.1±0.4	35.4±4.3
Neutral sugar content*	27.1±1.9ª	25.5±3.2 ª	20.6±0.7 ^b	13.5±2.3 ^b
Uronic acid content*	54.5±9.3 ª	55.1±0.9ª	58.3±5.2 ª	49.8±3.02 ª
Uronic/Neutral Sugar ratio	2.01	2.16	2.83	3.69
Monosaccharide Composition (%, g/100g - relative j	proportion)		
Fucose	0.9 ± 0.2^{b}	$0.7{\pm}0.07^{\text{ b}}$	1.8±0.02 ª	1.8±0.1 ª
Rhamnose	11.9±0.6 ª	9.9±0.8 ª	12.4±0.3 ª	6.8±1.1 ^b
Arabinose	16.4±0.9 ª	13.9±0.8 ª	8.6±0.01 ^b	7.8±0.6 ^b
Galactose	22.4±0.9 ª	18.5±0.4 ^b	8.1±0.2 °	17.5±1.3 ^b
Glucose	1.1±0.1 ^b	3.7±1.1 ª	5.2±0.5 ª	2.9±0.1 ab
Xylose+ Mannose	$4.4{\pm}0.9^{b}$	8.5±1.9 ^{ab}	12.6±2.0°	5.0±0.4 ^b
Uronic acids	42.9±2.5 ^b	44.9±2.6 ^b	51.4±3.1 ^{ab}	58.2±2.8 ^b
Pectic neutral sugars*	88.8±1.7 ^a	76.8±4.2 ^b	59.9±2.8°	76.8±1.8 ^b
Branching	3.3	3.3	1.3	3.7
4M KOH				
PS yield (%, w/w)	19.5±0.05	21.1±0.07	32.9±0.08	53.6±8.1
Neutral sugar content*	13.9±2.1 ª	12.3±0.7 ^a	11.4±1.4 ª	10.7±0.2 ª
Uronic acid content*	47.9±5.1 ^a	60.3±9.3 ^a	60.8±3.6 ª	51.6±1.4 ª
Uronic/Neutral Sugar ratio	3.44	4.89	5.31	4.84
Monosaccharide Composition (%, g/100g - relative j	proportion)		
Fucose	1.2±0.2 ^b	1.8±0.1 ^b	4.3±0.1 ª	3.4 ± 0.4^{a}
Rhamnose	3.5±0.1 ab	2.3±0.4 ^b	3.7±0.3 ^a	3.7±0.5 ª
Arabinose	5.9±0.7 °	4.2±0.3 °	10.6±0.6 ^b	13.4±0.6 ª
Galactose	14.9±1.8 ª	15.9±0.4 ª	10.7±0.4 ^b	9.9±0.2 ^b
Glucose	2.2±0.7 ^a	4.2±2.2 ª	1.5±0.1 ^a	2.03±0.04 ª
Xylose+ Mannose	15.9±0.6ª	7.1±1.8 ^b	2.6±0.04 °	3.03±0.2 °
Uronic acids	56.4±1.0 ª	64.6±4.9 ª	66.7±1.4 ª	64.5±1.02 ª
Pectic neutral sugars*	55.6±4.5 ^b	63.6±5.9 ^{ab}	74.85±0.62 ª	76.1±1.3 ª
Branching	6	8.9	5.8	6.3

Values represented as averages ± standard deviation Pectic neutral sugars calculated as (rhamnose + arabinose + galactose) x 100 / total neutral sugars



Figure 3.1. Molecular weight distribution of CWP alkaline extracts. Letters are denoted by the different varieties (NAC, CCN) subjected to different alkaline solution treatments (0.5M, 4M KOH) A: NAC 4M, B: NAC 0.5M, C: CCN 4M, D: CCN 0.5M, respectively.

3.4.4 Prebiotic activity

The prebiotic activity scores of CBS polysaccharides, rhamnogalacturonan, and inulin, a known prebiotic polysaccharide (Shoaib et al., 2016), were compared using them as carbon sources for the anaerobic growth of two probiotic bacteria strains, L. rhamnosus GG and B. longum. Both strains were handled and grown in anaerobic conditions. While L. rhamnosus GG is oxygentolerant, the anaerobic conditions were selected to simulate the intestinal environment more accurately. In general, *Lactobacillus* species show a significant heterogeneity in their abilities to ferment carbohydrates (Boguta et al., 2014; Broadbent et al., 2012; Kim et al., 2010; Spadoni Andreani et al., 2021). Figure 3.2 shows that CBS polysaccharides (NAC and CCN varieties) showed a prebiotic activity score similar to that of inulin and rhamnogalacturonan for L. rhamnosus GG (Figure 3.2). At 24hr of fermentation, the prebiotic activity scores of CBS polysaccharides were significantly higher ($P \le 0.05$) than that of inulin and rhamnogalacturonan. For *B. longum* (Figure 3.3), CBS polysaccharides showed statistically significant ($P \le 0.05$) prebiotic activity scores at 48hr of fermentation than that of inulin and rhamnogalacturonan; however, the prebiotic activity scores decreased at the 72hr fermentation period. Therefore, it appears that CBS polysaccharides, primarily rich in pectin, galactan, and arabinan, effectively stimulate the growth of both strains. Dietary fibers, rich in pectin, undergo bacterial fermentation in the ileum and colon, leading to acidification of the colonic contents as well as the production of short-chain fatty acids (Ashaolu et al., 2021; Koropatkin et al., 2012; Slavin, 2013). Studies reported effective growth of L. rhamnosus GG with pectic polysaccharides and galactooligosaccharides derived from pearl millet fibre, and F. kuhistanic leaves (Farooq et al., 2017; Islamova et al., 2017) and *B. longum* with pectic polysaccharides derived from cranberry pomace, pumpkin peel, and citrus pectin (Bianchi et al., 2018; Jun et al., 2006; Spadoni Andreani et al., 2021).

Interestingly, gene expression analysis of *B. longum* indicate the presence of over 40 glycosyl hydrolases, which include 2 xylanases, 9 arabinosidases, 2 α -galactosidases and more, whose predicted substrates cover a wide range of di, tri- and higher order oligo-saccharides, indicating the strain's high affinity for oligosaccharides with a degree of polymerization less than 8 (Schell et al., 2002).



Figure 3.2. Prebiotic activity scores of *Lactobacillus rhamnosus* GG (R0343 AR) on polysaccharide extracts, inulin and rhamnogalacturonan, measured at 6,24, 48, 72 and 96h of incubation. Values represented as an average \pm standard deviation. Poly NAC: polysaccharide NAC variety, Poly CCN: polysaccharide CCN variety, RhamnoG: Rhamnogalacturonan, respectively. For each incubation time, bars with different letters represent scores significantly different at P < 0.05.



Figure 3.3. Prebiotic activity scores of *Bifidobacterium longum* (ATCC® 15707) on polysaccharide extracts, inulin and rhamnogalacturonan, measured at 6, 24, 48, 72 and 96h of incubation. Values represented as an average \pm standard deviation. Poly NAC: polysaccharide NAC variety, Poly CCN: polysaccharide CCN variety, RhamnoG: Rhamnogalacturonan, respectively. For each incubation time, bars with different letters represent scores significantly different at P < 0.05.

Regarding, L. rhamnosus GG, this strain has shown the highest affinity towards fucosylated oligo- and polysaccharides as it has been found to express the fucose permease, fucose isomerase, fucolose kinase, fucose mutarotose, and fucolose-1-phosphate aldolase, promoting the fucose catabolic pathway to produce 1,2-propanediol or lactate (Becerra et al., 2015). According to one study assessing the probiotic activities of B. longum and L. rhamnosus GG using qPCR showed that both strains indirectly utilize pectic polysaccharides as a source of prebiotics. The polysaccharides undergo metabolization prior by other gut microbiota, namely E. coli H, resulting in metabolic end products that are then metabolized by B. longum and L. rhamnosus GG (Li et al., 2020). In addition, studies suggest the degree of methylation of PS may affect the fermentation rate. Based on FTIR and NMR analysis, CBS pectic fractions have been characterized as highly acetylated low methyoxyl homogalacturonan and type I rhamnogalacturonan with galactan or arabinogalactan side chains (Vriesmann et al., 2011). In addition, Larsen et al. (2019) reports that Lactobacillus strains showed higher fermentation rates with polysaccharides with a low degree of methylation. This may explain the faster fermentation rates observed with L. rhamnosus GG with CBS polysaccharides. Studies also found conflicting correlations between the degree of esterification and bacterial taxa in stimulating their growth (Chengxiao et al., 2021; Ferreira-Lazarte et al., 2019; Larsen et al., 2019). Further investigation on the structural linkage, degree of methylation and degree of esterification is needed to assess whether they could result in the differential promotion of the growth of selected bacteria.

SCFA are produced upon the fermentation of dietary fiber by gut microbiota (Lin et al., 2016; Spadoni Andreani et al., 2021). Prebiotic production of SCFA can be used to complement *L. rhamnosus GG* and *B. longum* fermentation of CBS polysaccharides. Table 3.3 and Table 3.4 show the concentration of four SCFA after 24 and 48 hr of fermentation of *L. rhamnosus GG* and *B. longum* with glucose, CBS polysaccharides (NAC and CCN varieties) and positive controls, inulin and rhamnogalacturonan. In the culture of *L. rhamnosus GG* (Table 3.3), the main increase was seen in the lactic acid followed by acetic acid. Although unabundant, the significance (P < 0.05) was observed with the production of propionic acid for CBS polysaccharides from both varieties (NAC and CCN) at 48hr of fermentation. *L. rhamnosus GG* is known for its ability to produce lactic and propionic acid during fermentation, which can promote anti-inflammatory and antimicrobial properties in our gastrointestinal tract (Al-Lahham et al., 2010; Markowiak-Kopeć & Śliżewska, 2020). The production of butyric acid was prevalent in glucose, inulin, and rhamnogalacturonan, indicating that CBS polysaccharides do not favor the production of butyrate in the *L. rhamnosus* fermentation pathway. However, the presence of butyric acid can be attributed to the metabolization of lactate, further explaining the decrease in the lactate concentration with increased incubation time (Cummings et al., 1987; Karboune et al., 2022).

Similarly, to the SCFA profile of L. rhamnosus GG, the culture of B. longum was characterized by the greatest increase in lactic acid, followed by acetic acid predominantly after 48hr of fermentation; with the except of CCN polysaccharides. The acetic acid content of the medium relatively remained unchanged for the polysaccharides. The lactic acid concentrations produced by polysaccharides were higher at 24hr of fermentation than that of glucose, inulin, and rhamnogalacturonan. At 48hr of fermentation, the lactic acid concentrations decreased and were surpassed by their positive controls. However, propionic, and butyric acid concentrations either decreased or were not detected after 24 hr of B. longum fermentation. Overall, the total SCFA profile of polysaccharides were higher at 24 hr than 48 hr, indicating possible SCFA degradation. Studies suggest that Bifidobacterium spp., specifically. B. longum, are producers of lactate, acetate and propionate in the human intestinal tract (Liu et al., 2020; Markowiak-Kopeć & Śliżewska, 2020). However, only the production of lactic and acetic acids was observed with CBS polysaccharides. In addition, when carbohydrates are in excess, Bifidobacterium utilize the fermentation pathway, to produce two molecules of acetate with 1 molecule of lactate (LeBlanc et al., 2017; Macfarlane & Macfarlane, 2003; Markowiak-Kopeć & Śliżewska, 2020). However, our findings suggest a higher proportion of lactic than acetic acid.

Overall, comparing the total SCFA released at 48hr revealed that CBS polysaccharides positively contributed towards effective growth of both strains, producing high proportions of lactic and acetic acid, compared to the growth initiated by glucose, inulin and rhamnogalacturonan. New research suggests that multi-strain probiotics can contribute to more enhanced health benefits compared to single-strain probiotics (Kwoji et al., 2021; Puvanasundram et al., 2021). Further investigation can assess the effect of multi-culturing, using *L. rhamnosus GG* and *B. longum*, to determine whether a synergistic approach can promote higher prebiotic activity scores and SCFA production at faster fermentation rates.

Table 3.3. Concentration (10⁻³mol/L) of short chain fatty acids released in the 0.5M alkaline treated cell-wall polysaccharide samples (NAC, CCN) as compared to glucose, inulin and rhamnogalacturonan standards which were obtained from *Lactobacillus rhamnosus* GG (R0343 AR) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation.

	Time (h)	Glucose	Inulin	RhamnoG	NAC	CCN
Lactic acid	24	313.0±4.0ª	212.5±15.5ª	508.8±10.3ª	427.6±62.0ª	343.0±85.9ª
	48	386.0±16.2 ª	$315.7{\pm}14.8^{a}$	293.5±26.0ª	354.7±41.1ª	393.0±44.2ª
Acetic acid	24	58.1±8.4ª	46.4*a	112.2±nd*a	42.5±0.9ª	71.9* ^a
	48	111.7*a	66.4*a	$39.0{\pm}8.5^{a}$	35.3±3.0ª	96.8*a
Propionic acid	24	4.7* ^a	1.0±0.04ª	1.8±0.03ª	1.1±0.1ª	1.2*a
	48	0.7* ^b	0.4* ^b	$0.6{\pm}0.06^{b}$	1.1 ± 0.2^{ab}	1.9±0.3ª
Butyric acid	24	1.8*a	0.3*a	$0.4{\pm}0.04$ a	-	-
	48	-	-	-	-	-
Total	24	377.6	260.2	623.2	471.2	416.1
	48	498.4	382.5	333.1	391.1	491.7

Values represented as average \pm standard deviation

RhamnoG; Rhamnogalacturonan

* Standard deviation is zero or not detected

^a Within the same row, means with different letters are significantly different at P < 0.05.

Table 3.4. Concentration (10⁻³mol/L) of short chain fatty acids released in the 0.5M alkaline treated cell-wall polysaccharide samples (NAC, CCN) as compared to glucose, inulin and rhamnogalacturonan standards which were obtained from *Bifidobacterium longum* (ATCC 15707®) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation.

	Time (h)	Glucose	Inulin	RhamnoG	NAC	CCN
Lactic acid	24	31.3±5.1ª	56.3*a	178.0* ª	169.0±2.4 ª	273.2±66.4 ª
	48	404.4 ± 96.6^{a}	281.3 ± 1.8 a	327.9±2.1 ^a	299.8±18.3ª	216.2±1.5 ª
Acetic acid	24	44.4±3.9ª	43.7±10.3ª	52.6* ^a	39.5*ª	53.9* ^a
	48	230.1±11.3ª	55.3±3.6 ^b	50.5±1.5 ^b	28.6 ± 2.9^{b}	53.3* ^b
Propionic acid	24	1.9±0.4ª	1.3±0.2ª	2.6±0.4ª	1.9±0.2ª	1.7* ^a
	48	0.6*a	0.5±0.1ª	0.5 ± 0.03^{a}	0.3* ^a	2.0* ^a
Butyric acid	24	0.3*	0.3*	-	-	-
	48	-	-	-	-	-
Total	24	77.9	101.6	233.3	210.4	328.8
	48	635.1	337.1	378.9	328.7	271.5

Values represented as average \pm standard deviation

RhamnoG; Rhamnogalacturonan

* Standard deviation is zero or not detected

^a Within the same row, means with different letters are significantly different at P < 0.05.

3.5 Conclusion

The chemical composition of CBS (fermented, dried, and roasted) is predominantly carbohydrates (65.5-80.7%), of which around 50% is dietary fiber. CBS was also found to contain a great source of protein (12.7-19.7%), ash (6.1-7.1%). As expected, the fat content (0.3-7.6%)was less the amounts reported for cocoa beans; however, fat migration from the bean to the outer shell is induced upon subjection to high temperatures involved in the drying and roasting processes. Upon isolation of the CBS CWM, the polysaccharide profile was found to contain both pectic (80.6-86%) and hemi-cellulosic (13.9-19.4%) polysaccharides. Furthermore, the protein content of CWM remained unchanged relative to their raw states, indicating possible interactions with dietary fiber. The cell wall polysaccharides were isolated from CWM using two concentrations of KOH, 0.5M and 4M. The less concentrated extract, although had a lower recovery yield, was favored as it provided a better neutral sugar and uronic acid profile. By increasing the alkaline concentration, defragmentation and debranching occurred. Overall, the nature of the CBS cell wall composition was found to be predominantly rich in pectic polysaccharides, such as rhamnogalacturonan. When tested for the promotion of the growth of selected probiotic strains, CBS cell wall polysaccharides performed similarly or more than inulin and rhamnogalacturonan based on the prebiotic activity scores. The short chain fatty acid profiles were characterized by high amounts of lactic acid, followed by acetic and propionic acid. Further investigation of the use of these polysaccharides using other single-strain and possibly multi-strain probiotics is needed to assess the effect of CBS polysaccharides on the selectivity of colonic fermentation.

References

- Adeyeye, E.I., R.O. Akinyeye, I. Ogunlade, O. Olaofe, & J.O. Boluwade. 2010. "Effect of Farm and Industrial Processing on the Amino Acid Profile of Cocoa Beans." *Food Chemistry* 118 (2): 357–363. doi:10.1016/j.foodchem.2009.04.127.
- Adeyina, A., D. Apata, A. Annongu, O. Olatunde, O. Alli, & K. M. Okupke. 2010. "Performance and Physiological Response of Weaner Rabbits Fed Hot Water Treated Cocoa Bean Shell Based Diet." *Research Journal of Veterinary Sciences* 5: 53–57.
- Aguiló-Aguayo, I., J. Walton, I. Viñas, & B.K. Tiwari. 2017. "Ultrasound Assisted Extraction of Polysaccharides from Mushroom By-Products." *LWT* 77 (April): 92–99. doi:10.1016/j.lwt.2016.11.043.
- Agus, B.A.P, N.N. Mohamad, & N. Hussain. 2018. "Composition of Unfermented, Unroasted, Roasted Cocoa Beans and Cocoa Shells from Peninsular Malaysia." *Journal of Food Measurement and Characterization* 12 (4): 2581–2589. doi:10.1007/s11694-018-9875-4.
- Al-Lahham, S.H., M.P. Peppelenbosch, H. Roelofsen, R.J. Vonk, & K. Venema. 2010.
 "Biological Effects of Propionic Acid in Humans; Metabolism, Potential Applications and Underlying Mechanisms." *Biochimica et Biophysica Acta (BBA) - Molecular and Cell Biology of Lipids* 1801 (11): 1175–1183. doi:10.1016/j.bbalip.2010.07.007.
- Andres-Lacueva, C., M. Monagas, N. Khan, M. Izquierdo-Pulido, M. Urpi-Sarda, J. Permanyer, & R. M. Lamuela-Raventós. 2008. "Flavanol and Flavonol Contents of Cocoa Powder Products: Influence of the Manufacturing Process." *Journal of Agricultural and Food Chemistry* 56 (9): 3111–3117. doi:10.1021/jf0728754.
- Ashaolu, T.J., J.O. Ashaolu, & S.A.O. Adeyeye. 2021. "Fermentation of Prebiotics by Human Colonic Microbiota in Vitro and Short-chain Fatty Acids Production: A Critical Review." *Journal of Applied Microbiology* 130 (3): 677–687. doi:10.1111/jam.14843.
- Becerra, J.E., M.J. Yebra, & V. Monedero. 2015. "An L-Fucose Operon in the Probiotic Lactobacillus Rhamnosus GG Is Involved in Adaptation to Gastrointestinal Conditions." Edited by H. L. Drake. *Applied and Environmental Microbiology* 81 (11): 3880–3888. doi:10.1128/AEM.00260-15.
- Bernaert, H., & I.D. Ruysscher. 2016. "Process of Producing Cocoa Shell Powder."
- Bianchi, F., N. Larsen, T. de Mello Tieghi, M.A.T. Adorno, W. Kot, S.M.I. Saad, L. Jespersen, & K. Sivieri. 2018. "Modulation of Gut Microbiota from Obese Individuals by in Vitro Fermentation of Citrus Pectin in Combination with Bifidobacterium Longum BB-46." *Applied Microbiology and Biotechnology* 102 (20): 8827–8840. doi:10.1007/s00253-018-9234-8.
- Blumenkrantz, N., & G. Asboe-Hansen. 1973. "New Method for Quantitative Determination of Uronic Acids." *Analytical Biochemistry* 54 (2): 484–489. doi:10.1016/0003-2697(73)90377-1.
- Boguta, A.M., F. Bringel, J. Martinussen, & P.R. Jensen. 2014. "Screening of Lactic Acid Bacteria for Their Potential as Microbial Cell Factories for Bioconversion of Lignocellulosic Feedstocks." *Microbial Cell Factories* 13 (1): 97. doi:10.1186/s12934-014-0097-0.
- Bonnina, E., M. Brunel, Y. Gouy, L. Lesage-Meessen, M. Asther, & J.F. Thibault. 2001.
 "Aspergillus Niger I-1472 and Pycnoporus Cinnabarinus MUCL39533, Selected for the Biotransformation of Ferulic Acid to Vanillin, Are Also Able to Produce Cell Wall Polysaccharide-Degrading Enzymes and Feruloyl Esterases." *Enzyme and Microbial Technology* 28 (1): 70–80. doi:10.1016/S0141-0229(00)00277-5.

- Brebu, M., & C. Vasile. 2010. "Thermal Degradation of Lignin—a Review." Cellulose Chemistry & Technology 44 (9): 353.
- Broadbent, J.R., E.C. Neeno-Eckwall, B. Stahl, K. Tandee, Hui Cai, W. Morovic, P. Horvath, et al. 2012. "Analysis of the Lactobacillus Casei Supragenome and Its Influence in Species Evolution and Lifestyle Adaptation." *BMC Genomics* 13 (1): 533. doi:10.1186/1471-2164-13-533.
- Chau, C.F., & Y.L. Huang. 2004. "Characterization of Passion Fruit Seed Fibres—a Potential Fibre Source." *Food Chemistry* 85 (2): 189–194. doi:10.1016/j.foodchem.2003.05.009.
- Chengxiao, Y., W. Dongmei, Z. Kai, L. Hou, H. Xiao, T. Ding, D. Liu, X. Ye, R.J. Linhardt, & S. Chen. 2021. "Challenges of Pectic Polysaccharides as a Prebiotic from the Perspective of Fermentation Characteristics and Anti-Colitis Activity." *Carbohydrate Polymers* 270 (October): 118377. doi:10.1016/j.carbpol.2021.118377.
- Choi, J., N. Kim, H.Y. Choi, & Y.S. Han. 2019. "Effect of Cacao Bean Husk Powder on the Quality Properties of Pork Sausages." *Food Science of Animal Resources* 39 (5): 742– 755. doi:10.5851/kosfa.2019.e62.
- Cummings, J.H., E.W. Pomare, W.J. Branch, C.P. Naylor, & G.T. Macfarlane. 1987. "Short Chain Fatty Acids in Human Large Intestine, Portal, Hepatic and Venous Blood." *Gut* 28 (10): 1221–1227. doi:10.1136/gut.28.10.1221.
- DuBois, M., K. A. Gilles, J. K. Hamilton, P. A. Rebers, & F. Smith. 1956. "Colorimetric Method for Determination of Sugars and Related Substances." *Analytical Chemistry* 28 (3): 350– 356. doi:10.1021/ac60111a017.
- El-Saied, H.M., M. K. Morsi, & M. M. A. Amer. 1981. "Composition of Cocoa Shell Fat as Related to Cocoa Butter." *Zeitschrift Für Ernährungswissenschaft* 20 (2): 145–151. doi:10.1007/BF02021260.
- Fakhlaei, R., Rozzamri A., & N. Hussain. 2019. "Composition, Color and Antioxidant Properties of Cocoa Shell at Different Roasting Temperatures." *Food Research* 4 (3): 585–593. doi:10.26656/fr.2017.4(3).251.
- Farooq, U., X. Liu, & H. Zhang. 2017. "Enhancement of Short Chain Fatty Acid Production by Co-Cultures of Probiotics Fermentation with Pearl Millet (Pennisetum Glaucum) Fibre Fractions." *Journal of Pure and Applied Microbiology* 11 (4): 2031–2038. doi:10.22207/JPAM.11.4.47.
- Ferreira-Lazarte, Alvaro, F.J. Moreno, C. Cueva, I. Gil-Sánchez, & M. Villamiel. 2019. "Behaviour of Citrus Pectin during Its Gastrointestinal Digestion and Fermentation in a Dynamic Simulator (Simgi®)." *Carbohydrate Polymers* 207 (March): 382–390. doi:10.1016/j.carbpol.2018.11.088.
- Forsyth, W. G. C., V. C. Quesnel, & J. B. Roberts. 1958. "The Interaction of Polyphenols and Proteins during Cacao Curing." *Journal of the Science of Food and Agriculture* 9 (3): 181–184. doi:10.1002/jsfa.2740090311.
- Grillo, G., L. Boffa, A. Binello, S. Mantegna, G. Cravotto, F. Chemat, T. Dizhbite, L. Lauberte, & G. Telysheva. 2019. "Cocoa Bean Shell Waste Valorisation; Extraction from Lab to Pilot-Scale Cavitational Reactors." *Food Research International* 115 (January): 200–208. doi:10.1016/j.foodres.2018.08.057.
- Hečimović, I., A. Belščak-Cvitanović, D. Horžić, & D. Komes. 2011. "Comparative Study of Polyphenols and Caffeine in Different Coffee Varieties Affected by the Degree of Roasting." *Food Chemistry* 129 (3): 991–1000. doi:10.1016/j.foodchem.2011.05.059.

- Hernández-Hernández, C., I. Viera-Alcaide, A.M. Morales-Sillero, J. Fernández-Bolaños, & G. Rodríguez-Gutiérrez. 2018. "Bioactive Compounds in Mexican Genotypes of Cocoa Cotyledon and Husk." *Food Chemistry* 240 (February): 831–839. doi:10.1016/j.foodchem.2017.08.018.
- Islamova, Zh. I., D. K. Ogai, O. I. Abramenko, A. L. Lim, B. B. Abduazimov, M. Kh. Malikova, R. K. Rakhmanberdyeva, Z. A. Khushbaktova, & V. N. Syrov. 2017. "Comparative Assessment of the Prebiotic Activity of Some Pectin Polysaccharides." *Pharmaceutical Chemistry Journal* 51 (4): 288–291. doi:10.1007/s11094-017-1600-9.
- Jiménez, A., R. Guillén, J. Fernández-Bolaños, & A. Heredia. 1994. "Cell Wall Composition of Olives." Journal of Food Science 59 (6): 1192–1196. doi:10.1111/j.1365-2621.1994.tb14674.x.
- Jun, H.I., C.H. Lee, G.S. Song, & Y.S. Kim. 2006. "Characterization of the Pectic Polysaccharides from Pumpkin Peel." *LWT - Food Science and Technology* 39 (5): 554– 561. doi:10.1016/j.lwt.2005.03.004.
- Karboune, S., S. Seo, M. Li, A. Waglay, & L. Lagacé. 2022. "Biotransformation of Sucrose Rich Maple Syrups into Fructooligosaccharides, Oligolevans and Levans Using Levansucrase Biocatalyst: Bioprocess Optimization and Prebiotic Activity Assessment." *Food Chemistry* 382 (July): 132355. doi:10.1016/j.foodchem.2022.132355.
- Khodaei, N., & S. Karboune. 2013. "Extraction and Structural Characterisation of Rhamnogalacturonan I-Type Pectic Polysaccharides from Potato Cell Wall." Food Chemistry 139 (1–4): 617–623. doi:10.1016/j.foodchem.2013.01.110.
- Kim, J.H., D.E. Block, S.P. Shoemaker, & D.A. Mills. 2010. "Atypical Ethanol Production by Carbon Catabolite Derepressed Lactobacilli." *Bioresource Technology* 101 (22): 8790– 8797. doi:10.1016/j.biortech.2010.06.087.
- Koropatkin, N.M., E.A. Cameron, & E.C. Martens. 2012. "How Glycan Metabolism Shapes the Human Gut Microbiota." *Nature Reviews Microbiology* 10 (5): 323–335. doi:10.1038/nrmicro2746.
- Kwoji, .D., O.A. Aiyegoro, M. Okpeku, & M.A. Adeleke. 2021. "Multi-Strain Probiotics: Synergy among Isolates Enhances Biological Activities." *Biology* 10 (4): 322. doi:10.3390/biology10040322.
- Larsen, N., C.B. de Souza, L. Krych, W. Kot, T.D. Leser, O.B. Sørensen, A. Blennow, K. Venema, & L. Jespersen. 2019. "Effect of Potato Fiber on Survival of Lactobacillus Species at Simulated Gastric Conditions and Composition of the Gut Microbiota in Vitro." *Food Research International* 125 (November): 108644. doi:10.1016/j.foodres.2019.108644.
- Latimer, G.W., & AOAC International, eds. 2016. *Official Methods of Analysis of AOAC International*. 20th Edition. Gaithersburg, Md: AOAC International.
- LeBlanc, G., F. Chain, R. Martín, L.G. Bermúdez-Humarán, S. Courau, & P. Langella. 2017.
 "Beneficial Effects on Host Energy Metabolism of Short-Chain Fatty Acids and Vitamins Produced by Commensal and Probiotic Bacteria." *Microbial Cell Factories* 16 (1): 79. doi:10.1186/s12934-017-0691-z.
- Lecumberri, E., R. Mateos, M. Izquierdo-Pulido, P. Rupérez, L. Goya, & L. Bravo. 2007.
 "Dietary Fibre Composition, Antioxidant Capacity and Physico-Chemical Properties of a Fibre-Rich Product from Cocoa (Theobroma Cacao L.)." *Food Chemistry* 104 (3): 948– 954. doi:10.1016/j.foodchem.2006.12.054.

- Li, M., M. Heckwolf, J.D. Crowe, D.L. Williams, T.D. Magee, S.M. Kaeppler, N. de Leon, & D.B. Hodge. 2015. "Cell-Wall Properties Contributing to Improved Deconstruction by Alkaline Pre-Treatment and Enzymatic Hydrolysis in Diverse Maize (Zea Mays L.) Lines." *Journal of Experimental Botany* 66 (14): 4305–4315. doi:10.1093/jxb/erv016.
- Li, Y., S. Wang, Y. Sun, H. Zheng, Y. Tang, X. Gao, C. Song, et al. 2020. "Apple Polysaccharide Could Promote the Growth of Bifidobacterium Longum." *International Journal of Biological Macromolecules* 152 (June): 1186–1193. doi:10.1016/j.ijbiomac.2019.10.210.
- Lin, Z., J. Fischer, & L. Wicker. 2016. "Intermolecular Binding of Blueberry Pectin-Rich Fractions and Anthocyanin." *Food Chemistry* 194 (March): 986–993. doi:10.1016/j.foodchem.2015.08.113.
- Liu, C., S. Kolida, D. Charalampopoulos, & R.A. Rastall. 2020. "An Evaluation of the Prebiotic Potential of Microbial Levans from Erwinia Sp. 10119." *Journal of Functional Foods* 64 (January): 103668. doi:10.1016/j.jff.2019.103668.
- Macfarlane, S., & G.T. Macfarlane. 2003. "Regulation of Short-Chain Fatty Acid Production." *Proceedings of the Nutrition Society* 62 (1): 67–72. doi:10.1079/PNS2002207.
- Markowiak-Kopeć, P., & K. Śliżewska. 2020. "The Effect of Probiotics on the Production of Short-Chain Fatty Acids by Human Intestinal Microbiome." *Nutrients* 12 (4): 1107. doi:10.3390/nu12041107.
- Martínez, R., P. Torres, M.A. Meneses, J.G. Figueroa, J.A. Pérez-Álvarez, & M. Viuda-Martos. 2012. "Chemical, Technological and in Vitro Antioxidant Properties of Cocoa (Theobroma Cacao L.) Co-Products." *Food Research International* 49 (1): 39–45. doi:10.1016/j.foodres.2012.08.005.
- Martínez-Cervera, S., A. Salvador, B. Muguerza, L. Moulay, & S.M. Fiszman. 2011. "Cocoa Fibre and Its Application as a Fat Replacer in Chocolate Muffins." *LWT - Food Science* and Technology 44 (3): 729–736. doi:10.1016/j.lwt.2010.06.035.
- Miller, K.B., D.A. Stuart, N.L. Smith, C.Y. Lee, N.L. McHale, J.A. Flanagan, B. Ou, & W. J. Hurst. 2006. "Antioxidant Activity and Polyphenol and Procyanidin Contents of Selected Commercially Available Cocoa-Containing and Chocolate Products in the United States." *Journal of Agricultural and Food Chemistry* 54 (11): 4062–4068. doi:10.1021/jf0602900.
- Mohd Nor, N.A.N., S. Abbasiliasi, M.N. Marikkar, A. Ariff, M. Amid, D.U. Lamasudin, M.Y. Abdul Manap, & S. Mustafa. 2017. "Defatted Coconut Residue Crude Polysaccharides as Potential Prebiotics: Study of Their Effects on Proliferation and Acidifying Activity of Probiotics in Vitro." *Journal of Food Science and Technology* 54 (1): 164–173. doi:10.1007/s13197-016-2448-9.
- Okiyama, D.C.G., S.L.B. Navarro, & C.E.C Rodrigues. 2017. "Cocoa Shell and Its Compounds: Applications in the Food Industry." *Trends in Food Science & Technology* 63 (May): 103–112. doi:10.1016/j.tifs.2017.03.007.
- Payne, M.J., W.J. Hurst, D.A. Stuart, B. Ou, E. Fan, H. Ji, & Y. Kou. 2010. "Determination of Total Procyanidins in Selected Chocolate and Confectionery Products Using DMAC." *Journal of AOAC INTERNATIONAL* 93 (1): 89–96. doi:10.1093/jaoac/93.1.89.
- Puvanasundram, P., C.M. Chong, S. Sabri, M.S. Yusoff, & M. Karim. 2021. "Multi-Strain Probiotics: Functions, Effectiveness and Formulations for Aquaculture Applications." *Aquaculture Reports* 21 (November): 100905. doi:10.1016/j.aqrep.2021.100905.

- Redgwell, R.J., & C.E. Hansen. 2000. "Isolation and Characterisation of Cell Wall Polysaccharides from Cocoa (Theobroma Cacao L.) Beans." *Planta* 210 (5): 823–830. doi:10.1007/s004250050685.
- Redgwell, R.J., V. Trovato, S. Merinat, D. Curti, S. Hediger, & A. Manez. 2003. "Dietary Fibre in Cocoa Shell: Characterisation of Component Polysaccharides." *Food Chemistry* 81 (1): 103–112. doi:10.1016/S0308-8146(02)00385-0.
- Rojo-Poveda, O., L. Barbosa-Pereira, G. Zeppa, & C. Stévigny. 2020. "Cocoa Bean Shell—A By-Product with Nutritional Properties and Biofunctional Potential." *Nutrients* 12 (4): 1123. doi:10.3390/nu12041123.
- Romanens, E., R. Näf, T. Lobmaier, V. Pedan, S.F. Leischtfeld, L. Meile, & S. M. Schwenninger. 2018. "A Lab-Scale Model System for Cocoa Bean Fermentation." *Applied Microbiology and Biotechnology* 102 (7): 3349–3362. doi:10.1007/s00253-018-8835-6.
- Saint-Denis, T., & J. Goupy. 2004. "Optimization of a Nitrogen Analyser Based on the Dumas Method." *Analytica Chimica Acta* 515 (1): 191–198. doi:10.1016/j.aca.2003.10.090.
- Schell, M.A., M. Karmirantzou, B. Snel, D. Vilanova, Bernard Berger, G. Pessi, M.C. Zwahlen, et al. 2002. "The Genome Sequence of Bifidobacterium Longum Reflects Its Adaptation to the Human Gastrointestinal Tract." *Proceedings of the National Academy of Sciences* 99 (22): 14422–14427. doi:10.1073/pnas.212527599.
- Shoaib, M., A. Shehzad, M. Omar, A. Rakha, H. Raza, H.R. Sharif, A. Shakeel, A. Ansari, & S. Niazi. 2016. "Inulin: Properties, Health Benefits and Food Applications." *Carbohydrate Polymers* 147 (August): 444–454. doi:10.1016/j.carbpol.2016.04.020.
- Singleton, Vernon L., Rudolf Orthofer, & Rosa M. Lamuela-Raventós. 1999. "[14] Analysis of Total Phenols and Other Oxidation Substrates and Antioxidants by Means of Folin-Ciocalteu Reagent." In *Methods in Enzymology*, 299:152–178. Elsevier. doi:10.1016/S0076-6879(99)99017-1.
- Slavin, J. 2013. "Fiber and Prebiotics: Mechanisms and Health Benefits." *Nutrients* 5 (4): 1417–1435. doi:10.3390/nu5041417.
- Soeharsono, M Amin, & A Cahyono. 2017. "The Use of Cocoa Bean Waste as a Supplement in Male Bali Cattle Feeding." In *Proceedings of International Seminar on Livestock Production and Veterinary Technology*. Indonesian Center for Animal Research and Development (ICARD). doi:10.14334/Proc.Intsem.LPVT-2016-p.45-49.
- Spadoni Andreani, E., & S. Karboune. 2020. "Comparison of Enzymatic and Microwaveassisted Alkaline Extraction Approaches for the Generation of Oligosaccharides from American Cranberry (Vaccinium Macrocarpon) Pomace." *Journal of Food Science* 85 (8): 2443–2451. doi:10.1111/1750-3841.15352.
- Spadoni Andreani, E., S. Karboune, & L. Liu. 2021. "Extraction and Characterization of Cell Wall Polysaccharides from Cranberry (Vaccinium Macrocarpon Var. Stevens) Pomace." *Carbohydrate Polymers* 267 (September): 118212. doi:10.1016/j.carbpol.2021.118212.
- Spadoni Andreani, E., M. Li, J. Ronholm, & S. Karboune. 2021. "Feruloylation of Polysaccharides from Cranberry and Characterization of Their Prebiotic Properties." *Food Bioscience* 42 (August): 101071. doi:10.1016/j.fbio.2021.101071.
- Sudha, M.L., V. Baskaran, & K. Leelavathi. 2007. "Apple Pomace as a Source of Dietary Fiber and Polyphenols and Its Effect on the Rheological Characteristics and Cake Making." *Food Chemistry* 104 (2): 686–692. doi:10.1016/j.foodchem.2006.12.016.

- Vojvodić, A., D. Komes, I. Vovk, A. Belščak-Cvitanović, & A. Bušić. 2016. "Compositional Evaluation of Selected Agro-Industrial Wastes as Valuable Sources for the Recovery of Complex Carbohydrates." *Food Research International* 89 (November): 565–573. doi:10.1016/j.foodres.2016.07.023.
- Vriesmann, L.C., R. de Mello Castanho Amboni, & C. de Oliveira Petkowicz. 2011. "Cacao Pod Husks (Theobroma Cacao L.): Composition and Hot-Water-Soluble Pectins." *Industrial Crops and Products* 34 (1): 1173–1181. doi:10.1016/j.indcrop.2011.04.004.
- Xiao, M., J. Yi, J. Bi, Y. Zhao, J. Peng, C. Hou, J. Lyu, & M. Zhou. 2018. "Modification of Cell Wall Polysaccharides during Drying Process Affects Texture Properties of Apple Chips." *Journal of Food Quality* 2018: 1–11. doi:10.1155/2018/4510242.
- Yang, J., I. Martínez, J. Walter, A. Keshavarzian, & D.J. Rose. 2013. "In Vitro Characterization of the Impact of Selected Dietary Fibers on Fecal Microbiota Composition and Short Chain Fatty Acid Production." *Anaerobe* 23 (October): 74–81. doi:10.1016/j.anaerobe.2013.06.012.
- Younes, A., M. Li, & S. Karboune. 2022. "Cocoa Bean Shells: A Review into the Chemical Profile, the Bioactivity and the Biotransformation to Enhance Their Potential Applications in Foods." *Critical Reviews in Food Science and Nutrition*, April, 1–25. doi:10.1080/10408398.2022.2065659.
- Zhu, J., F. Guo, C. Ma, H. Wang, J. Wen, & Y. Yu. 2022. "The Alkaline Extraction Efficiency of Bamboo Cell Walls Is Related to Their Structural Differences on Both Anatomical and Molecular Level." *Industrial Crops and Products* 178 (April): 114628. doi:10.1016/j.indcrop.2022.114628.
- Zykwinska, A., C. Rondeau-Mouro, C. Garnier, J.F. Thibault, & M.C. Ralet. 2006. "Alkaline Extractability of Pectic Arabinan and Galactan and Their Mobility in Sugar Beet and Potato Cell Walls." *Carbohydrate Polymers* 65 (4): 510–520. doi:10.1016/j.carbpol.2006.02.012.

CONNECTING STATEMENT II

This is an accepted manuscript of an article published MDPI Polymers, available at https://www.mdpi.com/2073-4360/15/3/745.

In chapter III, the composition of cocoa bean shells was analyzed, and the extraction of their cell wall polysaccharides was investigated using alkaline treatments. The obtained extracts were characterized by the presence of monosaccharide residues associated to both pectic and hemicellulosic polysaccharides, in variable proportions. The prebiotic properties of polysaccharides and the short chain fatty acid composition were determined, showing optimal prebiotic activity. The enzymatic generation of oligosaccharides has been assessed on other valorized by-products; however, those of CBS have received little attention. CBS polysaccharides were hydrolyzed into respective oligosaccharides of varying degrees of polymerization using a selection of multi-enzymatic products based on their galactanase activity. The oligosaccharide extracts were characterized in terms of their degrees of polymerization, prebiotic activity, and short chain fatty acid production. Furthermore, the feruloylation of oligo- and polysaccharide extracts from cocoa bean shells was assessed and characterized in terms of their feruloylation efficiency as well as their antioxidant capacities, based on three antioxidant assays.

The results of this study have been submitted for publication.

CHAPTER IV. ENZYMATIC GENERATION OF OLIGOSACCHARIDES FROM COCOA BEAN SHELLS USING MULTI-ENZYMATIC PRODUCTS AND THE FERULOYLATION OF OLIGO/POLYSACCHARIDES

4.1 Abstract

Cell wall polysaccharides isolated from two varieties (NAC, CCN) of cocoa bean shell (CBS) consisted of high pectic regions, primarily rhamnogalacturonan, and hemicelluloses. Enzymatic hydrolysis of CBS polysaccharides was performed using bi-enzymatic (arabinase/galactanase) and multi-enzymatic systems. Viscozyme® L led to the highest yield (76.64%) of NAC-based oligosaccharides of DP 3-5 (42.04%, mol) and 6-10 (36.78%, mol). The hydrolysis of CCN-based polysaccharides by DepolTM 670L led to 20.72% yield, while generating a diverse complex of oligosaccharides of DP up to 10. When tested for the promotion of the growth of selected probiotic strains, Lactobacillus rhamnosus GG and Bifidobacterium longum, oligosaccharides performed similarly or more than inulin and rhamnogalacturonan based on the prebiotic activity scores. In addition, the production of lactic, acetic, butyric and traces of propionic were prominent. Free feruloyl esterases from rumen microorganism provided best feruloylation efficiencies with CBS polysaccharides (56.9-60.1%), while that from H. insolens provided best feruloylation efficiencies with the oligosaccharides (30.8-50.5%). The antioxidant activity of feruloylated polysaccharides were expressed using three assays, DPPH, ORAC, and ABTS. Feruloylated CBS polysaccharides prompted significantly notable increases in the scavenging and antioxidant activity compared to their native forms as well as ferulic and ascorbic acid. Overall, CBS oligo- and feruloylated polysaccharides show great potential for use as optimal functional food ingredients.

4.2 Introduction

Cocoa bean shells (CBS) contain an abundant source of dietary fibers, consisting of about 45% pectic, 35% cellulosic, and 20% hemi-cellulosic polysaccharides (Redgwell et al., 2003). As a result, CBS' cell wall composition makes it a desirable source for the extraction and generation of non-digestible oligosaccharides. Oligosaccharides are naturally present in foods in very low amounts and can be produced through chemical or enzymatic approaches (Du et al., 2011; Holck, Hjernø, et al., 2011; Lama-Muñoz et al., 2012; Martínez Sabajanes et al., 2012). For instance, the use of multi-enzymatic systems, which comprise of a mixture of selected glycosyl-hydrolases working in synergy, has been proven as a potential approach for the generation of non-digestible oligosaccharides from complex biomasses, including cranberry pomace, potato pulp, sugar beet pulp, orange peel (Spadoni Andreani & Karboune, 2020; Waglay & Karboune, 2017; Khodaei & Karboune 2016; Holck, Hjernø, et al., 2011; Martínez Sabajanes et al., 2012). Multi-enzymatic systems, namely Pectinex® Ultra Mash, Depol™ 670L, Pectinex® Ultra Clear, Viscozyme® L, Viscomyl® Flow, have been used for a variety of commercial applications, such as juice clarification, viscosity reduction, and maceration (Waglay & Karboune, 2017).

Furthermore, the enzymatically generated non-digestible oligosaccharides from cell wall polysaccharides have shown to function as great prebiotics. From previous findings in par 3.4.4 when tested for the promotion of the growth of *Lactobacillus rhamnosus GG* and *Bifidobacterium*. *longum* probiotic strains, CBS cell wall polysaccharides showed significant prebiotic activity and hence ability to produce short chain fatty acids, namely lactic and acetic acid. In addition, Du et al. (2011) found that non-digestible oligosaccharides, prompted both higher solubility and prebiotic responses compared to their polysaccharide fractions from pumpkin pulp. We hypothesize that CBS cell wall non-digestible oligosaccharides would also act as prebiotics, while favoring faster metabolization by probiotic strains and contributing to the sweetness ability and techno-functionalities.

Feruloylated oligo- and polysaccharides exist in nature within the plant cell wall; however, their recovery is limited due to their low yield and structural inconsistencies (Tamayo-Cabezas & Karboune, 2019). Alternatively, enzymatic acylation of ferulic acid with cell wall oligo- and polysaccharides has shown great potential, as it has exhibited higher bioconversion yields as well as the generation of reproducible structures with phenolated moieties (Couto et al., 2011; Spadoni

Andreani et al., 2021; Tamayo-Cabezas & Karboune, 2019). This reaction, otherwise, known as feruloylation, can be catalyzed by feruloyl esterases (E.C. 3.1.1.73), which are a subclass of carboxylic ester hydrolases known for their catalytic efficiency for the release of hydroxycinnamic acids from the plant cell wall (Spadoni Andreani et al., 2021; Tamayo-Cabezas & Karboune, 2019). Feruloylation of oligo- and polysaccharides or their isolation from cell wall materials have been previously explored with cranberry pomace (Spadoni Andreani et al., 2021), sugar beet (Zaidel et al., 2011), and cultured rice (Mastihubová et al., 2006). To the author's knowledge, feruloylation of CBS oligo- and polysaccharides has yet to be investigated . One approach to enhance the feruloylation efficiency is enzyme immobilization. This technique can enhance the enzymatic synthesis of feruloylated compounds, through optimizing enzyme activity in specific macroenvironments while also ensuring enzyme stability and reusability (He et al., 2015; Spadoni Andreani et al., 2021; Tamayo-Cabezas & Karboune, 2019; Thörn et al., 2011). Ultimately, feruloylated oligo- and polysaccharides have been idealized for their combined prebiotic and antioxidant activities, as ferulic acid serves as a functional and bioactive phenolic acid (Couto et al., 2010).

In the present study, the enzymatic generation of CBS oligosaccharides using bi-enzymatic and multi-enzymatic systems, was investigated. The generated oligosaccharides were structurally characterized, and their prebiotic activity was assessed and correlated with the production of shortchain fatty acids. To our knowledge, studies on the recovery of oligosaccharides from CBS using multi-enzymatic systems have not been reported so far. In addition, this study inquired the feruloylation efficiencies of CBS polysaccharides and their corresponding enzymatically generated oligosaccharides using feruloyl esterases from different microbial sources, rumen microorganism and *Humicola insolens*. The assessment of their antioxidant activities was determined using DPPH, ORAC, and ABTS assays.

4.3 Materials and Methods

4.3.1 Materials

Commercial samples of CBS (Nacional/Arriba (NAC) and CCN 51) were supplied from Cocoa processing industry (Guayaquil, Ecuador) and blended using a Model 7011C commercial blender (Conair corporation, Stamford, CT, USA). Depol[™] 670L (from *Trichoderma reesei*) and Feruloyl esterase (from *H. insolens*) were provided by Biocatalysts Ltd. (Mid Glamorgan, UK).

Pectinex® Ultra Mash, Viscozyme® L, and Viscomyl® Flow were from Novozymes, endo-1,4β-galactanse, endo-1,5-α-arabinanase, soybean RG, and Feruloyl Esterase (from rumen microorganism) were from Megazyme (Wicklow, Ireland). 1-Kestose, nytose, and 1Ffructofuranoysl-nystose were obtained from Wake Pure Chemical (Osaka, Japan). The bacterial strains, L. rhamnosus GG and B. longum (ATCC 15707TM) strains were supplied from Lallemand and ATCC[©]. Pectinex[®] Ultra Clear and the analytical grade reagents were from Sigma-Aldrich Co., St. Louis, MO, USA. Salts were obtained from Fisher Scientific (Fair Lawn, NJ). 2,2-(>90%), (96%), diphenyl-1-picrylhydrazyl (DPPH) α -tocopherol 2,2'-Azobis(2methylpropionamidine) dihydrochloride (AAPH), fluorescein sodium salt, and 6-hydroxy-2,5,7,8tetramethylchroman-2-carboxylic acid (Trolox) were purchased from MilliporeSigma (Burlington, Massachusetts, United States).

4.3.2. Alkaline extraction of CBS polysaccharides

Extraction of CBS polysaccharides was performed as described by Khodaei and Karboune (2013) where CBS cell wall material (CWM, 2% w/v) was suspended in KOH solutions (0.5M) containing 0.02M NaBH₄. Mixtures were incubated (60°C, 24 hr), centrifuged (10,000 x g for 15 mins), and recovered by filtration (0.45 μ m). Recovered polysaccharides were neutralized with HCl, dialyzed (cut-off 5-8 kDa) and freeze-dried (-50°C, 0.003 mbar) for further analysis.

4.3.3. Bio-generation of oligosaccharides from CBS polysaccharides

The time course for the hydrolysis of CBS polysaccharides was carried out over reaction times of 1, 3, 7 and 24hr as described by Khodaei and Karboune (2016). Pectic polysaccharide suspensions in sodium acetate buffer (50mM, pH 5.0) at a concentration of 1% (w/v) were prepared. The multi-enzymatic preparations (DepolTM 670L, Viscozyme® L, Pectinex® Ultra Mash, Pectinex® Ultra Clear, Viscomyl® flow), and arabinase:galactanase bienzymatic system were added to the CBS polysaccharide suspensions to yield final enzyme/substrate ratio of 50 and 100 U galactanase /g substrate. After incubation at 40°C, the reaction mixtures were boiled for 5 min to inactivate the enzyme and subsequently centrifuged (10,000g, 10 min). Oligosaccharides (DP of 2–12) and oligomers (DP of 3–70) were quantified by HPSEC. Yield was defined as the number of generated oligosaccharides/oligomers over the initial amount of RG I-type pectic polysaccharides. The MW distribution of generated oligosaccharides was also determined by high

performance size exclusion chromatography (HPSEC), and their monosaccharide profile was determined by high performance anionic exchange chromatography (HPAEC).

4.3.4. Analytical methods for the quantification of oligo- and polysaccharide extracts 4.3.4.1. Determination of total neutral sugars and uronic acid contents

Uronic acid content was measured by sulphamate/m-hydroxydiphenyl assay (Blumenkrantz & Asboe-Hansen, 1973). The phenol–sulphuric acid colorimetric assay was used for the determination of neutral sugar content (DuBois et al., 1956).

4.3.4.2. Molecular weight distribution profile

A HPSEC system (Model 1525 binary HPLC pump, equipped with a Model 2414 refractive index detector, Waters Co., Milford, MA, USA) was used to estimate the molecular weight distribution of the carbohydrates. Oligosaccharide analysis was determined using a TSK G-Oligo-PW column (Tosoh Co., Yamaguchi, Japan) column. Galactose, sucrose, 1-kestose, nystose and 1F-fructofuranosyl nystose standards were used for molecular weight and concentration calibration. The temperature of the system was set at 30 °C, the eluent was 0.2 M NaCl, and the flow rate was set at 0.4 ml/min. Polysaccharide analysis was determined using a TSK G5000 PWXL (Tosoh Co., Yamaguchi, Japan) column, with dextrans (50– 670 kDa), soybean rhamnogalacturonan (0.125-1 g/l) and galactoglucomannan (0.125-1g/l) standards for molecular weight and concentration. The temperature of the system was 30 °C, the eluent was 0.1 M NaCl, and the flow rate was set at 0.4 ml/min.

4.3.5. Preparation of feruloyl esterase and feruloylation of oligo- and polysaccharides4.3.5.1. Immobilization of feruloyl esterase

Prior to immobilization, epoxy-activated Sepabeads® EC-EP R were modified by suspending them (0.2g/l) in a solution of 1.8M iminodiacetic acid (IDA) brought to alkaline pH with sodium hydroxide (pH 9) and shaken for 5hr at 55 rpm in Excella E24 orbital shaker (New Brunswick Scientific, Edison, NJ, USA) at 25°C. Excess reagents were removed by washing the supports (0.2g/l) eight times with water. After each wash, the supports were vacuum filtered on a fritted glass filter. The supports were then suspended (0.17g/l) in cupric sulfate (0.13M) for 2hr under the same conditions as the previous step. Activated supports were then washed (0.2g/l) eight times in 3-morpholinopropane-1-sulfonic acid (MOPS) buffer (0.02M, pH 6) and

stored in MOPS buffer at 4°C. Modified IDA-Cu-Sepabeads® EC-EP R support was washed with 0.2 M potassium phosphate buffer at pH 8 according to the optimized conditions for immobilization determined in our previous works (Tamayo-Cabezas and Karboune, 2019). The supports were then suspended (0.2 g/l) in the same buffers containing 0.01 g of enzyme per gram of support. The suspensions were kept for 28 to 36 hr at 6 °C under gentle shaking. Supernatants were recovered by centrifugation at $2400 \times g$ for 5 min, and the supports, containing immobilized enzyme, were washed (0.2 g/l) five times with MOPS buffer (0.02 M, pH of 6) and stored at 4 °C. Immobilization blanks were prepared by incubating the enzyme solutions in the absence of support. Protein contents of the initial enzyme solutions, the supernatants and the washing solutions were determined using methyl ferulate substrate as described by Tamayo-Cabezas and Karboune (2019). Immobilization yield was calculated as the difference between activity in the immobilization blank and the activities in supernatant and washing solutions, divided by the activity of the initial enzyme solution.

4.3.5.2. Esterification activity of free and immobilized feruloyl esterase

The esterification activity of feruloyl esterase (from rumen microorganism and *H. insolens)* was carried out according to the method reported by Spadoni Andreani et al. (2021) with raffinose and ferulic acid as substrates at a concentration of 1.5 and 4.5 mM, respectively. A surfactant-less microemulsion made of n-hexane:butanone:water was evaluated as a reaction medium at the selected ratio of 49.4:44.6:6 (v/v/v). In vacuum sealed flasks, free or immobilized feruloyl esterase were added to each emulsion to reach 10 g/l. The reaction mixtures were incubated for 24 hr at 25 °C under 70 rpm. Negative controls without enzyme and without raffinose were run in parallel. Reaction progress was monitored through the consumption of ferulic acid, determined by high performance liquid chromatography (HPLC) on Beckman 32 Karat system (Beckman Coultier) equipped with a binary pump (Model 1525), and a photodiode array detector (Model 2998). The reaction mixture was loaded on a Zorbax SB-C18 reversed-phase column (5 μ m, 250×4.6 mm, Agilent Technologies Canada Inc., Mississauga, Ca) and eluted with a linear gradient starting with 100% mobile A (water:acetonitrile:formic acid, 75.6:20:4.4 v/v/v) to 50:50 mobile A: mobile B (100% acetonitrile) for 20 min, followed by a linear gradient for 10 min until 100% mobile A is

reached. A constant flow rate of 0.7 ml/min was applied, and the absorbance was continually monitored at 320 nm. Ferulic acid standard was used to construct the calibration curve (R=0.95). One unit of esterification activity was defined as the amount of enzyme that consumes one micromole of ferulic acid per minute.

4.3.5.3. Feruloylation of oligo- and polysaccharides

The feruloylation of CBS oligo- and polysaccharides was carried out according to the method described by Spadoni Andreani et al. (2021) in the n-hexane:butanone:water surfactant-less microemulsion at the ratio of 49.4:44.6:6 (v/v/v). The enzymatic feruloylation reaction was initiated by adding the free feruloyl esterase (2 U/l) into the microemulsion containing ferulic acid (1.5 g/l) and CCN or NAC polysaccharide extracts and their corresponding oligosaccharide extracts (0.3 g/l) generated using DepolTM 670L (CCN variety) and Viscozyme L (NAC variety). The enzymatic reactions were conducted in vacuum sealed flasks at 25 °C for 144 hr, with continuous shaking at 70 rpm. The mixtures were then centrifuged at 2400 × g for 3 min and the feruloylated oligo- and polysaccharides (FA-OCCN, FA-ONAC, FA-PCCN, FA-PNAC) were isolated from the supernatants by extraction with water. The aqueous phase was dialyzed (MWCO 5-8 kDa) and freeze dried. The reaction yield was calculated from the decrease of ferulic acid concentration in the reaction supernatants, measured by HPLC.

4.3.6. Functional properties

4.3.6.1. Prebiotic activity assay

CBS Oligosaccharides (CCN-Depol 670L and NAC-Viscozyme L) were investigated as a carbon source for *B. longum* (ATCC 15707TM) and *L. rhamnosus GG* (Harmonium AR) growth compared to growth on inulin and rhamnogalacturonan, using the method reported by Yang et al. (2013), modified to consider the growth requirements of the selected strains. *B. longum* and *L. rhamnosus GG* were maintained in an anaerobic chamber at 37°C. Pre-reduced reinforced clostridial broth was used for the reanimation of *B. longum* from a freeze-dried stock, and MRS broth was used for *L. rhamnosus GG*. An aliquot (0.5 ml) of the primary cultures was then added into tubes containing 5 ml of pre-reduced, carbohydrate-free MRS broth, supplemented with 0.4% glucose. Bacterial growth was measured using colony counting on Tryptic soy agar supplemented with 5% sheep blood for *B. longum* and pre-reduced MRS agar for *L. rhamnosus GG*. Tubes containing 5 ml of pre-reduced carbohydrate-free MRS broth supplemented with 0.2% glucose,

inulin, rhamnogalacturonan, and cell wall oligosaccharides (CCN and NAC varieties) were inoculated with 0.1ml of secondary culture. Colonies were counted on Tryptic soy agar supplemented with 5% sheep blood for *B. longum* and pre-reduced MRS agar for *L. rhamnosus GG* after 6, 24, 48, 72 and 96 hr incubation intervals. Microbial counts were measured as colony forming units (CFU)/ml, and the resulting averaged data were transformed to log CFU/ml. A prebiotic activity score (PA) was calculated for each combination of bacterium and oligosaccharide as follows:

$$PA = \frac{\left(Log\frac{CFU}{ml}oligosaccharide at x hr - Log\frac{CFU}{ml}on oligosaccharide at 0hr\right)}{Log\frac{CFU}{ml}on glucose at x hr - Log\frac{CFU}{ml}on glucose at 0hr}$$

4.3.6.2. Determination of short chain fatty acid catabolites

An aliquot of cultures (0.5ml) was taken at 24 and 48hr. Bacteria were precipitated by centrifugation at 1400xg for 5min at 37°C in a Minispin Plus centrifuge (Eppendorf AG, Hamburg, Germany). The supernatants were filtered using 0.22 μ m polytetrafluorethylene syringe filters (Thermo Fischer Scientific) and stored at -80°C for a maximum of 2 weeks. Freshly thawed samples (0.02ml) were analyzed using the HPLC Model 1525 system (Waters) equipped with a refraction index detector and a Zorbax SB-C18 column (4.6x 250mm) (Agilent Technologies). Elution was carried out with water containing 0.005N sulfuric acid at a constant flow rate of 0.6ml/min. Standards were prepared by serial dilution of stock solutions of lactic, acetic, propionic, and butyric acids in water (R = 0.95).

4.3.6.3. Total phenolic content

CBS polysaccharides (NAC and CCN) were characterized in terms of total phenolic content by Folin- Ciocalteu colorimetric method (Singleton et al., 1999). To 0.8 ml of aqueous solution of the sample, 0.15 ml of sodium bicarbonate (250 g/l) and 0.05 ml of Folin-Ciocalteu reagent were added. The mixtures were heated at 40 °C for 30 min, and their absorbance was measured at 765 nm. Gallic acid standard solutions (0.001 M to 0.1 M) were used for calibration (R = 0.95).

4.3.6.4. Antioxidant Activity

4.3.6.4.1. 2,2-Diphenyl-1-picrylhydrazyl (DPPH) Radical Scavenging Assay

The DPPH assay was carried out according to the method reported by Srikanth et al. (2015) with modifications. 25 µl aliquots of varying concentrations of feruloylated (in 50% methanol) and native polysaccharides (in water, pH 7.4) were mixed with 25ul of DPPH (0.1mM), which were then incubated in the dark for 30 min at 25°C. In preliminary trials, 30 min was identified as the appropriate time at which the steady state of scavenging was achieved. Following incubation, the absorbance of each sample was read at 517nm using a Synergy HTX Multi-Mode Reader. Water was used as the blank. 50% methanol in water was used as the negative control. Ascorbic and ferulic acids prepared in 100% methanol were used as positive controls. The inhibition ratio (%) was calculated as follows:

Inhibition ratio (%) =
$$[(Ac - As)/Ac] \times 100$$

Where: Ac is the absorbance of the control; As is the absorbance of the sample The different concentration of feruloylated and native oligo- and polysaccharides were plotted against the estimated inhibition ratio. The IC_{50} was defined as the concentration required to reduce DPPH by 50%. The IC_{50} was expressed as ug sample/ml DPPH.

4.3.6.4.2. Oxygen Radical Absorption Capacity (ORAC) Assay

The ORAC assay was carried out according to the method outlined by Cao et al. (1993), with modifications. AAPH (2,2'-azobis(2-amidinopropane) dihydrochloride) was used as a free-radical generator and varying concentrations of selected feruloylated and native polysaccharides were used to prevent the decay of Fluorescein Sodium Salt. Trolox (\pm)-6-Hydroxy-2,5,7,8-tetramethylchromane-2-carboxylic acid (100, 50, 25, 12.5, 6.25, and 0mM) was used as a standard and phosphate buffer (75mM, pH 7.4) was used as a blank. Ascorbic and ferulic acids prepared in 100% methanol were used as positive controls. Aliquots of 25µl of feruloylated and native polysaccharides at different concentrations (Phosphate buffer pH 7.4) and 150µl of 16 mM fluorescein (prepared in PBS 10mM pH 7.4) was pipetted into a 96-well black-walled plate. Following incubation at 37°C for 30 min in a Synergy HTX Multi-Mode Reader, all wells were injected with 25µl of freshly prepared AAPH (79.65mmol/l). Fluorescence readings were taken for 1 hr at 485nm (excitation wavelength) and 520nm (emission wavelength) every 60 seconds.

The AUC (area under the curve) and Net AUC of the standards and samples were determined using Gen5 Data Analysis Software using the following equations respectively:

AUC =
$$\left(\frac{R_1}{R_1}\right) + \left(\frac{R_2}{R_2}\right) + \left(\frac{R_3}{R_3}\right) + \dots + \left(\frac{R_n}{R_n}\right)$$

Where: R₁ is the fluorescence reading at the initiation of the reaction; R_n is the last measurement

Net AUC =
$$AUC_{sample} - AUC_{blank}$$

The standard curve consisted of the Net AUC of different Trolox concentrations plotted against their concentration. Finally, to calculate the Trolox Equivalents (TE) of each sample range, the following equation was used:

TE (range of concentrations) = $m_{compound}/m_{Trolox}$

Where: $m_{compound}$ is the slope of the linear regression analysis of the compound; m_{Trolox} is the slope of the linear regression analysis of Trolox. Results are expressed as µmol Trolox Equivalents (TE)/g sample.

4.3.6.4.3. 2,2'-azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) decolorization assay

The ABTS assay was carried out according to the method described by Re et al. (1999) with modifications. ABTS radical cation (ABTS⁺⁺) was produced by reacting ABTS stock solution (7 mM) with 2.45 mM potassium persulfate (final concentration) and allowing the mixture to stand in the dark at room temperature for 12–16 hr before use. The ABTS⁺⁺ solution was diluted with ethanol to an absorbance of 0.70 (\pm 0.02) at 734 nm and equilibrated at 25°C. Stock solutions of feruloylated and native polysaccharides were prepared in 50% ethanol and phosphate buffered saline (PBS, pH 7.4). After addition of 1.0 ml of diluted ABTS⁺⁺ solution (A_{734nm} = 0.700 \pm 0.020) to 10 µl of sample or Trolox standards (final concentration 0–15 µM) in ethanol or PBS the absorbance reading was taken at room temperature immediately after mixing and up to 6 min. Appropriate solvent blanks were run in each assay, ascorbic and ferulic acid were used as positive controls. All determinations were carried out at least three times, and in triplicate, on each occasion and at each separate concentration of the standard and samples. The percentage inhibition of absorbance at 734 nm is calculated and plotted as a function of concentration of antioxidants and of Trolox for the standard reference data.

4.3.7. Statistical analysis

Statistical analyses were performed using XLSTAT software (Addinsoft, New York, NY, USA) in Microsoft Excel (Microsoft, Redmond, WA, USA). One-way analysis of variance (ANOVA) and the Tukey's honest significant difference (HSD) test were performed to detect significant differences (P < 0.05).

4.4 Results and Discussion

4.4.1 Bio-generation of oligosaccharides from cocoa bean shells

4.4.1.1 Composition of cocoa bean shell polysaccharides

Two polysaccharide extracts, NAC and CCN, were isolated from CBS CWM and analysed. As shown in Table 4.1, the extracts were recovered at a yield of 29.1 and 35.4% (w/w) for the NAC and CCN varieties, respectively. Both polysaccharide extracts showed three main populations of poly- and oligomers with different molecular weight distributions (NAC - 3.7, 12, 1300 kDa; CCN - 4.5, 9.1, 325 kDa). As for their monosaccharide profile, the NAC extract is predominantly comprised of pectic polysaccharides due to the presence of uronic acid (58.24 % mol), galactose (17.54 % mol), arabinose (7.75 % mol), and rhamnose (6.81 % mol). Alternatively, the CCN extract, showed a monosaccharide profile abundant in uronic acid (51.44 % mol), mannose and xylose (12.56 % mol), rhamnose (12.38 % mol), and galactose (8.09 % mol), indicating the presence of both pectic and hemi-cellulosic polysaccharides. Furthermore, both extracts contain a notable total phenolic amount (79.7-113.6 x10⁻⁶ mol GAE /g), higher than the amounts reported for cranberry extracts (31.9 - 80.8 x10⁻⁶ mol GAE /g, Spadoni Andreani et al., 2021) as well as other by-products, namely coffee silver-skin (76.4 x10⁻⁶ mol GAE/g, Ballesteros et al., 2014), and purple potato peel (42.3 x10⁻⁶ mol GAE /g, Albishi et al., 2013). Based on previous findings in par 3.4.2, the class of polyphenols predominantly present in CBS CWM extracts were flavan-3-ols, namely epicatechin, catechin. Other phenolics compounds were found which include quercetin and phenolic acids, such as gallic acid, protocatechuic, p-coumaric, ferulic, and caffeic acids.

	NAC	CCN
Polysaccharide yield (%, w/w)	35.4±4.3	29.1±0.4
Molecular weight of polysaccharide (PS) (kDa)	3.7, 12, 1300	4.5, 9.1, 325
Monosaccharide profile (% mol)		
Fucose	$1.76\pm0.14^{\rm a}$	$1.75\pm0.02^{\mathtt{a}}$
Rhamnose	$6.81 \pm 1.05^{\text{b}}$	$12.38\pm0.31^{\text{a}}$
Arabinose	$7.75\pm0.56^{\text{ b}}$	$8.58\pm0.01^{\text{b}}$
Galactose	$17.54 \pm 1.26^{\text{b}}$	$8.09\pm0.2^{\circ}$
Glucose	2.86 ± 0.13^{ab}	$5.19\pm0.5^{\rm a}$
Mannose, Xylose	$5.04\pm0.37^{\text{b}}$	$12.56\pm2.04^{\circ}$
Uronic acids	58.24 ± 2.76^{b}	51.44 ± 3.06^{ab}
Total phenolic content (10 ⁻⁶ mol GAE/g)	113.36±0.01ª	79.69±0.01 ^b

Table 4.1. Characterization of cocoa bean shell polysaccharide extracts

Values reported as average ± standard deviation *Standard deviation zero or not detected ^a Within the same row, means with different letters are significantly different at P < 0.05

4.4.2 Enzymatic generation of oligosaccharides from cocoa bean shells polysaccharides

Enzymatic hydrolysis of cell wall pectic polysaccharides from CBS was investigated in order to generate oligosaccharides. Polysaccharides in CBS were reported to be namely rhamnogalacturonan, xylogalacturonan, and glucuronoxylan heavily branched and linked to other polysaccharides (Redgwell et al., 2000). Six biocatalysts, including the bi-enzymatic system (arabinase/galactanase, 50:50 v/v) and multi-enzymatic preparations (Viscozyme® L, Pectinex® Ultra Mash, Pectinex ® Ultra Clear, Viscomyl ® Flow, DepolTM 670L) with varied levels of polysaccharide-hydrolyzing activities (including cellulase, polygalacturonase, endogalactanase, arabinase, galactanase, rhamnogalacturonase, xylanase), were screened for their ability to release oligosaccharides from CBS polysaccharides.

Provided that the cell wall polysaccharides were predominantly pectic ones, the selection of the multi-enzymatic preparations was based on their galactanase and pectinolytic activities (50 and 100U/g) as previously reported by Khodaei and Karboune (2016) and Spadoni Andreani and Karboune (2020). Table 7 and Table 4.8 show the hydrolysis extents as well as the MW distribution (DP 2-10 and DP > 10) of enzymatically generated oligosaccharides after 7 hr of hydrolysis. The highest hydrolysis extents were observed with Viscozyme® L (76.64%, 50U/g) for NAC-based oligosaccharides (Table 4.2). For CCN-based oligosaccharides, the highest hydrolysis extents were achieved with Pectinex® Ultra Mash (21.31%, 50U/g), followed by Depol® 670L (20.72%, 50U/g) and Viscomyl® Flow (20.31%, 50U/g) (Table 4.3). The enzymatically generated oligosaccharides from the bi-enzymatic system and the multi-enzymatic products varied in terms of their degrees of polymerization (DP). As seen in Table 7, all biocatalysts achieved a high proportion of NAC-based oligosaccharides of DP ≤ 2 (95.55-99.51%, mol) with the exception of Viscozyme® L (20.86%, mol). Indeed, Viscozyme® L generated diverse complexity of oligosaccharides of varying DP, where the highest proportion of oligosaccharides were of DP 3-5 (42.04%, mol). The results also show that NAC-based oligosaccharides of DP 6-10 were generated by Viscozyme® L (50 U/g, 36.78%, mol), while those of DP greater than 10 were only generated by Pectinex® Ultra Mash (1.00%, mol), Pectinex® Ultra Clear (100 U/g, 0.49%, mol) and Viscozyme[®] L (0.32%, mol).

	DP> 10	DP 6-10	DP 3-5	$DP \lesssim 2$	Hydrolysis extent (%)
Arabinase/Galactanase ^g	0.00 ^d	2.08 ^b	2.37 ^b	95.55^{f}	17.10
Arabinase/Galactanase ^h	0.00 ^d	0.00^{f}	2.35 ^b	97.65°	43.79
Viscozyme® L ^g	0.32°	36.78ª	42.04 ^a	20.86 ^g	76.64
Pectinex® Ultra Mash ^h	1.00 ^a	0.72 ^e	0.99^{d}	97.30 ^e	64.68
Pectinex® Ultra Clear ^h	0.49 ^b	0.00^{f}	0.00 ^e	99.51ª	24.38
Viscomyl® Flow ^g	0.00^{d}	1.35°	1.13°	97.53 ^d	21.81
Depol™ 670L ^g	0.00 ^d	0.92 ^d	0.00 ^e	99.08 ^b	29.06

Table 7. Efficiency of bi and multi-enzymatic systems used for the enzymatic generation of oligosaccharides from the NAC cell wall polysaccharides after 7hr reaction time

Values reported as mean (%, mol) DP – Degree of polymerization

^{a-f} Within the same column, means with different letters are significantly different at P < 0.05^g Enzyme concentration of 50U/g (galactanase units) ^hEnzyme concentration of 100U/g (galactanase units)

Table 4.8.	Efficiency	of bi and	multi-enzyn	natic system	ns used fo	or the e	enzymatic	generation	of oligos:	accharides
from the Co	CN cell wa	ll polysa	ccharides afte	er 7hr reac	tion time					

	DP>10	DP 6-10	DP 3-5	$DP \lesssim 2$	Hydrolysis extent (%)
Arabinase/Galactanase ^g	1.20 ^d	47.8ª	0.00^{f}	50.9 ^e	2.15
$Viscozyme$ \mathbb{R} L^{g}	1.31 ^c	5.93°	16.1 ^b	76.6 ^d	20.19
Pectinex® Ultra Mash ^g	1.97 ^b	0.17^{f}	0.24 ^e	97.6 ^b	21.31
Pectinex [®] Ultra Clear [®]	2.02 ^a	4.23 ^d	11.9°	81.8 ^c	1.36
Viscomyl® Flow ^g	0.21°	1.00 ^e	0.49 ^d	98.3ª	20.31
Depol™ 670L ^g	0.23 ^e	16.4 ^b	38.5ª	44.7 ^f	20.72

Values reported as mean (%, mol)

DP – Degree of polymerization

^{a-f} Within the same column, means with different letters are significantly different at P < 0.05

^g Enzyme concentration of 50U/g (galactanase units)

As seen in Table 4.8, CCN-based oligosaccharides hydrolyzed by Depol® 670L generated the most diverse oligosaccharides, with DP ≤ 2 (44.77%, mol), followed by DP 3-5 (38.56%, mol), DP 6-10 (16.44%, mol), and DP greater than 10 (0.23%, mol). Oligosaccharides of DP 6-10 were recovered more upon the use of the bi-enzymatic system (50 U/g, 47.81%, mol). In addition, compared to those from NAC, CCN-based oligosaccharides of DP >10 were more prevalent among the bi-enzymatic system and all multi-enzymatic products, ranging from 0.21 to- 2.02% mol. Overall, CCN-based polysaccharides were hydrolyzed less in comparison to the NAC variety, which may be attributed to enzyme low affinity and/or inaccessibility to hemicellulosic polysaccharides and to the branching. Pectinex[®] Ultra Mash, Viscozyme[®] L and Depol[™] 670L multi-enzymatic products all share specific enzyme activity towards pectinases, namely endogalacturonase and polygalacturonase. This may explain the higher hydrolysis extents favored towards NAC based polysaccharides using Pectinex[®] Ultra Mash, Viscozyme[®] L and Depol[™] 670L. Khodaei and Karboune (2018) suggested other possible explanations of this occurrence may include enzyme inhibition by substrate excess, occurrence of substrate diffusional limitations, as well as the presence of substrate/substrate interactions hindering their access to the enzyme's active site. This is an agreement with Emaga et al. (2012) as well as Efrinalia et al. (2022), where the strong interactions between neutral sugars and acid fractions had a significant effect on the enzymatic release of total sugar residues from flaxseed mucilage. In summary, Viscozyme® L was favored for the generation of NAC based oligosaccharides, while Depol 670®L was selected for the production of CCN-based oligosaccharides. In addition, arabinase/galactanase bienzymatic system was selected for both varieties in order to compare its efficiency to a multi-enzymatic system along the hydrolysis time course. The time courses for the hydrolysis extent by the three biocatalysts (arabinase/galactanase, Viscozyme[®] L and Depol[™] 670L) are reported in Figure 4.1A. With NAC based polysaccharides, the hydrolysis extent by the bi-enzymatic system and Viscozyme® L remained more or less constant over the first 7 hr reaction and increased thereafter at 24 hr to reach 66.0 and 100%, respectively. The results also show that the hydrolysis of CCN based polysaccharides by bi-enzymatic system did not exceed 8.9% over the 24 hr time course, while the hydrolysis extent by DepolTM 670L increased throughout the time course to reach 42.4%.


Figure 4.1. A) Hydrolysis extent obtained with selected multi-enzymatic products (50U/g); B) the corresponding MW distribution of enzymatically generated oligosaccharides; A- Arabinase: Galactanase NAC, B-Arabinase:Galactanase CCN, C – Viscozyme L® NAC, D – DepolTM 670L CCN, at various reaction times (1,3,7,24hr).

In Figure 4.1B, the MW distribution of oligosaccharides shows a shift from high to low DP over time, which is observed with the bi-enzymatic system and DepolTM 670L for the generation CCN-based oligosaccharides as well as with Viscozyme® L for the production of NAC-based oligosaccharides. The results also show that Depol[™] 670L with CCN oligosaccharides and the bienzymatic system with NAC oligosaccharides generated a high proportion of DP ≤ 2 within 1hr of reaction, likely indicating that the substrates were more accessible towards those biocatalysts. CCN oligosaccharides hydrolyzed by Depol[™] 670L after 3hr allowed for the generation of a diverse complex of oligosaccharides including DP ≤ 2 (44.5%), DP 3 to 5 (17.33%), DP 6 to 10 (28.87%), and DP > 10 (9.30%). Similarly, at 7hr, NAC oligosaccharides hydrolyzed by Viscozyme® L generated a profile comprised DP ≤ 2 (44.77%), DP 3 to 5 (38.56%), DP 6 to 10 (16.44%), and DP > 10 (0.23\%). All in all, the results indicate the generation of a versatile and complex oligosaccharide profile with the use of multi-enzymatic products, Depol[™] 670L and Viscozyme® L, compared to the bi-enzymatic system, arabinase: galactanase. Future investigation towards optimizing the hydrolysis reaction parameters, in terms of substrate/enzyme ratio, pH, reaction time would be needed to determine the favorable conditions to achieve maximal hydrolysis extents along with a diverse oligosaccharide profile. In addition, structural linkage analysis of the enzymatically generated CBS oligosaccharides would help provide a more complete visualization towards the enzyme cleavage specificity for each of the multi-enzymatic products.

4.4.3 Feruloylation of oligo and polysaccharides from cocoa bean shells

4.4.3.1 Feruloyl esterase immobilization

Immobilization of feruloyl esterases, from rumen microorganism and *H. insolens,* was investigated using an epoxy-based support, modified IDA-Cu-Sepabeads® EC-EP R and assessed for the hydrolytic efficiency compared to their free forms. The selection of the Sepabeads ® EC-EP R support was based on previous studies, which reported high immobilization yields with feruloyl esterase (Spadoni Andreani et al., 2021; Tamayo-Cabezas & Karboune, 2019). In addition, the use of this support applies the chelation immobilization approach, offering both an ease in the support regeneration as well as restricting enzyme leakage (Afaq & Iqbal, 2001). Table 9 shows the immobilization yields and the retained activities of feruloyl esterases from rumen microorganism and *H. insolens*.

	Hydrolytic activity (U/mg enzyme)	Immobilization Yield (%)	Retention of Activity (%)	Esterification activity (U/mg protein)
Free enzyme from rumen microorganism	8.66±0.72ª	-	-	0.17±0.03 ^b
Immobilized enzyme from rumen microorganism	$0.23{\pm}0.03^{b}$	99.30+10.31ª	2.66±0.27 ^a	n.d.*
Free enzyme from <i>H</i> . <i>insolens</i>	2.07±0.04°	-	-	1.36±0.22ª
Immobilized enzyme from <i>H. insolens</i>	0.015±0.0001°	56.9±0.55 ^b	59.13±0.57 ^b	0.0029±0.001 ^b

Table 9. Comparison of the hydrolytic and esterification activities of feruloyl esterases in their free and immobilized (modified IDA-Cu-Sepabeads EC-ER®) forms.

Values reported as average \pm standard deviation

*n.d.- not detected

 $^{\rm a}$ Within the same column, means with different letters are significantly different at $P \le 0.05$

Table 10. Feruloylation extent (%) of cocoa bean shell oligo- and polysaccharides (FA-OCCN, FA-ONAC, FA-PCCN, FA-PNAC enzymatically generated oligosaccharides using free feruloyl esterases (2U/l) from rumen microorganism and H. insolens at 144hr.

6				
	FA-PNAC (%)	FA-ONAC (%)	FA-PCCN (%)	FA-OCCN (%)
Free enzyme from rumen microorganism	56.9±10.5 ^{ab}	38.2±0.3 ^{bc}	60.1±6.3ª	38.7±2.1 ^{bc}
Free enzyme from <i>H. insolens</i>	47.3* ^{abc}	$50.5{\pm}8.4^{ab}$	49.5 ± 8.6^{abc}	30.8±7.9°
** 1 . 1				

Values reported as average \pm standard deviation

*Standard deviation zero or not detected

^a Within the same row, means with different letters are significantly different at P < 0.05

Feruloyl esterase from rumen microorganism attained a higher immobilization yield (99.3%) than feruloyl esterase from *H. insolens* (56.9%). Based on previous findings, the immobilization yields are higher than those previously reported, ranging between 66.2-72.5% (Spadoni Andreani et al., 2021; Tamayo-Cabezas & Karboune, 2019).

Although higher immobilization yield was attained with feruloyl esterase rumen microorganism, the retention of activity was poor (2.66%) in comparison to feruloyl esterase from H. insolens (59.13%). According to the manufacturer, the Sepabeads® EC-EP R support has a pore size of 10-20nm. The large pore size may have impacted the efficiency of the enzyme loading onto the support (Kang et al., 2007; Miao et al., 2022). In addition, He et al. (2015) reported that the immobilization time affected the immobilisation efficiency and resulted in both lower enzyme and specific activities of feruloyl esterases. This may be attributed to the saturated adsorption on the supports, thereby distributing the conformation of the biocatalyst (He et al., 2015; Juang et al., 2014). Furthermore, the hydrolytic and esterification activities of free and immobilized feruloyl esterases from rumen microorganism and H. insolens were compared. As seen in Table 9, free feruloyl esterases expressed significant (P < 0.05) hydrolytic activities (2.07-8.66 U/mg enzyme) compared to their immobilized forms (0.015-0.23 U/mg enzyme). The findings of the esterification of feruloyl esterases with raffinose show higher activities attained with free enzyme (0.17-1.36 U/mg protein) than immobilized enzyme (0-0.00029 U/mg protein). One study reported similar results where free feruloyl esterase attained higher raffinose esterification activity than its immobilized form by 54.2% (Spadoni Andreani et al. 2021). Based on our findings, the use of free feruloyl esterase, from rumen microogranism and H, insolens, was selected for further analysis.

4.4.3.2 Feruloylation cocoa bean shell oligo- and polysaccharide extracts

The feruloylation of oligo- and polysaccharides from CBS using free feruloyl esterases from rumen microorganism and *H. insolens* was investigated using a surfactant-less microemulsion comprised of n-hexane:butanone:water (49.4:44.6:6 % v/v/v). CBS oligosaccharides were obtained from multi-enzymatic hydrolysis with Viscozyme® L (NAC variety) and Depol 670^{TM} L (CCN variety), while polysaccharides (both varieties, NAC and CCN) were prepared from alkaline extraction.

As shown in Table 10, FA-PCCN and FA-PNAC using free feruloyl esterase from rumen microorganism attained significantly (P <0.05) higher feruloylation extents than with

oligosaccharides. Having a closer look between the two polysaccharide varieties, FA-PCCN (60.1%) were feruloylated at a slightly, yet significantly, higher extent than FA-PNAC (56.9%). On the other hand, free feruloyl esterase from *H. insolens* attained a significantly (P < 0.05) higher feruloylation extent (50.5%) with FA-ONAC, compared to the extent achieved with FA-OCCN (30.8%). Indeed, feruloyl esterase from *H. insolens* has previously shown large substrate specificity towards mono, di, and oligosaccharides (Couto et al., 2011). The latter authors suggested a correlation between the feruloylation extent and the degree of polymerization, where the feruloyl esterase-catalyzed reaction is dependant on the structural characteristics, linkage, and length of glycosides. According to the manufacturer, feruloyl esterase from rumen microorganism has a substrate specificity towards arabinose. In addition, studies have reported natural occurring pectic feruloylation confined to O2 or O3-positions of arabinan and the O6-position of galactans (Holck et al., 2011; Ralet et al., 1994; Sato et al., 2013; Spadoni Andreani et al., 2021). This may explain the higher feruloylation extents attained with FA-PCCN and FA-PNAC, which were found to be primarily composed of pectic regions (56.4-88.8%), with arabinose (7.8 to 16.4%) and galactose (8.1-22.4%) as one of the predominant monosaccharides.

The feruloylation efficiency was found to be dependent on the structural characteristics of the carbohydrate substrate (Couto et al., 2011; Tamayo-Cabezas & Karboune, 2019). To the author's knowledge, this is the first study on the feruloylation of CBS oligo- and polysaccharides by feruloyl esterases. The observed feruloylation extents are above the range reported for analogous reaction systems on non-digestible oligosaccharides from sugar-beet pulp and wheat bran (27.4%), pectic polysaccharides from cranberry pomace (4.2-18%), arabinobiose (10-18%), xylobiose (12.8%), xylooligosaccharides (10.6%) (Tamayo-Cabezas & Karboune, 2019; Spadoni Andreani et al., 2021; Vafiadi et al., 2006).

4.4.4 Functional Properties

4.4.4 Prebiotic Activity

The prebiotic activity scores of enzymatically generated oligosaccharides by selected multienzymatic products, Viscozyme[®] L for NAC variety and DepolTM 670L for CCN variety, rhamnogalacturonan, and inulin, a known prebiotic polysaccharide (Shoaib et al., 2016), were compared by using them as carbon sources for the anaerobic growth of two probiotic bacteria strains, *L. rhamnosus GG* and *B. longum*. Studies reported effective growth of *L. rhamnosus GG* with oligosaccharides derived from pearl millet fibre, *F. kuhistanic* leaves, wheat fibre, and mulberries (Farooq et al., 2017; Islamova et al., 2017; Li et al., 2022; Soukoulis et al., 2014) and *B. longum* with oligosaccharides derived from wheat flour, apple, and sugar beet pectin (Holck et al., 2011; Van Laere et al., 2000). In terms of specific oligosaccharide affinity, *L. rhamnosus GG* has been most analyzed for its affinity towards galacto-oligosaccharides. A study conducted by Li et al. (2022) reported that the proliferation of *L. rhamnosus GG* was significant due to up-regulated genes associated with galactose metabolism as well as those involved with glycolysis. In addition, *L. rhamnosus GG* has been previously known to express key enzymes involved in the fucose pathway, further promoting it affinity towards fucosylated oligosaccharides (Becerra et al., 2015). On the other hand, *B. longum* was found to exhibit excellent affinity towards fructooligosaccharides, arabino-furanosyl containing oligosaccharides, as well as α -galactosyloligosaccharides, such as raffinose and stachyose (Garro et al., 1999; Minami et al., 1983; van den Broek et al., 2008; Van Laere et al., 2000). Furthermore, *B. longum* has been found to possess over 40 glycosyl hydrolases which may indicate the strain's high affinity towards a wide range of oligosaccharides with a degree of polymerization less than 8 (Schell et al., 2002).

CBS oligosaccharides (NAC and CCN) showed a prebiotic activity score similar to that of inulin and rhamnogalacturonan for *L. rhamnosus GG* (Figure 4.2 A). At 24hr of fermentation, the prebiotic activity scores of oligosaccharides were slightly higher ($P \le 0.05$) than that of inulin and rhamnogalacturonan. For *B. longum* (Figure 4.2 B), CBS oligosaccharides



Figure 4.2. A) Prebiotic activity scores of *Lactobacillus rhamnosus GG* (Harmonium R0343 AR) on oligosaccharide extracts prepared from multi-enzymatic hydrolysis, inulin and rhamnogalacturonan, measured at 6,24, 48, 72 hr of incubation. B) Prebiotic activity scores of *Bifidobacterium longum* (ATCC ® 15707) on oligosaccharide extracts prepared from multi-enzymatic hydrolysis, inulin and rhamnogalacturonan, measured at 6, 24, 48, 72 hr of incubation. Values represented as an average \pm standard deviation. Oligo NAC: oligosaccharide NAC variety, Oligo CCN: oligosaccharide CCN variety, RhamnoG: Rhamnogalacturonan, respectively. For each incubation time, bars with different letters represent scores significantly different at P < 0.05

showed statistically significant (P ≤ 0.05) prebiotic activity scores as early as 6hr of fermentation than that of inulin and rhamnogalacturonan, while maintaining a relatively constant score throughout the 72 hr fermentation period. Therefore, it appears that CBS enzymatically generated oligosaccharides (Viscozyme® L and DepolTM 670L, 50U/g, 7hr) effectively stimulate the growth of both strains.

One study suggested that pectic polysaccharides used as a source of prebiotics underwent metabolization prior to consumption by B. longum and L. rhamnosus GG (Li et al., 2020). With that being said, one can assume that using enzymatically generated oligosaccharides offered faster and more direct consumption from both strains, while also providing an increased production of SCFA than those from cell wall polysaccharides. SCFA are produced upon the fermentation of dietary fiber by gut microbiota (Lin et al., 2016; Spadoni Andreani et al., 2021). Table 11 shows the concentration of four SCFA after 24 and 48 hr of fermentation of L. rhamnosus GG and B. longum with glucose, enzymatically generated oligosaccharides (NAC and CCN varieties) and positive controls, inulin and rhamnogalacturonan. In the culture of L. rhamnosus GG (Table 11), the main increase was seen in the lactic acid followed by acetic acid, similar to the trend observed with CBS polysaccharides in par 3.4.4. For oligosaccharides of CCN variety, the significance (P < 0.05) was observed with the increased production of propionic acid from 24 to 48 hr of fermentation. L. rhamnosus GG is known for its ability to produce lactic and propionic acid during fermentation, which can promote anti-inflammatory and anti-microbial properties in our gastrointestinal tract (Al-Lahham et al., 2010; Markowiak-Kopeć & Śliżewska, 2020). The production of butyric acid was prevalent in glucose, inulin, and rhamnogalacturonan, as well as CBS enzymatically generated oligosaccharides of NAC variety (24hr), indicating the favored production of butyrate in the L. rhamnosus fermentation pathway. This can be attributed to the metabolization of lactate as observed in Table 11, further explaining the decrease in the lactate concentration with increased incubation time (Cummings et al., 1987; Karboune et al., 2022).

Similarly, to the SCFA profile of *L. rhamnosus GG*, the culture of *B. longum* was characterized by the greatest increase in lactic acid, followed by acetic acid predominantly after 48 hr of fermentation. The acetic acid content of the medium relatively remained unchanged for the CBS oligosaccharides. The lactic acid concentrations produced by oligosaccharides were significantly comparable to their positive controls, glucose, inulin, and rhamnogalacturonan, at 24 hr of fermentation. Interestingly, at 48hr of fermentation, the lactic acid concentrations continuously increased with their positive controls, contrary to the behavior of CBS polysaccharides as described in par 3.4.4. Furthermore, CBS oligosaccharides, showed a notable significant (P < 0.05) production of butyric and propionic acids at 24hr of fermentation, either surpassing or similar to the positive controls. According to studies assessing the prebiotic effects of plant-derived oligosaccharides, smaller oligosaccharides, with a degree of polymerization less than 4, have shown to induce a "bifidogenic effect" as well as contribute to higher production of acetate and butyrate (Sanchez et al., 2009; Snelders et al., 2014; Van Craeyveld et al., 2008). However, the onset of this "bifidogenic effect" was not observed after 24 hr of *B. longum* fermentation, as butyric acid concentrations were decreased or undetected depending on the CBS variety.

Overall, the total SCFA profile of oligosaccharides continuously increased throughout 48 hr of fermentation with *B. longum*, contrary to those with *L. rhamnosus GG*. Studies suggest that *Bifidobacterium spp.*, specifically. *B. longum*, are producers of lactate, acetate and propionate in the human intestinal tract, as observed in Table 11 (Liu et al., 2020; Markowiak-Kopeć & Śliżewska, 2020). In addition, when carbohydrates are in excess, *Bifidobacterium* utilize the fermentation pathway, to produce two molecules of acetate with one molecule of lactate (LeBlanc et al., 2017; Macfarlane & Macfarlane, 2003; Markowiak-Kopeć & Śliżewska, 2020); on the contrary, our findings suggest a higher proportion of lactic than acetic acid.

4.4.4.2 Antioxidant Activity

The antioxidant activities of native and feruloylated polysaccharides were assessed using assays based on both single electron transfer (SET), which include DPPH and ORAC antioxidant assays, and hydrogen atom transfer (HAT), which include 2,2'-azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) (ABTS) antioxidant assay. The selection of different antioxidant mechanisms will help in providing a complete and precise evaluation of the antioxidant activity. The DPPH assay relies on the radical quenching capabilities of antioxidant by reducing DPPH in the presence of the antioxidant extract in a concentration dependant manner (Baumann et al., 1979; Rahman et al. 2015). Results from the DPPH assay were expressed as the IC₅₀ value, the concentration of the antioxidant needed to inhibit radical scavenging by 50% inhibition. As seen in Table 12, feruloylated polysaccharides led to a statistically significant (P < 0.05) decrease, by as much as 92-98%, in their IC₅₀ values as compared to native polysaccharides.

Table 11. Concentration (10^{-3} mol/l) of short chain fatty acids released in the cell-wall oligosaccharides samples (NAC, CCN) prepared from multi-enzymatic hydrolysis as compared to glucose, inulin and rhamnogalacturonan (RhamnoG) standards which were obtained from *Lactobacillus rhamnosus GG* (R0343 AR) and *Bifidobacterium longum* (ATCC® 15707) fermentation using different carbon sources. Values determined as difference with concentrations detected immediately after inoculation.

		hr	Glucose	Inulin	RhamnoG	NAC	CCN
	L. rhamnosus GG	24	312.9±4.0°	212.5±15.5°	508.8±10.3ª	458.6±37.3 ^{ab}	393.7±42.7 ^{bc}
Lactic acid		48	385.7±16.2ª	315.6±14.8 ª	293.5±25.9ª	337.3±67.7 ^a	383.9±29.2ª
Lattic aciu	B. longum	24	31.3±5.1ª	56.3*ª	178.0* ^a	275.3±15.5ª	174.2±21.3ª
		48	404.4 ± 96.6^{a}	281.3±1.8ª	327.9±2.1ª	369.8 ± 21.8^{a}	320.9±32.5ª
	L. rhamnosus GG	24	58.1 ± 8.4^{a}	46.4* ^a	112.2* ^a	38.9 ± 3.6^{a}	35.6±3.9ª
A patia paid		48	111.7 * ª	66.4*ª	38.9 ± 8.5^{a}	42.1* ^a	36.8 ± 3.5^{a}
Acetic acid	B. longum	24	44.4 ± 3.9^{a}	43.7±10.3ª	52.62* ^a	43.9±3.4ª	30.1±3.1ª
		48	230.1±11.3ª	55.3 ± 3.6^{b}	50.5 ± 1.5^{b}	37.6±1.7 ^b	33.8 ± 6.3^{b}
	L. rhamnosus GG	24	4.71* ^a	$0.9{\pm}0.04^{a}$	1.8±0.03ª	1.3±0.3ª	1.7* ^a
Propionic		48	0.66* ^{ab}	0.4* ^{ab}	$0.6{\pm}0.1^{ab}$	-	2.55*a
acid	B. longum	24	$1.9{\pm}0.4^{a}$	1.3±0.2ª	2.6 ± 0.4^{a}	1.75 * ª	$0.8{\pm}0.2^{a}$
		48	0.6* ^a	0.5±0.1ª	$0.5{\pm}0.03^{a}$	0.91* ^a	-
	L. rhamnosus GG	24	1.8* ^a	0.33*a	$0.4{\pm}0.04^{a}$	3.48* ^a	-
Butyric		48	-	-	-	-	-
acid	B. longum	24	0.3* ^b	0.33* ^b	-	5.33*ª	-
		48	-	-	-	3.05±0.3ª	-
	L. rhamnosus GG	24	377.6	220.2	623.2	502.4	430.9
Total		48	498.1	382.4	332.9	379.4	423.2
10121	B. longum	24	77.8	101.6	233.3	326.2	205.1
		48	635.2	337.1	378.8	411.3	355.5

Values reported as average \pm standard deviation

*Standard deviation zero or not detected

^a Within the same row, means with different letters are significantly different at P < 0.05

Table 12. Antioxidant activity as determined by the DPPH, ORAC, and ABTS assays, expressed as IC50 µg/	ml
for DPPH, µmol TE /mg sample for ORAC and ABTS of native and feruloylated polysaccharides (FA-PCC	CN
and FA-PNAC) as compared to ferulic and ascorbic acids	

	DPPH (IC50, ug/mL)	ORAC (µmol TE/mg sample)	ABTS (µmol TE/ mg sample)
NAC Native	6417.6±931.4ª	202.6±0.2°	312.6±97.5°
FA-NAC	39.3*°	252.3±0.7 ^b	1542.7±264°
CCN Native	4100.8±45 ^b	204.1±1.3°	211.6±49.6°
FA-CCN	170.6±17°	266.8±2.5ª	1517.7±292.7°
Ferulic Acid	37.4±0.8°	202.4±0.9°	15291.7±3279.1ª
Ascorbic Acid	57.6±16.1°	168.2 ± 1.7^{d}	9773.1±2239.6 ^b

Values reported as average \pm standard deviation

*Standard deviation zero or not detected

^a Within the same column, means with different letters are significantly different at P < 0.05

In addition, that FA-PNAC and FA-PCCN attained IC₅₀ values in proximity to their positive controls, ferulic and ascorbic acid. The decrease in the IC₅₀ indicates that a lower concentration of the FA-NAC and FA-PCCN (39.3-170.6 μ g/ml) is needed to inhibit and scavenge free radicals as compared to native polysaccharides (4100.8-6417.6 μ g/ml). Comparing the feruloylated polysaccharides from two CBS varieties, FA-PNAC attained lower IC₅₀ (4 fold) compared to FA-PCCN. As compared to the literature, FA-PNAC and FA-PCCN showed stronger antioxidant activity than feruloylated hemicelluloses and oligosaccharides from wheat bran (IC₅₀ 53.6 -1174.3 μ g/ml, Ruthes et al., 2017) and feruloylated xylo-oligosaccharides from pearl millet fibre (IC₅₀ 51.2 μ g/ml, Singh & Eligar 2021). Furthermore, CBS FA-PCCN and FA-PNAC showed a higher antioxidant capacity than those reported by Khodaei et al. (2021) as the best overall antioxidant capacities of essential oils which include Pimento berries (1.14x10⁵ μ g/ml), Ceylon cinnamon (1.76x10⁵ μ g/ml), and clove bud oil (1.90x10⁵ μ g/ml).

The ORAC assay relies on fluorescence-time based radical scavenging using peroxyl radicals to provide a better antioxidant reaction model as well as providing a continuous generation of radicals on a time-scale (Schaich et al., 2015). The findings from the ORAC assay exhibited a similar trend to that observed with the DPPH assay. The results are expressed in terms of TE, where a higher TE mediates a higher antioxidant activity. As seen in Table 12, the findings suggested that FA-PNAC and FA-PCCN (252.3-266.8 μ mol TE/mg) achieved a higher TE (P < 0.05) by as much as 10.9-13.3% as compared to CBS native polysaccharides (202.6-204.1 μ mol TE/mg). Furthermore, a significant (P < 0.05) difference between the feruloylated polysaccharides from the two varieties was observed, where FA-PCCN polysaccharides were superior. In addition, that FA-PNAC and FA-PCCN also showed a significant (P < 0.05) increase as compared to ferulic and ascorbic acids. Furthermore, the antioxidant activities of that FA-PNAC and FA-PCCN are higher than those reported for roasted cocoa beans (0.25 μ mol TE/mg), cloves (3.14 μ mol TE/mg), and cinnamon (2.67 μ mol TE/mg) (Jolić et al., 2011; Schaich et al., 2015).

The ABTS assays adopts the principles of hydrogen atom transfer antioxidant mechanism in which test the ability of antioxidant extracts to intercept initial oxidation and prevent production of ABTS⁺⁺ (Schaich et al., 2015). Our findings, in Table 12, suggest that that FA-PCCN and FA-PNAC (1517.7-1542.7µmol TE/mg) attained a higher TE by as much as 66.3-75.5% (P < 0.05) compared to native polysaccharides (211.6-312.6 µmol TE/mg). Between FA-PNAC and FA-

PCCN, no significant (P<0.05) differences were observed. These results are higher than those reported for raw CBS (0.0135-0.046 μ mol TE/mg, Botella-Martínez et al., 2021), native and feruloylated arabinoxylans from maize dried grains (39-67 nmol TE/mg, Marquez-Escalante & Carvajal-Millan 2019), feruloylated non-starch polysaccharides isolated from barley and wheat (58.0-105.0 μ M TE/g, Malunga & Beta 2015).

As an overall, CBS polysaccharides exhibited stronger antioxidant capacities upon feruloylation. Studies suggest that specific linkages between polysaccharides and ferulic acid moieties may promote a higher antioxidant activity (Cos et al., 2002; Kylli et al., 2008; Ou & Sun, 2014). One study conducted by Kylli et al. (2008) found that the 6-O-feruloyl-glucoside, 2-O- and 3-O-feruloyl-glycoside linkages exhibited high antioxidant activities higher or comparable to free ferulic acid. As mentioned earlier, CBS native polysaccharides contain a high phenolic content, which may provide an early indication of high antioxidant activity. There are conflicting agreements concerning the correlation between phenolic content and antioxidant activities. Using electron spin resonance spectroscopy to assess the antioxidant capacity of cocoa fibre found a positive correlation (R=0.87) between total phenolic compounds and antioxidant activity (Lecumberri et al., 2007). On the other hand, other studies assessed on cocoa beans and flaxseed found no correlation among the two attributes, as the antioxidant activities due to the high content of reducing agents, such as ascorbic acid (Agus et al., 2018; Bruna et al., 2012; Georgé et al., 2005; Ikram et al., 2009; Othman et al., 2007).

4.5 Conclusion

Two CBS polysaccharide extracts were assessed in terms of the cell wall composition and MW distribution. Both varieties contained high proportion of pectic polysaccharides, identified to be primarily rhamnogalacturonan. CCN based polysaccharides also contained a shared proportion of hemi-cellulosic polysaccharides, including galactomannan and xylan. Additionally, the two CBS varieties showed high contents of phenolic compounds compared to other by-products, such as cranberry pomace. Of the six biocatalysts, Viscozyme® L led to the highest hydrolysis extent of NAC based polysaccharides and resulted in oligosaccharides with DP 3-5 and 6-10. As for CCN based polysaccharides, Depol® 670L had the highest hydrolysis extent and generated a mixture rich in oligosaccharides with DP up to 10, likely from the degradation of pectic and hemi-cellulosic polysaccharides.

The raffinose esterification efficiency using free and immobilized feruloyl esterases, from rumen microorganism and *H. insolens,* was also explored. Although feruloyl esterases attained high immobilization yields (56.9-99.3%), their hydrolytic and retention of activities were minimal, further impacting their esterification activity. Based on the results, the selection of free feruloyl esterases was deemed favorable for feruloylation reaction with CBS oligo- and polysaccharides. Overall, feruloyl esterase from rumen microorganism provided best feruloylation efficiencies with CBS polysaccharides (56.9-60.1%), while the enzyme from *H. insolens* provided best feruloylation efficiencies with the oligosaccharides (30.8-50.5%).

Enzymatically generated oligosaccharides as well as feruloylated polysaccharides were assessed for their prebiotic and antioxidant activities, respectively. When tested for the promotion of the growth of selected probiotic strains, select CBS oligosaccharides (Viscozyme® L for NAC and Depol® 670L for CCN) performed similarly or more than inulin and rhamnogalacturonan based on the prebiotic activity scores. The short chain fatty acid profiles were characterized by high amounts of lactic acid and acetic. In addition, a significant production of butyric and propionic acids were observed with CBS oligosaccharides. Based on the DPPH, ORAC, and ABTS assays, the feruloylated CBS polysaccharides prompted significantly notable increases in the scavenging and antioxidant activity compared to their native forms as well as ferulic and ascorbic acid.

References

- Afaq, S., & Iqbal, J. (2001). Immobilization and stabilization of papain on chelating sepharose: A metal chelate regenerable carrier. *Electronic Journal of Biotechnology*, 4(3). https://www.scielo.cl/scielo.php?pid=S0717-34582001000300006&script=sci arttext&tlng=en
- Agus, B. A. P., Mohamad, N. N., & Hussain, N. (2018). Composition of unfermented, unroasted, roasted cocoa beans and cocoa shells from Peninsular Malaysia. *Journal of Food Measurement and Characterization*, 12(4), 2581–2589. https://doi.org/10.1007/s11694-018-9875-4
- Albishi, T., John, J. A., Al-Khalifa, A. S., & Shahidi, F. (2013). Phenolic content and antioxidant activities of selected potato varieties and their processing by-products. *Journal of Functional Foods*, 5(2), 590–600. https://doi.org/10.1016/j.jff.2012.11.019
- Al-Lahham, S. H., Peppelenbosch, M. P., Roelofsen, H., Vonk, R. J., and Venema, K. (2010).
 Biological effects of propionic acid in humans; metabolism, potential applications and underlying mechanisms. *Biochimica et Biophysica Acta (BBA) Molecular and Cell Biology of Lipids*, 1801(11), 1175–1183. https://doi.org/10.1016/j.bbalip.2010.07.007
- Ballesteros, L. F., Teixeira, J. A., & Mussatto, S. I. (2014). Selection of the Solvent and Extraction Conditions for Maximum Recovery of Antioxidant Phenolic Compounds from Coffee Silverskin. *Food and Bioprocess Technology*, 7(5), 1322–1332. https://doi.org/10.1007/s11947-013-1115-7
- Baumann, J., Wurn, G., & Bruchlausen, F. V. (1979). Prostaglandin synthetase inhibiting O2radical scavenging properties of some flavonoids and related phenolic compounds. *Naunyn-Schmiedebergs Arch Pharmacol*, 308, 27–32. https://cir.nii.ac.jp/crid/1572824499867306112
- Becerra, J. E., Yebra, M. J., & Monedero, V. (2015). An L-Fucose Operon in the Probiotic Lactobacillus rhamnosus GG Is Involved in Adaptation to Gastrointestinal Conditions. *Applied and Environmental Microbiology*, 81(11), 3880–3888. https://doi.org/10.1128/AEM.00260-15
- Blumenkrantz, N., & Asboe-Hansen, G. (1973). New method for quantitative determination of uronic acids. *Analytical Biochemistry*, 54(2), 484–489. https://doi.org/10.1016/0003-2697(73)90377-1
- Botella-Martínez, C., Lucas-Gonzalez, R., Ballester-Costa, C., Pérez-Álvarez, J. A., Fernández-López, J., Delgado-Ospina, J., Chaves-López, C., & Viuda-Martos, M. (2021). Ghanaian Cocoa (Theobroma cacao L.) Bean Shells Coproducts: Effect of Particle Size on Chemical Composition, Bioactive Compound Content and Antioxidant Activity. *Agronomy*, 11(2), 401. https://doi.org/10.3390/agronomy11020401
- Bruna, C. ., Eichholz, I., Rohn, S., Kroh, L. W., & Huyskens-Keil, S. (2012). Bioactive compounds and antioxidant activity of cocoa hulls (Theobroma cacao L.) from different origins. *Journal of Applied Botany and Food Quality*.
- Cao, G., Alessio, H. M., & Cutler, R. G. (1993). Oxygen-radical absorbance capacity assay for antioxidants. *Free Radical Biology and Medicine*, 14(3), 303–311. https://doi.org/10.1016/0891-5849(93)90027-R
- Cos, P., Rajan, P., Vedernikova, I., Calomme, M., Pieters, L., Vlietinck, A. J., Augustyns, K., Haemers, A., & Berghe, D. V. (2002). In Vitro Antioxidant Profile of Phenolic Acid Derivatives. *Free Radical Research*, *36*(6), 711–716. https://doi.org/10.1080/10715760290029182

- Couto, J., Karboune, S., & Mathew, R. (2010). Regioselective synthesis of feruloylated glycosides using the feruloyl esterases expressed in selected commercial multi-enzymatic preparations as biocatalysts. *Biocatalysis and Biotransformation*, *28*(4), 235–244. https://doi.org/10.3109/10242422.2010.493209
- Couto, J., St-Louis, R., & Karboune, S. (2011). Optimization of feruloyl esterase-catalyzed synthesis of feruloylated oligosaccharides by response surface methodology. *Journal of Molecular Catalysis B: Enzymatic*, 73(1–4), 53–62. https://doi.org/10.1016/j.molcatb.2011.07.016
- Cummings, J. H., Pomare, E. W., Branch, W. J., Naylor, C. P., & Macfarlane, G. T. (1987). Short chain fatty acids in human large intestine, portal, hepatic and venous blood. *Gut*, 28(10), 1221–1227. https://doi.org/10.1136/gut.28.10.1221
- Du, B., Song, Y., Hu, X., Liao, X., Ni, Y., & Li, Q. (2011). Oligosaccharides prepared by acid hydrolysis of polysaccharides from pumpkin (Cucurbita moschata) pulp and their prebiotic activities: Oligosaccharides prepared and their activities. *International Journal* of Food Science & Technology, 46(5), 982–987. https://doi.org/10.1111/j.1365-2621.2011.02580.x
- DuBois, M., Gilles, K. A., Hamilton, J. K., Rebers, P. A., & Smith, F. (1956). Colorimetric Method for Determination of Sugars and Related Substances. *Analytical Chemistry*, 28(3), 350–356. https://doi.org/10.1021/ac60111a017
- Efrinalia, W., Novia, N., & Melwita, E. (2022). Kinetic Model for Enzymatic Hydrolysis of Cellulose from Pre-Treated Rice Husks. *Fermentation*, 8(9), 417. https://doi.org/10.3390/fermentation8090417
- Emaga, T. H., Rabetafika, N., Blecker, C. S., & Paquot, M. (2012). Kinetics of the hydrolysis of polysaccharide galacturonic acid and neutral sugars chains from flaxseed mucilage. *Biotechnologie, Agronomie, Société et Environnement/Biotechnology, Agronomy, Society and Environment, 16*(2), 139–147. https://popups.uliege.be/1780-4507/index.php?id=16951&file=1&pid=8563
- Farooq, U., Liu, X., & Zhang, H. (2017). Enhancement of Short Chain Fatty Acid Production by Co-cultures of Probiotics Fermentation with Pearl Millet (Pennisetum glaucum) Fibre Fractions. *Journal of Pure and Applied Microbiology*, 11(4), 2031–2038. https://doi.org/10.22207/JPAM.11.4.47
- Garro, M. S., de Valdez, G. F., Oliver, G., & de Giori, G. S. (1999). Hydrolysis of soya milk oligosaccharides by Bifidobacterium longum CRL 849. Zeitschrift F r Lebensmitteluntersuchung Und -Forschung A, 208(1), 57–59. https://doi.org/10.1007/s002170050375
- Georgé, S., Brat, P., Alter, P., & Amiot, M. J. (2005). Rapid Determination of Polyphenols and Vitamin C in Plant-Derived Products. *Journal of Agricultural and Food Chemistry*, 53(5), 1370–1373. https://doi.org/10.1021/jf048396b
- He, F., Zhang, S., & Liu, X. (2015). Immobilization of feruloyl esterases on magnetic nanoparticles and its potential in production of ferulic acid. *Journal of Bioscience and Bioengineering*, 120(3), 330–334. https://doi.org/10.1016/j.jbiosc.2015.01.006
- Holck, J., Hjernø, K., Lorentzen, A., Vigsnæs, L. K., Hemmingsen, L., Licht, T. R., Mikkelsen, J. D., & Meyer, A. S. (2011). Tailored enzymatic production of oligosaccharides from sugar beet pectin and evidence of differential effects of a single DP chain length difference on human faecal microbiota composition after in vitro fermentation. *Process Biochemistry*, 46(5), 1039–1049. https://doi.org/10.1016/j.procbio.2011.01.013

- Holck, J., Lorentzen, A., Vigsnæs, L. K., Licht, T. R., Mikkelsen, J. D., & Meyer, A. S. (2011). Feruloylated and Nonferuloylated Arabino-oligosaccharides from Sugar Beet Pectin Selectively Stimulate the Growth of Bifidobacterium spp. In Human Fecal in Vitro Fermentations. *Journal of Agricultural and Food Chemistry*, 59(12), 6511–6519. https://doi.org/10.1021/jf200996h
- Ikram, E. H. K., Eng, K. H., Jalil, A. M. M., Ismail, A., Idris, S., Azlan, A., Nazri, H. S. M., Diton, N. A. M., & Mokhtar, R. A. M. (2009). Antioxidant capacity and total phenolic content of Malaysian underutilized fruits. *Journal of Food Composition and Analysis*, 22(5), 388–393. https://doi.org/10.1016/j.jfca.2009.04.001
- Islamova, Zh. I., Ogai, D. K., Abramenko, O. I., Lim, A. L., Abduazimov, B. B., Malikova, M. Kh., Rakhmanberdyeva, R. K., Khushbaktova, Z. A., & Syrov, V. N. (2017). Comparative Assessment of the Prebiotic Activity of Some Pectin Polysaccharides. *Pharmaceutical Chemistry Journal*, 51(4), 288–291. https://doi.org/10.1007/s11094-017-1600-9
- Jolić, S. M., Redovniković, I. R., Marković, K., Šipušić, Đ. I., & Delonga, K. (2011). Changes of phenolic compounds and antioxidant capacity in cocoa beans processing: Changes of phenolic in cocoa beans processing. *International Journal of Food Science & Technology*, 46(9), 1793–1800. https://doi.org/10.1111/j.1365-2621.2011.02670.x
- Juang, T. Y., Kan, S. J., Chen, Y. Y., Tsai, Y. L., Lin, M. G., & Lin, L. L. (2014). Surface-Functionalized Hyperbranched Poly(Amido Acid) Magnetic Nanocarriers for Covalent Immobilization of a Bacterial γ-Glutamyltranspeptidase. *Molecules*, 19(4), 4997–5012. https://doi.org/10.3390/molecules19044997
- Kang, Y., He, J., Guo, X., Guo, & Song, Z. (2007). Influence of Pore Diameters on the Immobilization of Lipase in SBA-15. *Industrial & Engineering Chemistry Research*, 46(13), 4474–4479. https://doi.org/10.1021/ie061440n
- Karboune, S., Seo, S., Li, M., Waglay, A., & Lagacé, L. (2022). Biotransformation of sucrose rich Maple syrups into fructooligosaccharides, oligolevans and levans using levansucrase biocatalyst: Bioprocess optimization and prebiotic activity assessment. *Food Chemistry*, 382, 132355. https://doi.org/10.1016/j.foodchem.2022.132355
- Khodaei, N., & Karboune, S. (2013). Extraction and structural characterisation of rhamnogalacturonan I-type pectic polysaccharides from potato cell wall. *Food Chemistry*, 139(1–4), 617–623. https://doi.org/10.1016/j.foodchem.2013.01.110
- Khodaei, N., & Karboune, S. (2016). Enzymatic generation of galactose-rich oligosaccharides/oligomers from potato rhamnogalacturonan I pectic polysaccharides. *Food Chemistry*, 197, 406–414. https://doi.org/10.1016/j.foodchem.2015.10.122
- Khodaei, N., & Karboune, S. (2018). Optimization of enzymatic production of prebiotic galacto/galacto(arabino)-oligosaccharides and oligomers from potato rhamnogalacturonan I. *Carbohydrate Polymers*, 181, 1153–1159. https://doi.org/10.1016/j.carbpol.2017.11.011
- Khodaei, N., Nguyen, M. M., Mdimagh, A., Bayen, S., & Karboune, S. (2021). Compositional diversity and antioxidant properties of essential oils: Predictive models. *LWT*, 138, 110684. https://doi.org/10.1016/j.lwt.2020.110684
- Kylli, P., Nousiainen, P., Biely, P., Sipilä, J., Tenkanen, M., & Heinonen, M. (2008). Antioxidant Potential of Hydroxycinnamic Acid Glycoside Esters. *Journal of Agricultural and Food Chemistry*, 56(12), 4797–4805. https://doi.org/10.1021/jf800317v

- Lama-Muñoz, A., Rodríguez-Gutiérrez, G., Rubio-Senent, F., & Fernández-Bolaños, J. (2012). Production, characterization and isolation of neutral and pectic oligosaccharides with low molecular weights from olive by-products thermally treated. *Food Hydrocolloids*, 28(1), 92–104. https://doi.org/10.1016/j.foodhyd.2011.11.008
- LeBlanc, G., Chain, F., Martín, R., Bermúdez-Humarán, L. G., Courau, S., & Langella, P. (2017). Beneficial effects on host energy metabolism of short-chain fatty acids and vitamins produced by commensal and probiotic bacteria. *Microbial Cell Factories*, 16(1), 79. https://doi.org/10.1186/s12934-017-0691-z
- Lecumberri, E., Mateos, R., Izquierdo-Pulido, M., Rupérez, P., Goya, L., & Bravo, L. (2007). Dietary fibre composition, antioxidant capacity and physico-chemical properties of a fibre-rich product from cocoa (Theobroma cacao L.). *Food Chemistry*, *104*(3), 948–954. https://doi.org/10.1016/j.foodchem.2006.12.054
- Li, E., Zhu, Q., Pang, D., Liu, F., Liao, S., & Zou, Y. (2022). Analysis of Lactobacillus rhamnosus GG in Mulberry Galacto-Oligosaccharide Medium by Comparative Transcriptomics and Metabolomics. *Frontiers in Nutrition*, *9*, 853271. https://doi.org/10.3389/fnut.2022.853271
- Li, Y., Wang, S., Sun, Y., Zheng, H., Tang, Y., Gao, X., Song, C., Liu, J., Long, Y., Liu, L., & Mei, Q. (2020). Apple polysaccharide could promote the growth of Bifidobacterium longum. *International Journal of Biological Macromolecules*, 152, 1186–1193. https://doi.org/10.1016/j.ijbiomac.2019.10.210
- Lin, Z., Fischer, J., & Wicker, L. (2016). Intermolecular binding of blueberry pectin-rich fractions and anthocyanin. *Food Chemistry*, 194, 986–993. https://doi.org/10.1016/j.foodchem.2015.08.113
- Liu, C., Kolida, S., Charalampopoulos, D., & Rastall, R. A. (2020). An evaluation of the prebiotic potential of microbial levans from Erwinia sp. 10119. *Journal of Functional Foods*, 64, 103668. https://doi.org/10.1016/j.jff.2019.103668
- Macfarlane, S., & Macfarlane, G. T. (2003). Regulation of short-chain fatty acid production. *Proceedings of the Nutrition Society*, 62(1), 67–72. https://doi.org/10.1079/PNS2002207
- Malunga, L. N., & Beta, T. (2015). Antioxidant Capacity of Water-Extractable Arabinoxylan from Commercial Barley, Wheat, and Wheat Fractions. *Cereal Chemistry Journal*, 92(1), 29–36. https://doi.org/10.1094/CCHEM-11-13-0247-R
- Markowiak-Kopeć, P., & Śliżewska, K. (2020). The Effect of Probiotics on the Production of Short-Chain Fatty Acids by Human Intestinal Microbiome. *Nutrients*, *12*(4), 1107. https://doi.org/10.3390/nu12041107
- Marquez-Escalante, J. A., & Carvajal-Millan, E. (2019). Feruloylated Arabinoxylans from Maize Distiller's Dried Grains with Solubles: Effect of Feruloyl Esterase on their Macromolecular Characteristics, Gelling, and Antioxidant Properties. *Sustainability*, 11(22), 6449. https://doi.org/10.3390/su11226449
- Martínez Sabajanes, A., Yáñez, R., Alonso, J. L., & Parajó, J. C. (2012). Pectic oligosaccharides production from orange peel waste by enzymatic hydrolysis: Pectic oligosaccharides production from orange peel waste. *International Journal of Food Science & Technology*, 47(4), 747–754. https://doi.org/10.1111/j.1365-2621.2011.02903.x
- Mastihubová, M., Mastihuba, V., Bilaničová, D., & Boreková, M. (2006). Commercial enzyme preparations catalyse feruloylation of glycosides. *Journal of Molecular Catalysis B: Enzymatic*, *38*(1), 54–57. https://doi.org/10.1016/j.molcatb.2005.11.003

- Miao, H., Li, M., Sun, X., Xia, J., Li, Y., Li, J., Wang, F., & Xu, J. (2022). Effects of Pore Size and Crosslinking Methods on the Immobilization of Myoglobin in SBA-15. *Frontiers in Bioengineering and Biotechnology*, 9, 827552. https://doi.org/10.3389/fbioe.2021.827552
- Minami, Y., Yazawa, K., Tamura, Z., Tanaka, T., & Yamamoto, T. (1983). Selectivity of utilization of galactosyl-oligosaccharides by bifidobacteria. *Chemical and Pharmaceutical Bulletin*, 31(5), 1688–1691. https://doi.org/10.1248/cpb.31.1688
- Organizers, Deutsche Pharmakologische Gesellschaft Abstracts of the 20th Spring Meeting. (1979). Deutsche Pharmakologische Gesellschaft Abstracts of the 20th Spring Meeting: March 20?30, 1979, Mainz. *Naunyn-Schmiedeberg's Archives of Pharmacology*, *307*(S1), R1–R77. https://doi.org/10.1007/BF00647307
- Othman, A., Ismail, A., Abdul Ghani, N., & Adenan, I. (2007). Antioxidant capacity and phenolic content of cocoa beans. *Food Chemistry*, *100*(4), 1523–1530. https://doi.org/10.1016/j.foodchem.2005.12.021
- Ou, J., and Sun, Z. (2014). Feruloylated oligosaccharides: Structure, metabolism and function. *Journal of Functional Foods*, 7, 90–100. https://doi.org/10.1016/j.jff.2013.09.028
- Rahman, M. M., Islam, M. B., Biswas, M., & Khurshid Alam, A. H. M. (2015). In vitro antioxidant and free radical scavenging activity of different parts of Tabebuia pallida growing in Bangladesh. *BMC Research Notes*, 8(1), 621. https://doi.org/10.1186/s13104-015-1618-6
- Ralet, M., Thibault, J., Faulds, C. B., & Williamson, G. (1994). Isolation and purification of feruloylated oligosaccharides from cell walls of sugar-beet pulp. *Carbohydrate Research*, 263(2), 227–241. https://doi.org/10.1016/0008-6215(94)00175-8
- Re, R., Pellegrini, N., Proteggente, A., Pannala, A., Yang, M., & Rice-Evans, C. (1999). Antioxidant activity applying an improved ABTS radical cation decolorization assay. *Free Radical Biology and Medicine*, 26(9–10), 1231–1237. https://doi.org/10.1016/S0891-5849(98)00315-3
- Redgwell, R. J., Trovato, V., Merinat, S., Curti, D., Hediger, S., & Manez, A. (2003). Dietary fibre in cocoa shell: Characterisation of component polysaccharides. *Food Chemistry*, *81*(1), 103–112. https://doi.org/10.1016/S0308-8146(02)00385-0
- Ruthes, A. C., Martínez-Abad, A., Tan, H., Bulone, V., & Vilaplana, F. (2017). Sequential fractionation of feruloylated hemicelluloses and oligosaccharides from wheat bran using subcritical water and xylanolytic enzymes. *Green Chemistry*, 19(8), 1919–1931. https://doi.org/10.1039/C6GC03473J
- Sanchez, J. I., Marzorati, M., Grootaert, C., Baran, M., Van Craeyveld, V., Courtin, C. M., Broekaert, W. F., Delcour, J. A., Verstraete, W., & Van de Wiele, T. (2009). Arabinoxylan-oligosaccharides (AXOS) affect the protein/carbohydrate fermentation balance and microbial population dynamics of the Simulator of Human Intestinal Microbial Ecosystem: AXOS effect on protein/carbohydrate fermentation balance. *Microbial Biotechnology*, 2(1), 101–113. https://doi.org/10.1111/j.1751-7915.2008.00064.x
- Sato, N., Takano, Y., Mizuno, M., Nozaki, K., Umemura, S., Matsuzawa, T., Amano, Y., & Makishima, S. (2013). Production of feruloylated arabino-oligosaccharides (FA-AOs) from beet fiber by hydrothermal treatment. *The Journal of Supercritical Fluids*, 79, 84– 91. https://doi.org/10.1016/j.supflu.2013.01.012

- Schaich, K. M., Tian, X., & Xie, J. (2015). Hurdles and pitfalls in measuring antioxidant efficacy: A critical evaluation of ABTS, DPPH, and ORAC assays. *Journal of Functional Foods*, 14, 111–125. https://doi.org/10.1016/j.jff.2015.01.043
- Schell, M. A., Karmirantzou, M., Snel, B., Vilanova, D., Berger, B., Pessi, G., Zwahlen, M. C., Desiere, F., Bork, P., Delley, M., Pridmore, R. D., & Arigoni, F. (2002). The genome sequence of Bifidobacterium longum reflects its adaptation to the human gastrointestinal tract. *Proceedings of the National Academy of Sciences*, 99(22), 14422–14427. https://doi.org/10.1073/pnas.212527599
- Shoaib, M., Shehzad, A., Omar, M., Rakha, A., Raza, H., Sharif, H. R., Shakeel, A., Ansari, A., & Niazi, S. (2016). Inulin: Properties, health benefits and food applications. *Carbohydrate Polymers*, 147, 444–454. https://doi.org/10.1016/j.carbpol.2016.04.020
- Singh, A., & Eligar, S. M. (2021). Bioactive feruloylated xylooligosaccharides derived from Pearl millet (Pennisetum glaucum) bran with antiglycation and antioxidant properties. *Journal of Food Measurement and Characterization*, 15(6), 5695–5706. https://doi.org/10.1007/s11694-021-01139-7
- Singleton, V. L., Orthofer, R., & Lamuela-Raventós, R. M. (1999). [14] Analysis of total phenols and other oxidation substrates and antioxidants by means of folin-ciocalteu reagent. In *Methods in Enzymology* (Vol. 299, pp. 152–178). Elsevier. https://doi.org/10.1016/S0076-6879(99)99017-1
- Snelders, J., Olaerts, H., Dornez, E., Van de Wiele, T., Aura, A. M., Vanhaecke, L., Delcour, J. A., & Courtin, C. M. (2014). Structural features and feruloylation modulate the fermentability and evolution of antioxidant properties of arabinoxylanoligosaccharides during in vitro fermentation by human gut derived microbiota. *Journal of Functional Foods*, 10, 1–12. https://doi.org/10.1016/j.jff.2014.05.011
- Soukoulis, C., Behboudi-Jobbehdar, S., Yonekura, L., Parmenter, C., & Fisk, I. D. (2014). Stability of Lactobacillus rhamnosus GG in prebiotic edible films. *Food Chemistry*, 159, 302–308. https://doi.org/10.1016/j.foodchem.2014.03.008
- Spadoni Andreani, E., & Karboune, S. (2020). Comparison of enzymatic and microwave-assisted alkaline extraction approaches for the generation of oligosaccharides from American Cranberry (Vaccinium macrocarpon) Pomace. *Journal of Food Science*, 85(8), 2443– 2451. https://doi.org/10.1111/1750-3841.15352
- Spadoni Andreani, E., Li, M., Ronholm, J., & Karboune, S. (2021). Feruloylation of polysaccharides from cranberry and characterization of their prebiotic properties. *Food Bioscience*, 42, 101071. https://doi.org/10.1016/j.fbio.2021.101071
- Srikanth, R., Siddartha, G., Sundhar Reddy, C. H., Harish, B. S., Janaki Ramaiah, M., & Uppuluri, K. B. (2015). Antioxidant and anti-inflammatory levan produced from Acetobacter xylinum NCIM2526 and its statistical optimization. *Carbohydrate Polymers*, 123, 8–16. https://doi.org/10.1016/j.carbpol.2014.12.079
- Tamayo-Cabezas, J., & Karboune, S. (2019). Immobilized feruloyl esterase from Humicola insolens catalyzes the synthesis of feruloylated oligosaccharides. *Process Biochemistry*, 79, 81–90. https://doi.org/10.1016/j.procbio.2018.12.013
- Thörn, C., Gustafsson, H., & Olsson, L. (2011). Immobilization of feruloyl esterases in mesoporous materials leads to improved transesterification yield. *Journal of Molecular Catalysis B: Enzymatic*, 72(1–2), 57–64. https://doi.org/10.1016/j.molcatb.2011.05.002

- Vafiadi, C., Topakas, E., & Christakopoulos, P. (2006). Regioselective esterase-catalyzed feruloylation of l-arabinobiose. *Carbohydrate Research*, 341(12), 1992–1997. https://doi.org/10.1016/j.carres.2006.05.022
- Van Craeyveld, V., Swennen, K., Dornez, E., Van de Wiele, T., Marzorati, M., Verstraete, W., Delaedt, Y., Onagbesan, O., Decuypere, E., Buyse, J., De Ketelaere, B., Broekaert, W. F., Delcour, J. A., & Courtin, C. M. (2008). Structurally Different Wheat-Derived Arabinoxylooligosaccharides Have Different Prebiotic and Fermentation Properties in Rats. *The Journal of Nutrition*, *138*(12), 2348–2355. https://doi.org/10.3945/jn.108.094367
- van den Broek, L. A., Hinz, S. W. A., Beldman, G., Vincken, J. P., & Voragen, A. G. J. (2008). Bifidobacterium carbohydrases-their role in breakdown and synthesis of (potential) prebiotics. *Molecular Nutrition & Food Research*, 52(1), 146–163. https://doi.org/10.1002/mnfr.200700121
- Van Laere, K. M. J., Hartemink, R., Bosveld, M., Schols, H. A., & Voragen, A. G. J. (2000). Fermentation of Plant Cell Wall Derived Polysaccharides and Their Corresponding Oligosaccharides by Intestinal Bacteria. *Journal of Agricultural and Food Chemistry*, 48(5), 1644–1652. https://doi.org/10.1021/jf990519i
- Waglay, A., & Karboune, S. (2017). A novel enzymatic approach based on the use of multienzymatic systems for the recovery of enriched protein extracts from potato pulp. *Food Chemistry*, 220, 313–323. https://doi.org/10.1016/j.foodchem.2016.09.147
- Yang, J., Martínez, I., Walter, J., Keshavarzian, A., & Rose, D. J. (2013). In vitro characterization of the impact of selected dietary fibers on fecal microbiota composition and short chain fatty acid production. *Anaerobe*, 23, 74–81. https://doi.org/10.1016/j.anaerobe.2013.06.012
- Zaidel, D. N. A., Arnous, A., Holck, J., & Meyer, A. S. (2011). Kinetics of Enzyme-Catalyzed Cross-Linking of Feruloylated Arabinan from Sugar Beet. *Journal of Agricultural and Food Chemistry*, 59(21), 11598–11607. https://doi.org/10.1021/jf203138u

Supplementary Data

Table I. Multi-enzymatic product profile and commercial specification according to indicated providers. PGase-polygalacturonase, EndoGlnase – endoglucanase, Arase- arabinase, Galase – galactanase, RGase – rhamnogalacturonase, XYase - xylanase

Product name	Provider	PGase	Endo-	Arase	Galase	RGase	XYase Amylase	Commercial specification
			Glnase					
Pectinase A. Aculeatus	Sigma Aldrich	1689.2						Pectinase
endo- 1,5, Arabinanase	Megazyme			52.9				Arabinanase
Viscozyme L	Novozyme			175.9	1493.1	91.9	115.1	beta-glucanases (107.1 U/g), pectinases, hemicellulases and xylanases
Pectinex Mesh	Novozyme	5474.6	170.5	181.6	0.7	285.8	404.2	Pectinases (10031 U/g) and hemicellulases and viscosity reduction
Viscomyl Flow	Dupont	14.3	8964.4	2.8	1.3	0.8	2028.1	Cellulase, 6200-7580 IU/g

* One unit of enzyme was defined as μ mol of reduced sugar was generated per min, measured using sodium acetate 0.1M, pH 5.5 as buffer, at 40 °C.

CONNECTING STATEMENT III

The research in Chapter IV focused on the enzymatic generation of cocoa bean shell oligosaccharides using bi-enzymatic and commercially available multi-enzymatic products. The assessment of the structural components and MW distribution were performed. Similarly, to Chapter III, the prebiotic potential of enzymatically generated oligosaccharides were determined. Chapter IV also explores enzymatic esterification of ferulic acid, otherwise known as feruloylation, to cocoa bean shell oligo- and polysaccharides, using free and immobilized feruloyl esterases from *Humicola insolens* and rumen microorganism. The antioxidant activity of feruloylated polysaccharides were explored using three assays, DPPH, ABTS, and ORAC, and compared to the scavenging activity of ferulic and ascorbic acids.

Chapter V explores the pilot plant scale up for the isolation and enzymatic generation of cocoa bean shell oligo- and polysaccharides accounted for a large quantity with a good understanding of their structural components. Chapter V also examines the incorporation of the scaled-up cocoa bean shell polysaccharides for the development of functionally enriched chocolate-based formulations. Sensory evaluation of the chocolate-based formulations were assessed for their consumer acceptability and purchase intent.

The results from this study have been submitted for publication.

CHAPTER V. PILOT PLANT EXTRACTION OF OLIGO/POLYSACCHARIDES FROM COCOA BEAN SHELLS AND THEIR INCORPORATION INTO CHOCOLATE BASED FORMULATIONS

5.1 Abstract

A pilot plant extraction of cocoa bean shell oligo- (hCHO) and polysaccharide (CHO) extracts using alkali isolation (0.5M KOH) and commercially available multi-enzymatic product, DepolTM 670L, was performed. Compared to the recovery yield attained at a laboratory scale (29.1%), pilotscale CHO extracts were recovered by as much as 3.69%. However, both laboratory and pilot scale extracts contained similar monosaccharide profiles; showing a predominant composition of pectic polysaccharides followed by hemicellulosic polysaccharides as indicated by the presence of uronic acid (55.9%µM), galactose (17.4%µM), glucose (13.3%µM), arabinose (5.8%µM), mannose and xylose (4.7%µM) contents. The MW distribution of the extracts showed four main populations of 0.83, 2.1, 320, and 2400 kDa for the CHO extract, while 0.87, 1.1, 1.6, and 3.3 kDa for the hCHO extract. CHO extracts were utilized for the enrichment of chocolate-based formulations for its use as a functional food ingredient without altering the flavor. Sensory panels consisted of 60 untrained panelists per session, evaluating the formulations before and after enrichment rating the intensity and 9-point hedonic likeness for the following sensory attributes: chocolate flavor, sweetness, sourness, bitterness, and melting in mouth, as well as a binary question for overall acceptance and purchase intent. Overall, chocolate formulations enriched with CHO extracts accounted for higher overall acceptance and purchase intent consumer ratings. Through factor analysis, the correlation and interaction between intensity and likeness attributes were assessed. In addition, boxplot analysis was performed to gain insight regarding the relationship between the perceived complexity of the sensory attributes and the mean quantitative descriptive responses specified by consumers.

5.2 Introduction

Cocoa bean shells (CBS), the outermost layer of the cocoa bean, are one of many by-products heavily accumulated in the cocoa industry. CBS by-product is rich in dietary fiber (13.86-60.6 % w/w) and polyphenols (1.45-94.95 mg GAE/g) (Delgado-Ospina et al., 2020; Lessa et al., 2018; Martínez et al., 2012; Mellinas et al., 2020; Nsor-Atindana et al., 2012; Rojo-Poveda et al., 2020; Utami et al., 2016; Younes et al., 2022). This composition makes CBS a potential biomass for its

valorized use as a source of low-cost, clean label, functional ingredients such us fiber enhancer, stabilizer, and color enhancer (Adamafio et al., 2013; Barišić et al., 2019; Choi et al., 2019; Jozinović et al., 2019; Martínez-Cervera et al., 2011; Okiyama et al., 2017; Panak Balentić et al., 2018; Rojo-Poveda et al., 2020; Younes et al., 2022). However, studies have yet to assess isolated CBS cell wall polysaccharides in food applications. From our previous findings in par 3.4.3 alkaline treatment served as an optimal process for the isolation of cell wall polysaccharides, which was predominantly comprised of pectic regions, namely rhamnogalacturonan, arabinan, and arabinogalactan. In addition, the enzymatic generation of oligosaccharides of diverse degrees of polymerization (DP), ranging from 2 to 10, from CBS cell wall polysaccharides can be achieved using a commercially available multi-enzymatic system namely, Depol[™] 670L. Our previous findings also suggested that CBS oligo- and polysaccharides can be classified as a prebiotic and possess antioxidant properties, thereby broadening its industrial applications (Younes et al., 2022).

Lab-scale practices were optimal towards efficient recovery for structural characterization and compositional analysis yet rendered minimal quantities. Therefore, CBS scale-up efforts were performed to account for a high-volume recovery of CBS polysaccharide extracts. To the author's knowledge, no study has assessed the recovery of CBS oligo- and polysaccharides in a pilot plant scale. The present study investigates the efficiency of two approaches using alkaline extraction and commercially available multi-enzymatic system (DepolTM 670L) for the recovery of CBS carbohydrate (polysaccharide) and hydrolyzed carbohydrate (oligosaccharide) extracts at a pilot-scale level. In addition, this study also explores the enrichment of chocolate-based formulations composed of cocoa, carob at selected ratios with CBS carbohydrate extracts. This study will generate consumer predictive models thereby establishing a link between key sensory properties of enriched chocolate formulations and consumer acceptability and purchase intent.

5.3 Materials and Methods

5.3.1 Materials

Commercial samples of CBS (CCN 51) were supplied by (Guayaquil, Ecuador). Commercial multi-enzymatic system Depol[™] 670L from *Trichoderma reesei*, was provided by Biocatalyst Ltd (IL, USA). 1-Kestose, nytose, and 1F-fructofuranoysl-nystose were obtained from Wake Pure Chemical (Osaka, Japan). Food grade potassium hydroxide was purchased from Spectrum Chemical Mfg Group (New Brunswick, NJ, USA). Food grade hydrochloric acid as well as

analytical grade reagents were from Sigma- Aldrich Co., St. Louis, MO, USA. Salts were obtained from Fisher Scientific (Fair Lawn, NJ, USA).

5.3.2 Lab-scale extraction of cocoa bean shell polysaccharides

The cell wall material (CWM) was sequentially isolated as described by Redgwell et al. (2003). Using de-fatted CBS (4g), the isolation proceeded as follows: 1) 80% ethanol (2hrs, 50 rpm), 2) chloroform: methanol (1:1, 1.5 hrs, 50 rpm), 3) chloroform: methanol (1:1, 1.5 hrs, 50 rpm), 4) phenol: acetic acid: water (PAW, 2:1:1, overnight), 5) phenol: acetic acid: water (PAW, 2:1:1, 2hrs). Each treatment was followed by centrifugation (7,000xg, 15mins) and vacuum filtration prior to the following step. The recovered CWM was dialyzed (5-8kDa cutoff) against distilled water and freeze-dried (-50°C, 0.003 mbar). Extraction of CBS polysaccharides was performed as described by Khodaei and Karboune (2013) where CWM material (2% w/v) was suspended in 0.5M KOH solution containing 0.02M NaBH₄. Mixtures were incubated (60°C, 24 hrs), thereafter centrifuged (10,000 x g for 15 mins) and recovered via filtration (0.45µm). Recovered polysaccharides were neutralized with HCl, dialyzed (5-8 kDa cutoff) and freeze-dried (-50°C, 0.003 mbar).

5.3.3 Pilot-plant extraction of cocoa bean shell oligo- and polysaccharides

CBS of CCN 51 variety were suspended in 751 KOH solution (0.5M) to reach a concentration of 2% (w/v) with constant stirring for 3 hr at 60°C. The suspension was then left with constant stirring for 18hr at 25°C. The solution was neutralized with HCl to a desired pH between 6 to 8. The resulting suspension was subjected to two rounds of centrifugation, basket centrifugation using the Western States STM-2000 (3600 rpm, MWCO basket 100um) followed by centrifugation with a speed of 11,000 rpm, feed rate 3PSI using the Dexter MiSR 1010 (SRS A USI Company, MI, USA). The filtrate was collected and subjected to subsequent ultra- and diafiltration (1 time) steps, on the recovered precipitate, using a Koch Hollow Fiber Cartridge (with a MWCO of 5000 Da). A fraction of the retentate was collected and lyophilized (Sublimator 50-3 EKS, Zirbus Technology). The freeze-dried powder of pilot-scale cocoa bean shell polysaccharides is further abbreviated as CHO extract. The remaining fraction was taken further for enzymatic-based hydrolysis for the generation of oligosaccharides. The pH of the solution was adjusted to pH 5.5 to achieve the optimal conditions of the multi-enzymatic product, DepolTM 670L (135 galactanase units/ml) which were suspended with the CHO extract with constant stirring for 3hr at 40°C. The

hydrolysate was collected and lyophilized. The freeze-dried powder of pilot-scale cocoa bean shell oligosaccharides is further abbreviated as hydrolyzed carbohydrate extract, hCHO extract.

5.3.4 Analytical methods for the quantification and characterization of carbohydrate extracts

5.3.4.1 Determination of total neutral sugars and uronic acid contents

Uronic acid content was measured by sulphamate/m-hydroxydiphenyl assay (Blumenkrantz & Asboe-Hansen, 1973). The phenol–sulphuric acid colorimetric assay was used for the determination of neutral sugar content (DuBois et al., 1956).

5.3.4.2 Monosaccharide profile

The carbohydrate extracts were subjected to acid hydrolysis according to the method of Spadoni Andreani et al. (2021). Briefly, 1 ml of a mixture of HCl/methanol (1:4 v/v) was added to the sample (200 μ l of liquid extract, or 5% (w/v) of solid in water). The mixtures were incubated at 70 °C for 24 hr and dried by heating in a convection incubator (Excella E24, New Brunswick Scientific, Canada). To complete hydrolysis, 3 ml of water and 1 ml of trifluoroacetic acid were added and heated at 100 °C for 1 hr. After evaporating the trifluoroacetic acid. The samples were neutralized with NaOH and centrifuged (8,000 rpm, 5 min). The supernatants (0.02 ml) were analyzed by high performance anion exchange chromatography (HPAEC) on a Dionex ICS 3000 system equipped with pulsed amperometric detection using a Carbopac PA-20 column (Dionex Co., Sunnyvale, CA, USA) at a temperature of 30 °C. Mobile phase was 20mM NaOH at flow rate of 0.4 ml/min. Rhamnose, arabinose, glucose, xylose, galactose, and mannose were used at varying concentrations (2.5 – 50 μ M) as standards.

5.3.4.3 Molecular weight profile

A high-performance size-exclusion chromatography (HPSEC) system (Model 1525 binary HPLC pump, equipped with a Model 2414 refractive index detector, Waters Co., Milford, MA, USA) was used to estimate the molecular weight distribution of the carbohydrates. Analysis was determined using a TSK G5000 PWXL (Tosoh Co., Yamaguchi, Japan) column, with dextrans (50–670 kDa), soybean rhamnogalacturonan (0.125-1 g/l) and galactoglucomannan (0.125-1g/L) standards for molecular weight and concentration calibration. The temperature of the system was 30 °C, the eluent was 0.1 M NaCl, and the flow rate was set at 0.4 ml/min.

5.3.5 Preparation of chocolate

Chocolate formulations, containing 40-65% cocoa mass with and without 25-50% (w/w) carob powder, were prepared according to CODEX STAN 87 -1981 (Codex Alimentarius, 1981) with slight modifications. The organoleptic properties of chocolate formulations were enhanced with the addition of cane sugar (10.5-11.25%, w/w) and soy lecithin (0.05%, w/w). Formulations were prepared by combining cocoa butter (15-19.5%), carob powder (25-50%), and soy lecithin (0.05%) together. In a separate mixture, cocoa mass (40-65%) and cane sugar (10.5-11.25%) were combined. Batters were melted in the microwave, with mixing and scraping of bowl sides at uniform intervals. Once a uniform mixture consistency was achieved, the two batters were blended using a Robot Coupe Food Processer (R100) with blade attachment for 10 minutes with scraping of bowl sides after each minute, this allowed for the development of a uniform emulsion. Batters were then tempered to 32°C before molding. Three selected chocolate formulations were enriched with CHO extract from CBS (1%), namely 40% Cocoa/Carob/CHO (75:24:1, w/w/w), 40% Cocoa/Carob/CHO (50:49:1), and 65% Cocoa/Carob/ CHO (75:24:1, w/w/w). Table 1 shows the proportion of ingredients for formulations with and without CHO extract. Each chocolate bar consisted of 30 grams of combined batter and were left to cool at 4°C. Following cooling, chocolates were analyzed for sensory properties. Measurements were done the same day as baking.

5.3.6 Sensory evaluation of chocolate formulations

The chocolates were analyzed for their intensity and likeness using untrained panelists consisting of 60 McGill University students, staff, and faculty. The sensory attributes were evaluated using the Compusense software (Compusense Inc, On, CA) and a mean quantitative descriptive 9-point hedonic scale (extremely dislike/weak intensity, moderately dislike/weak intensity, dislike/weak intensity, slightly dislike/weak intensity, neutral liking/intensity, slightly like/strong intensity, like/strong intensity, moderately like/strong intensity, extremely like/strong intensity).

			% Chocolate Proportion (w/w)				
	Formulation	Chocolate:Carob:CHO	Cocoa mass	Cocoa butter	Sugar	Lecithin	
Before addition	40% Cocoa	100:0:0	40	39	20.9	0.1	
	40% Cocoa + Carob	75:25:0	40	39	20.9	0.1	
	65% Cocoa + Carob	75:25:0	65	19.9	15	0.1	
	40% Cocoa + Carob	50:50:0	40	39	20.9	0.1	
	65% Cocoa + Carob	50:50:0	65	19.9	15	0.1	
After addition	40% Cocoa + Carob + CHO	75:24:1	40	39	20.9	0.1	
	40% Cocoa + Carob + CHO	50:49:1	40	39	20.9	0.1	
	65% Cocoa + Carob + CHO	75:24:1	65	19.9	15	0.1	

Table 5.13. Chocolate formulation composition before and after addition of carbohydrate extract

The panelists were considered untrained as they were not familiarized with the attributes prior or during testing. The sensory attributes consisted of chocolate flavor, sweetness, sourness, bitterness, and melting in mouth. Moreover, panelists were asked three binary choice questions (yes/no) regarding if the attributes were too complex to assess, overall acceptance, and purchase intent of the product.

5.3.7 Statistical Interpretation

Statistical analysis of the saccharide profile was performed using XLSTAT software (Addinsoft, New York, NY, USA) in Microsoft Excel (Microsoft, Redmond, WA, USA). Oneway analysis of variance (ANOVA) and the Tukey's honest significant difference (HSD) test were performed to detect significant differences (P < 0.05). Regarding the sensory evaluation data, Statistical Analysis System (SAS version 9.4) was used. The Proc Means was used to calculate the average and standard deviation of quantitative descriptive scores and liking scores. The Proc Can was used to assess the multivariate analysis of variance and canonical results, which were used to interpret the relationship between attributes and formulations. Consumer data were coded with a positive response denoted as 1 and a negative response as 2. The Proc Logistic was used to relate the consumer response for overall acceptance and purchase intent to the liking scores of attributes according to the following logistic function:

$$P\left(\frac{1}{x}\right) = \exp(\alpha + \beta_1 x_1 + \beta_2 x_2 + \dots + \beta_n x_n) / \{1 + \exp(\alpha + \beta_1 x_1 + \beta_2 x_2 + \dots + \beta_n x_n)$$

Where, P (1/x) is the probability of overall acceptance or purchase intent; x_1 , x_2 ,..., x_n are the intensities of attributes; and α , β_1 , β_2 ,..., β_n are parameter estimates associated with the model terms (Malundo et al., 2001). Additionally, factor analysis was performed on the mean scores using RStudio (RStudio, Inc., Version 4.0.2) for all samples to visually observe sample grouping and differentiation. To assess the correlations between sensory attributes, a heatmap was created using RStudio (RStudio, Inc., Version 4.0.2).

5.4 Results and Discussion

5.4.1 Recovery of carbohydrate extracts using laboratory and pilot-plant scale up isolation

CBS were used in a continuous flow pilot-plant facility for the isolation of their CHO (enriched with polysaccharides) and hCHO (enriched with oligosaccharides) extracts using alkaline and

commercial multi-enzymatic treatment, respectively. Conversely, the lab-scale process was performed as an individual batch system. For batch comparisons, lab-scale extraction commenced using 100 mL total volume and 2% (w/v) CBS cell wall material, while pilot-plant extraction consisted of 88.31 L total volume and 2% (w/v) CBS starting material. Figure 5.5 shows the pilotplant process, where CBS (2%, w/v) was suspended in potassium hydroxide (0.5 M) with constant stirring for 3 hours at 60°C followed by overnight incubation at 25°C. Two consecutive rounds of centrifugation were necessary for retentate and precipitate separation, one using a basket centrifuge, followed by a pilot-scale centrifugation step. In pilot-scale the CBS suspension mimicked a gelatinous consistency, making it difficult to separate the retentate from the precipitate; this was attributed to the rich pectin content in CBS. This phenomenon was encountered at the lab-scale, however to a lesser extent. This can be attributed towards the use of NaBH₄ as a reducing agent, at the lab-scale level. Without a reducing agent, as is the case at the pilot-scale, accelerated hydrolysis rates are facilitated, contributing to a higher degree of debranching (Minkina et al., 2012). The suspension was then subjected to ultra- and diafiltration steps to isolate and concentrate, which yielded an extract rich in pectic and hemi-cellulosic polysaccharides (CHO). Diafiltration steps were performed to remove interfering salts, accumulated from the alkaline extraction and neutralization steps (Resende et al., 2012). That being said, a balance should be established between recurring rounds of diafiltration as they are attributed to significant losses due to carbohydrate adhesion to the membrane filter. In addition, membrane fouling, indicated by irreversible loss of membrane permeability is possible which can prompt further column washing and chemical cleaning (Pu et al., 2012). The second process commenced with CHO extract, enriched with polysaccharides, in which Depol[™] 670L multienzymatic system was added; the enzymatic hydrolysis for the efficient release of oligosaccharides was carried out with constant gentle stirring for 3 hrs at 40°C (Figure 1).

Table 5.14 shows the recovery yield of the CHO extract on the lab- and pilot-scales. The crude extract, containing mono-, di- and saccharides greater than 5 kDa, was recovered by 78.4% (w/w), with some loss attributed to the sequential centrifugation processes as they follow a continuous flow. Following ultra/diafiltration and freeze drying steps, the CHO extract recovery yield was achieved at 3.69% (w/w) at the pilot scale level. This yield is significantly (P < 0.05) lower than that recovered at the lab-scale level (29.1%, w/w).



Figure 5.5. Flow diagram of pilot-scale production of cocoa bean shell (CBS) oligo- and polysaccharides by alkaline extraction (CHO extract) and upon commercial multi-enzymatic system DepolTM 670L treatment (hCHO extract).

The low recovery can be explained by the high proportion of saccharides lower than 5 kDa contained within the extract, removed during ultra- and diafiltration, as the focus was on the recovery of polysaccharides (> 5kDa). Waglay et al. (2019) reported similar findings where lower yields were attained at the pilot-scale level (21.6-44.6%) for the recovery of potato proteins when compared to laboratory extraction (50.9-74.4%). On the contrary, the total carbohydrate content at the pilot-scale level (90.6% w/w) was higher than at the lab-scale level (78.9% w/w), where significantly (P < 0.05) higher proportion of neutral sugars were recovered. This was attributed to the interference of proteins present within the extracts at the lab-scale level as reported in our previous study.

The monosaccharide profile of the extracts, recovered at the lab-scale level, showed a distribution of both pectic and hemicellulosic regions, indicated by the high proportion of uronic acid (51.4%), rhamnose (12.4%), mannose and xylose (12.6%), and arabinose (8.6%). Alternatively, the extract recovered at the pilot-scale level, resulted in higher proportions of uronic acid (55.9%), galactose (17.4%) and glucose (13.3%) indicating the presence of rich pectic regions. It is important to note that lab-scale alkaline extraction incorporated a reducing agent, sodium borohydride (NaBH₄). This may explain the higher pectic regions recovered at the pilot-scale level, where the exclusion of NaBH₄ mediated faster hydrolysis rates than at the lab-scale level (Minkina et al., 2012).

Figure 5.3 shows the MW distribution of CHO and hydrolyzed CHO extracts retrieved at pilotscale as determined by size exclusion chromatography. The presence of four main populations are observed for both extracts, where CHO populations were found at 0.83, 2.1, 320, and 2400 kDa; similar to those reported in par 3.4.1 for hydrolyzed CHO populations were found at 0.87, 1.1, 1.6, and 3.3 kDa. This shows that high enzymatic hydrolytic activity was expressed by DepolTM 670L, which led to the enzymatic generation of oligosaccharides of varying DP compared to isolated polysaccharides in the CHO extract.



Figure 5.6. Images of equipment utilized in the pilot-plant extraction process, A - 250L double jacketed vessel equipped with an immersion mixer; B – Basket centrifugation (Western States STM-2000) equipped with 100 μ m MWCO; C – Centrifuge (Dexter MiSR 1010); D – Ultra- and diafiltration with Koch Hollow Fiber Cartridge (5000 Da MWCO); E – Lyophilizer (Sublimator 50-3 EKS, Zirbus Technology).

Table 5.14. Effects of extraction process scale-up of cocoa bean shell polysaccharides (carbohydrate extraction) on the recovery yield, total carbohydrate content, neutral sugar and uronic acid contents, and the relative proportion of saccharide profile.

		Carbohydrate extract		
		Laboratory Scale	Pilot Scale	
Crude extract recovery yield (%, w/w)*		-	78.4	
Extract recovery yield (%, w/w)		29.1 ^a	3.69 ^b	
Total CHO content (%, w/w)		78.9	90.6	
Neutral Sugar (%, w/w)		$20.6{\pm}0.7^{a}$	30.5 ± 7.6^{b}	
Uronic content (%, w/w)		58.3 ± 5.2^{a}	60.1±22.3ª	
Saccharide Profile (%, rel. proportion)	Fucose	$1.8{\pm}0.02^{a}$	$1.8{\pm}0.1^{b}$	
	Rhamnose	12.4±0.3ª	$1.2{\pm}0.1^{b}$	
	Arabinose	$8.6{\pm}0.01^{a}$	5.8±1.1ª	
	Galactose	$8.1{\pm}0.2^{a}$	17.4 ± 1.1^{b}	
	Glucose	$5.2{\pm}0.5^{a}$	13.3±2.9 ^b	
	Mannose,			
	Xylose	$12.6{\pm}2.0^{a}$	$4.7{\pm}0.4^{b}$	
	Uronic acids	51.4±3.1ª	55.9±1.3 ^b	

Values reported as mean \pm standard deviation

*Calculated as recovery yield prior to ultra- and diafiltration

 $^{\rm a\text{-}b}$ Within the same row, means with different letters are significantly different at $P \le 0.05$



Figure 5.3. Size exclusion chromatogram of carbohydrate extract recovered in pilot scale by alkaline extraction and hydrolyzed carbohydrate extract recovered using commercial multi-enzymatic product (DepolTM 670L)

5.4.2 Chocolate-based formulations enriched with carbohydrate extract

5.4.2.1 Statistical interpretation of chocolate-based formulations

In order to assess multiple attributes on a single set of chocolate formulation design, multivariate statistics analysis was used (Lawless & Heymann, 2010). Table 5.3 shows the multivariate analysis of variance (MANOVA) results, represented by *F-values* greater than 1 and *P values* less than 0.05, which indicate that the chocolate formulations were perceived differently by the panelists before addition of the CHO extract. Alternatively, this was not that case for chocolate formulations upon enrichment, suggesting that the addition of CHO prompted consumers to perceive no significant differences between the samples. Canonical discriminant analysis can be used to compliment MANOVA by determining the multi-dimensional mean separation of the attributes, otherwise known as the canonical coefficient (CAN), allowing us to determine the attribute(s) responsible for the group differences (Lawless & Heymann, 2010). Our results indicate that chocolate flavor intensity (0.7278), melting in mouth intensity (0.7123), bitterness intensity (0.7155), chocolate flavor likeness (0.4664), and bitterness likeness (0.4536) are the attributes most responsible for group differences before addition of CHO extract.

Figure 5.4 outlines the sensorial responses of 60 panelists to the following intensity and likeness sensory scores of chocolate flavor, sweetness, bitterness, sourness and melting in mouth before and after the addition of CHO extract in chocolate-based formulations containing carob. Overall, the results show that higher likeness scores for all attributes were obtained with the enrichment of CHO extract in chocolate-based formulations. On the other hand, intensity scores of chocolate flavor, sweetness, bitterness, and sourness were decreased with the addition of CHO. This suggests that CBS polysaccharides play a role on masking attribute intensities while achieving desirable flavor profiles of chocolate-based formulations. It is possible that reduced intensities, with accompanying higher likeness scores for bitterness, are attributed to the enrichment with carob. There is an opposite trend for the 40% cocoa + carob (50:50) formulation, where an increase in the intensity is observed, further revealing the contribution of CBS CHO extract to the sweetening and chocolate flavor masked by the use of higher proportion of carob. García-Díez et al. (2022) reported carob contributed to reduced bitterness scores due to its rich dietary fibre (55.67 g/100g) and soluble sugar content (7.87 g/100g), namely glucose and fructose. Interestingly, only one attribute, namely, melting in the mouth showed increases in both intensity and likeness scores.
MANOVA						
	Statistic	Value	F Value	Num DF	Den DF	Pr > F
Before addition	Wilks' Lambda	0.660	3.37	30	664.03	<.0001
	Pillai's Trace	0.370	3.21	30	684.00	<.0001
	Hotelling-Lawley Trace	0.470	3.52	30	508.55	<.0001
	Roy's Greatest Root	0.356	8.12	10	228.00	<.0001
After addition	Wilks' Lambda	0.887	0.93	30	664.03	0.581
	Pillai's Trace	0.117	0.92	30	684.00	0.5881
	Hotelling-Lawley Trace	0.124	0.93	30	508.55	0.5724
	Roy's Greatest Root	0.087	1.99	10	228.00	0.0353
Canonical Analysis						

Table 5.15. Statistical interpretation of chocolate formulation design prepared with carob powder before and after addition of pilot-scale carbohydrate (CHO) extract

	Variable	Can1	Can2
Before addition	Chocolate Flavor Intensity	0.727821	0.10073
	Sweetness Intensity	0.604509	-0.032326
	Bitterness Intensity	0.71553	0.169553
	Sourness Intensity	0.564908	0.361457
	Melting in Mouth Intensity	0.712296	-0.218191
	Chocolate Flavor Likeness	0.466456	0.388913
	Sweetness Likeness	0.115789	-0.093993
	Bitterness Likeness	-0.453639	0.219391
	Sourness Likeness	-0.319053	0.515693
	Melting in Mouth Likeness	-0.166455	-0.00337



Figure 5.4. Average mean quantitative descriptive scores of selected chocolates enriched with and without carob and carbohydrate extract from pilot scale based on intensity (A) or likeness (B): 40% cocoa (blue); 40% cocoa + carob (75:25, orange); 65% + carob (75:25, gray); 40% cocoa + carob (50:50, yellow); 40% cocoa + carob + CHO (75:24:1, light blue); 40% cocoa + carob + CHO (50:49:1, green); 65% cocoa + carob + CHO (75:24:1, dark blue).



Figure 5.5. Boxplot analysis for perceived complexity of sensory attributes (chocolate flavor, sweetness, bitterness, sourness, melting in mouth) chocolates enriched with carob and carbohydrate extract from pilot scale based on before addition - intensity (A) likeness (B) and after addition (right) - intensity (C) and likeness (D).

As seen in Table 5.4, the proportion of overall acceptance and purchase intent responses increased by as much as 16% and 14% likely suggesting that consumers favored formulations enriched with CBS CHO extract. Logistic regression analysis (LRA) can be applied using dependant responses (probability of overall acceptance and purchase intent of chocolate formulation) with independent variables (sensory attributes) to behave according to a multivariate normal distribution (Malundo et al., 2001; Waglay & Karboune, 2020). Determinants impacting the overall acceptance and purchase intent are identified based on the lowest *P-value* and highest odds ratio scores. According to LRA results (Table 5.4), chocolate flavor intensity and sweetness likeness were found to be the most significant ($\alpha = 0.05$) and determinant factors for the overall acceptance and purchase intent of chocolate-based formulations enriched with CBS carbohydrate extract.

Consumers' preferences are influenced by several factors, one being the perceived complexity of sensory attributes. Berlyne (1967) states that perceived complexity and hedonic measurement are interconnected, where an individual's hedonic response to a stimulus increases with its complexity until an optimal threshold is achieved followed by a decline, resembling an inverted U-curve. Figure 5.5 depicts the boxplot analysis of perceived complexity of sensory attributes in terms of chocolate flavor, sweetness, bitterness, sourness, and melting in mouth of chocolate-based formulations before (Figures 5.5A and 5.5B) and after enrichment with CBS CHO extract (Figures 5.5C and 5.5D). Indication of differences in the perceived complexity of attributes is identified by mean scores (bold line) residing outside the comparison boxplots. Prior to enrichment, there lies a divide between the perceived complexity of intensity attributes, in which chocolate flavor, sweetness and bitterness attained lower degree of perceived complexity compared to sourness and melting in mouth. Interestingly, in terms of the complexity of likeness attributes, bitterness and sourness are perceived with a lower degree of complexity compared to chocolate flavor, sweetness, and melting in mouth. Upon enrichment of the chocolate-based formulations, boxplot analysis suggests no differences among the intensity and likeness attributes in terms of their perceived complexity. According to Köster & Mojet (2007a, 2007b), exposure to a lower complexity will have no effect on the consumer's optimal complexity level. This may explain the account of no observed differences in the perceived complexity of intensity and likeness attributes upon enrichment.

Table 5.4. Percentage overall acceptance and purchase intent of selected chocolate-based formulations and the statistical interpretation of chocolate-based formulation prepared with carob and pilot-scale CHO extract based on logistic regression analysis (P < 0.05).

	% Overall	% Purchase Intent	
	Acceptance		
40% Cocoa + Carob + CHO (75:24:1)	89	68	
40% Cocoa + Carob + CHO (50:49:1)	82	69	
65% Cocoa + Carob + CHO (75:24:1)	86	68	
65% Cocoa + Carob (50:50:0)	73	55	

	8	8	v			
	Overall Acceptance			Purchase Intent		
	Estimate	$\Pr > \chi^2$	Odds Ratio	Estimate	$Pr > \chi^2$	Odds Ratio
Chocolate Flavor Intensity	0.48	<.0001	1.62	0.35	<.0001	1.42
Sweetness Intensity	0.10	0.30	1.11	0.01	0.91	1.01
Bitterness Intensity	-0.01	0.93	0.99	0.02	0.83	1.02
Sourness Intensity	0.05	0.63	1.05	-0.06	0.49	0.94
Melting in Mouth Intensity	0.17	0.22	1.18	0.23	0.03	1.26
Chocolate Flavor Likeness	-0.15	0.38	0.86	0.12	0.30	1.13
Sweetness Likeness	0.59	0.00	1.80	0.31	0.01	1.37
Bitterness Likeness	0.02	0.86	1.02	0.13	0.16	1.14
Sourness Likeness	-0.13	0.30	0.88	-0.05	0.61	0.95
Melting in Mouth Likeness	0.44	0.00	1.56	0.09	0.40	1.10

Logistic Regression Analysis

In addition, enrichment may have imparted a synergistic blend among sensory attributes towards the consumers experience. Furthermore, studies suggest that perceived complexity are influenced by several factors, namely the number of attributes assessed, the consumers level of expertise, familiarity, or gender (Lawless & Heymann, 2010; Livermore & Laing, 1998; Palczak et al., 2019). Humans have a limited capacity to distinguish many attributes, suggesting that sensory attributes should be confined to no more than four at a time (Lawless & Heymann, 2010).

5.4.2.2 Correlation of sensory attributes

A correlation study of intensity and likeness sensory attributes was performed to determine the strongest correlations impacting panelists' responses. Figure 5.6 presents the correlation coefficients of intensity and likeness attributes before and after enrichment of CHO extract in chocolate-based formulations. The results prior to enrichment suggest that chocolate flavor likeness with sweetness likeness (R = 0.69), melting in mouth likeness with chocolate flavor likeness (R = 0.64), and melting in mouth likeness with sweetness likeness (R = 0.61) were viewed as the strongest positive correlations with significance (P < 0.001). In addition, sourcess intensity with melting in mouth intensity (R = 0.49), bitterness intensity with sourcess intensity (R = 0.43), and bitterness intensity with melting in mouth intensity (R = 0.34) were viewed as moderately positive correlations. Interestingly, only one correlation between intensity and likeness attributes was observed, being chocolate flavor intensity with melting in mouth likeness (R = 0.34). Contrary findings were reported by (Leite et al., 2013), who found a negative correlation (R = -0.89) between melting quality, defined as the length of time for chocolate to melt in the mouth, with chocolate flavor, defined as the residual of dark chocolate flavor. The length of melting time is directly proportional to the particle size distribution of non-fat solids within chocolates, where studies have shown the relationship significantly contributing towards the sensory quality (Beckett, 1994; Lee et al., 1992; Pangborn et al., 1973; Ziegler et al., 2001). Indeed, according to several studies, the particle size can impact the perceived intensities and duration of the perception of flavor, sweetness and other attributes (Beckett, 1994; Lee et al., 1992; Pangborn et al., 1973; Ziegler et al., 2001).

Enrichment of chocolate-based formulations provided a higher extent of correlations with significance (P < 0.001) between intensity and likeness attributes, which include chocolate flavor likeness with chocolate flavor intensity (R = 0.65), chocolate flavor intensity with sweetness likeness (R = 0.53), melting in mouth likeness with melting in mouth intensity (R = 0.5), chocolate

flavor intensity with sourness likeness (R = 0.46), and chocolate flavor intensity with bitterness likeness (R = 0.44). Overall, chocolate flavor intensity showed moderate to strong correlations among all likeness attributes, suggesting that CBS polysaccharide enrichment may have played a role in the development of a complex chocolate matrix, due to the increase in intensity and likeness attribute interactions. One study conducted by (Al Sheraji et al., 2017) observed similar responses with polysaccharides, derived from mango peels, in non-fat yoghurt, which can suggest that polysaccharide enrichment can impart better mouthfeel consistencies in even fat-free food formulations

To show an overview of the sample variation in relation to perceived complexity and the correlation of intensity and likeness sensory attribute individually, factor analysis was applied. Figure 5.7 presents the two-dimensional (MR1, MR2) biplots generated for intensity and likeness sensory attributes before and after enrichment with CHO extract. Before enrichment, the cumulative variance accounts for 42.5% and 69% of the dataset; where MR1 accounts for 23.7% and 43.7% of the variance for intensity and likeness biplots. Each attribute is represented by a vector, where the vector length expresses the degree of variance; suggesting that a longer vector would indicate higher variance. In addition, the angle of the vector would indicate the degree of correlation between the attributes, where angles smaller than 90° indicate strong correlations.

As seen in Figure 5.7A, all intensity attributes show positive correlations, where the strongest correlations are observed for both, sourness and melting in mouth with bitterness. Moreover, sourness shows the highest level of variance, suggesting that this attribute is most responsible for the differences within the dataset. Alternatively, weaker correlations, signified by angles smaller than 90°, are observed for both chocolate flavor with sweetness and sourness. Furthermore, chocolate flavor and sweetness attributes have a lower contribution towards the dataset, considering their low level of variance as determined by their small vector lengths.



Figure 5.6. Correlation between intensity and likeness scores of each attribute, chocolate flavor, sweetness, bitterness, sourness, and melting in mouth, before (A) and after (B) addition of CHO extract in chocolate formulations; * P < 0.05, ** P < 0.01, *** P < 0.001.



Figure 5.7. Bi-plot factor analysis of sensory attributes (chocolate flavor, sweetness, bitterness, sourness, melting in mouth) chocolates enriched with carob and carbohydrate extract from pilot scale based on before addition - intensity (A) likeness (B) and after addition (right) - intensity (C) and likeness (D).

In Figures 5.7C and 5.7D, the cumulative variance of the two dimensions represents 34.8% and 70.9%, with MR1 which explains the 18.9% for intensity and MR2 which explains the 47.3% for likeness. Similar to the intensity biplot, sourness is observed as the attribute responsible for differences within the dataset, while a shared level of variance is observed for the remaining attributes for likeness (Figure 5.7B). Strong positive correlations are observed between chocolate flavor with sweetness and melting in mouth with bitterness; while weak correlations were found for sourness with sweetness. Based on the high proportion of variance, where chocolate flavor (for intensity) and bitterness (for likeness) were responsible for the differences after enrichment (Figures 5.7C and 5.7D). All attributes expressed positive correlations, with strongest interactions being observed between chocolate flavor with bitterness for intensity. On the other hand, likeness attributes only showed weak positive correlations between chocolate flavor, sweetness, melting and sourness.

5.5 Conclusion

A pilot-plant scale up process for the isolation of CBS oligo- and polysaccharides using commercially available multi-enzymatic system and an alkali treatment was developed. As expected, the recovery yields were lower than those encountered at lab-scale. However, the saccharide profile was comparable; showing that pilot plant CBS polysaccharides contained a combination of pectic and hemi-cellulosic polysaccharides based on the uronic acid ($55.9\%\mu$ M), galactose (17.4%µM), glucose (13.3%µM), arabinose (5.8%µM), as well as the mannose and xylose (4.7%µM) contents. Saccharides found in both extracts were predominantly of DP 2-5 (85.2% w/v) for CHO and DP 6-9 (57.6% w/v) for hCHO extracts. Based on the MW distribution profile, four main populations were found for each CHO (0.83, 2.1, 320, and 2400 kDa) and hCHO (0.87, 1.1, 1.6, and 3.3 kDa) extract. CBS polysaccharides were used for the enrichment of chocolate-based formulations, where sensory evaluation panels were held to assess the intensity and likeness of the chocolate flavor, sweetness, bitterness, sourness, and melting in mouth attributes. Based on the canonical discriminant analysis before enrichment, the formulations were perceived as significantly different from one another with chocolate flavor (intensity and likeness), bitterness (intensity and likeness), melting in mouth (intensity) which were responsible for group differences. Spider-plot analysis suggested that higher likeness responses were attained after enrichment, with a significant increase observed for melting in mouth. Furthermore, the perceived

complexity of the intensity and likeness attributes varied before and after enrichment. The results suggested that enrichment of chocolate-based formulations mediated no differences between the perceived complexity of the attributes. According to the correlation heatmaps, more interactions between intensity and likeness of the attributes was observed after enrichment, likely indicating that CBS polysaccharides contributed towards the development of a diverse and complex formulation matrix. Overall, all three enriched formulations attained highest overall acceptance (82-89%) and purchase intent responses (68-69%).

References

- Adamafio, N. A., Kyeremeh, K., Matey, A. T., Mingle, C., & Kolawole, O. M. (2013). Theobromine Toxicity and Remediation of Cocoa By-products: An Overview. *Journal of Biological Sciences*, 13(7), 570–576. https://doi.org/10.3923/jbs.2013.570.576
- Agus, B. A. P., Mohamad, N. N., & Hussain, N. (2018). Composition of unfermented, unroasted, roasted cocoa beans and cocoa shells from Peninsular Malaysia. *Journal of Food Measurement and Characterization*, 12(4), 2581–2589. https://doi.org/10.1007/s11694-018-9875-4
- Al Sheraji, S. H., Alburihi, S., Nasser, Y. H., & Mohammed Nagi, Y. (2017). Effects of Polysaccharides from Mango' Peel on Physiochemical and Sensory Properties of Non-Fat Yoghurts. *Advances in Dairy Research*, 05(03). https://doi.org/10.4172/2329-888X.1000181
- Bamba, Y., Ouattara, N., Ouattara, S., Ouatarra, A., & Gourene, G. (2015). Effect of diets containing cocoa bean shell and coconut oil cake on the growth of Oreochromis niloticus (LINNE, 1758) in pond. *International Journal of Biological and Chemical Sciences*, 8(4), 1368. https://doi.org/10.4314/ijbcs.v8i4.2
- Barišić, V., Jozinović, A., Flanjak, I., Šubarić, D., Babić, J., Miličević, B., Jokić, S., Grgić, I., & Ačkar, D. (2019). Effect of Addition of Fibres and Polyphenols on Properties of Chocolate – A Review. *Food Reviews International*, 1–19. https://doi.org/10.1080/87559129.2019.1701008
- Beckett, S. T. (1994). Control of particle size reduction during chocolate grinding. *Manufacturing Confectioner*, 74, 80.
- Berlyne, D. E. (1967). Arousal and reinforcement. *Nebraska Symposium on Motivation*, 15, 1–110.
- Blumenkrantz, N., & Asboe-Hansen, G. (1973). New method for quantitative determination of uronic acids. *Analytical Biochemistry*, 54(2), 484–489. https://doi.org/10.1016/0003-2697(73)90377-1
- Choi, J., Kim, N., Choi, H. Y., & Han, Y. S. (2019). Effect of Cacao Bean Husk Powder on the Quality Properties of Pork Sausages. *Food Science of Animal Resources*, 39(5), 742–755. https://doi.org/10.5851/kosfa.2019.e62
- Delgado-Ospina, J., Di Mattia, C. D., Paparella, A., Mastrocola, D., Martuscelli, M., & Chaves-Lopez, C. (2020). Effect of Fermentation, Drying and Roasting on Biogenic Amines and Other Biocompounds in Colombian Criollo Cocoa Beans and Shells. *Foods*, 9(4), 520. https://doi.org/10.3390/foods9040520
- DuBois, M., Gilles, K. A., Hamilton, J. K., Rebers, P. A., & Smith, F. (1956). Colorimetric Method for Determination of Sugars and Related Substances. *Analytical Chemistry*, 28(3), 350–356. https://doi.org/10.1021/ac60111a017
- Fakhlaei, R., Rozzamri A., & Hussain, N. (2019). Composition, color and antioxidant properties of cocoa shell at different roasting temperatures. *Food Research*, 4(3), 585–593. https://doi.org/10.26656/fr.2017.4(3).251
- García-Díez, E., Sánchez-Ayora, H., Blanch, M., Ramos, S., Martín, M. A., & Pérez-Jiménez, J. (2022). Exploring a cocoa–carob blend as a functional food with decreased bitterness: Characterization and sensory analysis. *LWT*, *165*, 113708. https://doi.org/10.1016/j.lwt.2022.113708
- Jozinović, A., Panak Balentić, J., Ačkar, D., Babić, J., Pajin, B., Miličević, B., Guberac, S., Vrdoljak, A., & Šubarić, D. (2019). Cocoa husk application in the enrichment of extruded

snack products. *Journal of Food Processing and Preservation*, 43(2), e13866. https://doi.org/10.1111/jfpp.13866

- Khodaei, N., & Karboune, S. (2013). Extraction and structural characterisation of rhamnogalacturonan I-type pectic polysaccharides from potato cell wall. *Food Chemistry*, 139(1–4), 617–623. https://doi.org/10.1016/j.foodchem.2013.01.110
- Khodaei, N., & Karboune, S. (2016). Enzymatic generation of galactose-rich oligosaccharides/oligomers from potato rhamnogalacturonan I pectic polysaccharides. *Food Chemistry*, 197, 406–414. https://doi.org/10.1016/j.foodchem.2015.10.122
- Köster, E. P., & Mojet, J. (2007a). Boredom and the reasons why some new food products fail. In Consumer-Led Food Product Development (pp. 262–280). Elsevier. https://doi.org/10.1533/9781845693381.2.262
- Köster, E. P., & Mojet, J. (2007b). Theories of food choice development. In Understanding Consumers of Food Products (pp. 93–124). Elsevier. https://doi.org/10.1533/9781845692506.1.93
- Lawless, H. T., & Heymann, H. (2010). *Sensory Evaluation of Food: Principles and Practices*. Springer New York. https://doi.org/10.1007/978-1-4419-6488-5
- Lee, W. E., Takahashi, K., & Pruitt, J. S. (1992). Temporal aspects of the oral processing of viscous solutions. *Food Technology*, *46*(11), 106–112.
- Leite, P. B., Bispo, E. D. S., & Santana, L. R. R. (2013). Sensory profiles of chocolates produced from cocoa cultivars resistant to Moniliophtora Perniciosa. *Revista Brasileira de Fruticultura*, 35(2), 594–602. https://doi.org/10.1590/S0100-29452013000200031
- Lessa, O. A., Reis, N. S., Leite, S. G. F., Gutarra, M. L. E., Souza, A. O., Gualberto, S. A., de Oliveira, J. R., Aguiar-Oliveira, E., & Franco, M. (2018). Effect of the solid state fermentation of cocoa shell on the secondary metabolites, antioxidant activity, and fatty acids. *Food Science and Biotechnology*, 27(1), 107–113. https://doi.org/10.1007/s10068-017-0196-x
- Livermore, A., & Laing, D. G. (1998). The influence of chemical complexity on the perception of multicomponent odor mixtures. *Perception & Psychophysics*, 60(4), 650–661. https://doi.org/10.3758/BF03206052
- Malundo, T. M. M., Shewfelt, R. L., Ware, G. O., & Baldwin, A. E. A. (2001). An Alternative Method for Relating Consumer and Descriptive Data Used To Identify Critical Flavor Properties Of Mango (Mangifera Indica L.). *Journal of Sensory Studies*, 16(2), 199–214. https://doi.org/10.1111/j.1745-459X.2001.tb00296.x
- Martínez, R., Torres, P., Meneses, M. A., Figueroa, J. G., Pérez-Álvarez, J. A., & Viuda-Martos, M. (2012). Chemical, technological and in vitro antioxidant properties of cocoa (Theobroma cacao L.) co-products. *Food Research International*, 49(1), 39–45. https://doi.org/10.1016/j.foodres.2012.08.005
- Martínez-Cervera, S., Salvador, A., Muguerza, B., Moulay, L., & Fiszman, S. M. (2011). Cocoa fibre and its application as a fat replacer in chocolate muffins. *LWT Food Science and Technology*, *44*(3), 729–736. https://doi.org/10.1016/j.lwt.2010.06.035
- Mellinas, A. C., Jiménez, A., & Garrigós, M. C. (2020). Optimization of microwave-assisted extraction of cocoa bean shell waste and evaluation of its antioxidant, physicochemical and functional properties. *LWT*, 127, 109361. https://doi.org/10.1016/j.lwt.2020.109361
- Minkina, V. G., Shabunya, S. I., Kalinin, V. I., Martynenko, V. V., & Smirnova, A. L. (2012). Stability of alkaline aqueous solutions of sodium borohydride. *International Journal of Hydrogen Energy*, 37(4), 3313–3318. https://doi.org/10.1016/j.ijhydene.2011.10.068

- Nsor-Atindana, J., Zhong, F., Mothibe, K. J., Bangoura, M. L., & Lagnika, C. (2012). Quantification of Total Polyphenolic Content and Antimicrobial Activity of Cocoa (Theobroma cacao L.) Bean Shells. *Pakistan Journal of Nutrition*, *11*(7), 672–677. https://doi.org/10.3923/pjn.2012.672.677
- Okiyama, D. C. G., Navarro, S. L. B., & Rodrigues, C. E. C. (2017). Cocoa shell and its compounds: Applications in the food industry. *Trends in Food Science & Technology*, 63, 103–112. https://doi.org/10.1016/j.tifs.2017.03.007
- Palczak, J., Blumenthal, D., Rogeaux, M., & Delarue, J. (2019). Sensory complexity and its influence on hedonic responses: A systematic review of applications in food and beverages. *Food Quality and Preference*, 71, 66–75. https://doi.org/10.1016/j.foodqual.2018.06.002
- Panak Balentić, J., Ačkar, D., Jokić, S., Jozinović, A., Babić, J., Miličević, B., Šubarić, D., & Pavlović, N. (2018). Cocoa Shell: A By-Product with Great Potential for Wide Application. *Molecules*, 23(6), 1404. https://doi.org/10.3390/molecules23061404
- Pangborn, R. M., Trabue, I. M., & Szczesniak, A. S. (1973). Effect of hydrocolooids on oral viscosity and basic taste intensities. *Journal of Texture Studies*, 4(2), 224–241. https://doi.org/10.1111/j.1745-4603.1973.tb00666.x
- Pu, Y., Zou, Q., Liu, L., Han, Z., Wang, X., Wang, Q., & Chen, S. (2012). Clinical dextran purified by fractional ultrafiltration coupled with water washing. *Carbohydrate Polymers*, 87(2), 1257–1260. https://doi.org/10.1016/j.carbpol.2011.09.006
- Redgwell, R. J., Trovato, V., Merinat, S., Curti, D., Hediger, S., & Manez, A. (2003). Dietary fibre in cocoa shell: Characterisation of component polysaccharides. *Food Chemistry*, 81(1), 103–112. https://doi.org/10.1016/S0308-8146(02)00385-0
- Resende, M., Geraldes, V., Catarino, S., & Norberta de Pinho, M. (2012). Optimization of Ultrafiltration / Diafiltration for Wine Polysaccharides Separation. Conference: 35th World Congress of Vine and Wine.
- Rojo-Poveda, O., Barbosa-Pereira, L., Mateus-Reguengo, L., Bertolino, M., Stévigny, C., & Zeppa, G. (2019). Effects of Particle Size and Extraction Methods on Cocoa Bean Shell Functional Beverage. *Nutrients*, 11(4), 867. https://doi.org/10.3390/nu11040867
- Rojo-Poveda, O., Barbosa-Pereira, L., Zeppa, G., & Stévigny, C. (2020). Cocoa Bean Shell—A By-Product with Nutritional Properties and Biofunctional Potential. *Nutrients*, *12*(4), 1123. https://doi.org/10.3390/nu12041123
- Spadoni Andreani, E., Karboune, S., & Liu, L. (2021). Extraction and characterization of cell wall polysaccharides from cranberry (Vaccinium macrocarpon var. Stevens) pomace. *Carbohydrate Polymers*, 267, 118212. https://doi.org/10.1016/j.carbpol.2021.118212
- Utami, R., Armunanto, R., Rahardjo, S., & Supriyanto. (2016). Effects of Cocoa Bean (Theobroma cacao L.) Fermentation on Phenolic Content, Antioxidant Activity and Functional Group of Cocoa Bean Shell. *Pakistan Journal of Nutrition*, *15*(10), 948–953. https://doi.org/10.3923/pjn.2016.948.953
- Waglay, A., Achouri, A., Karboune, S., Zareifard, M. R., & L'Hocine, L. (2019). Pilot plant extraction of potato proteins and their structural and functional properties. *LWT*, 113, 108275. https://doi.org/10.1016/j.lwt.2019.108275
- Waglay, A., & Karboune, S. (2020). Predictive Consumer Acceptance Models and Quality Attributes for Cookies Enriched with Potato Protein Isolate and Concentrate. *Food and Bioprocess Technology*, 13(9), 1645–1660. https://doi.org/10.1007/s11947-020-02508-8

- Younes, A., Li, M., & Karboune, S. (2022). Cocoa bean shells: A review into the chemical profile, the bioactivity and the biotransformation to enhance their potential applications in foods. *Critical Reviews in Food Science and Nutrition*, 1–25. https://doi.org/10.1080/10408398.2022.2065659
- Ziegler, G. R., Mongia, G., & Hollender, R. (2001). The role of particle size distribution of suspended solids in defining the sensory properties of milk chocolate. *International Journal of Food Properties*, 4(2), 353–370. https://doi.org/10.1081/JFP-100105199

CHAPTER VI. GENERAL CONCLUSION AND FUTURE WORK

This research focuses primarily on the isolation and characterization of cocoa bean shell oligoand polysaccharides, respectively. These cell wall components are of high interest for their valorization as a potential functional food ingredient. Insight pertaining to the nutritional composition of cocoa bean shells, from two varieties (NAC and CCN) was explored. Cocoa bean shells were found to be an abundant source of carbohydrates, specifically dietary fiber, protein, ash, and polyphenols. Their chemical profile showed alterations upon various degrees of processing, namely fermentation, drying and roasting. This was primarily attributed to the subjection of fluctuating temperatures throughout the process. The isolation of the cell wall material was evaluated according to their total sugar composition, monosaccharide profile, and molecular weight distribution. The cell wall material extracts were found to contain mostly pectic polysaccharides and smaller amounts of hemicellulose. Pectic polysaccharides in the extract were characterized by the presence of uronic acid, galactose, arabinose, and rhamnose.

Diluted and concentrated alkali treatments, involving potassium hydroxide, were investigated for the isolation of polysaccharides from cell wall material of cocoa bean shells from both varieties (NAC and CCN). Both alkali extracts were successful in the recovery of polysaccharide predominantly enriched with pectic polysaccharide regions, possibly indicating the presence of rhamnogalacturonan and arabinogalactan. Increasing the alkali concentration resulted in a shift in the recovery from high to low molecular weight fractions, due to a higher degree of defragmentation. The more concentrated extracts were found to contain four main molecular weight populations, compared to the three main molecular weight fractions found in the diluted alkali extracts.

Enzymatic hydrolysis of cocoa bean shell cell wall polysaccharides using pure enzymes and five multi-enzymatic products was investigated for the generation of oligosaccharides. The multi-enzymatic products expressed varied levels of polysaccharide-degrading activities. Viscozyme® L and Depol[™] 670L attained the highest hydrolysis extents for NAC and CCN cocoa bean shell varieties, while also generating a diverse complex of oligosaccharides with degree of polymerization between 3 and 10. A majority of the oligosaccharides (generated by multi-enzymatic biocatalysts had a degree of polymerization between 2 and 5. To explore their potential

functional properties, the prebiotic activities were assessed using cell wall polysaccharides (diluted alkali extracts) and oligosaccharides bio-generated by Viscozyme® L and DepolTM 670L, respectively. For the promotion of the growth of selected probiotic strains, *Lactobacillus rhamnosus GG* and *Bifidobacterium longum*, cell wall polysaccharides and enzymatically generated oligosaccharides performed similarly or more than inulin and rhamnogalacturonan based on their prebiotic activity scores. The short chain fatty acid profiles were mostly characterized by high amounts of lactic and acetic acid. In addition, enzymatically generated oligosaccharides also contributed towards a significant production of butyric and propionic acids.

Feruloylation of enzymatically generated oligosaccharides and diluted alkali polysaccharides was achieved by the esterifying activity of free feruloyl esterases, from *Humicola insolens* and rumen microorganism, using a surfactant-less microemulsion made of n-hexane/butanone/water (49.4/44.6/6, % v/v/v). Feruloyl esterase, from rumen microorganism, contributed towards the highest feruloylation efficiencies for polysaccharides, while that from *H. insolens* contributed more towards a higher feruloylation extent for enzymatically generated oligosaccharides. The antioxidant activity of feruloylated polysaccharides were assessed to gain insight on their radical scavenging functionalities. Based on three antioxidant assays, DPPH, ORAC and ABTS, significantly notable increases in the scavenging and antioxidant activity of feruloylated polysaccharides were forms, i.e., non-feruloylated polysaccharides, as well as to ferulic and ascorbic acids.

The isolation of polysaccharides and the enzymatic generation of oligosaccharides were further scaled-up in a pilot plant facility. Comparing the extracts recovered at the laboratory-scale, those recovered in the pilot plant were similar in terms of their monosaccharide profile and total neutral sugar content, with the exception of a slightly higher recovery of hemicellulose. The molecular weight distributions for each of the oligo- and polysaccharide extracts showed four main populations. Taking advantage of the functional properties the extracts exhibited, the development of a chocolate-based product was investigated. Cocoa bean shell polysaccharide extracts produced select chocolate products which were evaluated sensorially to be more desirable than those without their enrichment. Sensory evaluation consisted of five attributes related to the intensity and likeness of the chocolate product, namely, chocolate flavor, sweetness, sourness, bitterness, and melting in mouth. The attributes were evaluated according to mean quantitative descriptive scores and mean liking scores on a 9-point hedonic scale. According to the statistical analysis, the chocolate flavor (intensity and likeness), bitterness (intensity and likeness), melting in mouth (intensity) were responsible for the significant group differences prior to enrichment. Overall, higher likeness responses were obtained with enrichment, specifically in terms of the melting in mouth attribute. The relationships between the consumers liking scores and the perceived complexity were assessed to understand the degree of complexity inflicted upon consumers. Enrichment of the chocolate products did not contribute to differences between the perceived complexity of the sensory attributes. An understanding of the correlation between the intensity and likeness attributes indicated that polysaccharide enrichment contributed towards the development of a diverse and complex formulation matrix. All in all, consumers sought approval for the enriched chocolate products based on the overall acceptance and purchase intent responses.

Overall, this research contributes to the scientific knowledge providing a detailed description of the abundance and functional properties of cocoa bean shell oligo- and polysaccharides. It also offers a look at the feasibility of the generation of these extracts on an industrial scale, which may broaden the spectrum of their applications as a functional food ingredient.

References

- Acosta R.A., Sanabria J.R., & Nabarlatz D. 2018. Biomass from Colombian Agroindustrial Activities: Characterization and Potential for Oligosaccharides Production. *Chemical Engineering Transactions* 65 (June): 667–672. doi:10.3303/CET1865112.
- Adamafio, N.A. 2013. "Theobromine Toxicity and Remediation of Cocoa By-Products: An Overview." *Journal of Biological Sciences* 13 (7): 570–576. doi:10.3923/jbs.2013.570.576.
- Addison, D., & D. Aguilar. 2011. Diabetes and Cardiovascular Disease: The Potential Benefit of Incretin-Based Therapies. *Current Atherosclerosis Reports* 13 (2): 115–122. doi:10.1007/s11883-010-0153-0.
- Adeyina, A.O, D. Apata, A. Annongu, O. Olatunde, O. Alli, & K. M. Okupke. 2010.
 "Performance and Physiological Response of Weaner Rabbits Fed Hot Water Treated Cocoa Bean Shell Based Diet." *Research Journal of Veterinary Sciences* 5: 53–57.
- Adeyina, A.O., A.A. Annongu, & O. Awe. 2011. "Effects of Cocoa Bean Shell on the Performance, Blood Indices and Organ Characteristics of Cockerels." *Agrosearch* 10 (1– 2). doi:10.4314/agrosh.v10i1-2.69805.
- Agus, B.A.P, N.N. Mohamad, & N. Hussain. 2018. Composition of Unfermented, Unroasted, Roasted Cocoa Beans and Cocoa Shells from Peninsular Malaysia. *Journal of Food Measurement and Characterization* 12 (4): 2581–2589. doi:10.1007/s11694-018-9875-4.
- Ahmad, F., W.M.A.W Daud, M.A. Ahmad, & R. Radzi. 2012. "Cocoa (Theobroma Cacao) Shell-Based Activated Carbon by CO2 Activation in Removing of Cationic Dye from Aqueous Solution: Kinetics and Equilibrium Studies." *Chemical Engineering Research* and Design 90 (10): 1480–1490. doi:10.1016/j.cherd.2012.01.017.
- Alemawor, F., V. P. Dzogbefia, E. O. K. Oddoye, & J. H. Oldham. 2009. Enzyme Cocktail for Enhancing Poultry Utilisation of Cocoa Pod Husk. *Scientific Research and Essays* 4 (6): 555–559.
- Allerdings, E., J. Ralph, P. Schatz, D. Gniechwitz, H. Steinhart, & M. Bunzel. 2005. Isolation and Structural Identification of Diarabinosyl 8--4-Dehydrodiferulate from Maize Bran Insoluble Fibre. *Phytochemistry* 66 (1): 113–124. doi:10.1016/j.phytochem.2004.10.026.
- Altin, G., M. Gültekin-Özgüven, & B. Ozcelik. 2018. "Liposomal Dispersion and Powder Systems for Delivery of Cocoa Hull Waste Phenolics via Ayran (Drinking Yoghurt): Comparative Studies on in-Vitro Bioaccessibility and Antioxidant Capacity." *Food Hydrocolloids* 81 (August): 364–370. doi:10.1016/j.foodhyd.2018.02.051.
- American Association of Cereal Chemists. 2001. *The Definition of Dietary Fiber*. W-2001-0222-01O. Cereal & Grains Association.
 - http://online.cerealsgrains.org/initiatives/definitions/Documents/DietaryFiber/DFDef.pdf.
- Anderson, J.W., P. Baird, R.H. Davis Jr, S. Ferreri, M. Knudtson, Ashraf Koraym, Valerie Waters, & Christine L Williams. 2009. Health Benefits of Dietary Fiber. *Nutrition Reviews* 67 (4): 188–205. doi:10.1111/j.1753-4887.2009.00189.x.
- Anvoh, K.Y.B., A. Zoro Bi, & D. Gnakri. 2009. "Production and Characterization of Juice from Mucilage of Cocoa Beans and Its Transformation into Marmalade." *Pakistan Journal of Nutrition* 8 (2): 129–133. doi:10.3923/pjn.2009.129.133.
- Aprotosoaie, A.C., S.V. Luca, & A. Miron. 2016. Flavor Chemistry of Cocoa and Cocoa Products-An Overview: Flavor Chemistry of Cocoa. *Comprehensive Reviews in Food Science and Food Safety* 15 (1): 73–91. doi:10.1111/1541-4337.12180.

- Aregheore, E.M. 2002. Chemical Evaluation and Digestibility of Cocoa (Theobroma Cacao) Byproducts Fed to Goats. *Tropical Animal Health and Production* 34 (4): 339–348. doi:10.1023/A:1015638903740.
- Aremu, C. Y., M. A. Agiang, & J. O. I. Ayatse. 1995. Nutrient and Antinutrient Profiles of Raw and Fermented Cocoa Beans. *Plant Foods for Human Nutrition* 48 (3): 217–223. doi:10.1007/BF01088443.
- Arlorio, M., J.D. Coïsson, F. Travaglia, F. Varsaldi, G. Miglio, G. Lombardi, & A. Martelli. 2005. Antioxidant and Biological Activity of Phenolic Pigments from Theobroma Cacao Hulls Extracted with Supercritical CO2. *Food Research International* 38 (8–9): 1009– 1014. doi:10.1016/j.foodres.2005.03.012.
- Aspinal, G. 1980. Chemistry of Cell Wall Polysaccharides. In *The Biochemistry of Plants: A Comprehensive Treatise. 3: Carbohydrates: Structure and Function. 1980*, edited by Paul K. Stumpf. New York: Acad. Pr.
- Awolu, O.O., & O.S. Oyetuji. 2015. "Optimization of Bioethanol Production from Cocoa (Theobroma Cacao) Bean Shell." *International Journal of Current Microbiology and Applied Sciences* 4 (4): 506–514.
- Bacic, A, P.J. Harris, & B.A. Stone. 1988. Chapter 8 Structure and Function of Plant Cell Walls. In *The Biochemistry of Plants: A Comprehensive Treatise*. 14: Carbohydrates, edited by P. K. Stumpf, 14:297–358. Orlando, Fla.: Acad. Pr.
- Ballard, T.S., P. Mallikarjunan, K. Zhou, & S. O'Keefe. 2010. Microwave-Assisted Extraction of Phenolic Antioxidant Compounds from Peanut Skins. *Food Chemistry* 120 (4): 1185– 1192. doi:10.1016/j.foodchem.2009.11.063.
- Ballesteros, L.F., J.A. Teixeira, & S.I. Mussatto. 2014. Chemical, Functional, and Structural Properties of Spent Coffee Grounds and Coffee Silverskin. *Food and Bioprocess Technology* 7 (12): 3493–3503. doi:10.1007/s11947-014-1349-z.
- Bamba, Y., N. Ouattara, S. Ouattara, A. Ouatarra, and G. Gourene. 2015. "Effect of Diets Containing Cocoa Bean Shell and Coconut Oil Cake on the Growth of Oreochromis Niloticus (LINNE, 1758) in Pond." *International Journal of Biological and Chemical Sciences* 8 (4): 1368. doi:10.4314/ijbcs.v8i4.2.
- Barbosa-Pereira, L., A. Guglielmetti, & G. Zeppa. 2018. Pulsed Electric Field Assisted Extraction of Bioactive Compounds from Cocoa Bean Shell and Coffee Silverskin. *Food* and Bioprocess Technology 11 (4): 818–835. doi:10.1007/s11947-017-2045-6.
- Barišić, V., A. Jozinović, I. Flanjak, D. Šubarić, J. Babić, Borislav Miličević, Stela Jokić, Ivanka Grgić, & Đurđica Ačkar. 2019. Effect of Addition of Fibres and Polyphenols on Properties of Chocolate – A Review. *Food Reviews International*, December, 1–19. doi:10.1080/87559129.2019.1701008.
- Bartolomé, B., C.B. Faulds, & G. Williamson. 1997. Enzymic Release of Ferulic Acid from Barley Spent Grain. *Journal of Cereal Science* 25 (3): 285–288. doi:10.1006/jcrs.1996.0091.
- Bartolome, B, & C Gomez-Cordoves. 1999. Barley Spent Grain: Release of Hydroxycinnamic Acids (Ferulic and p-Coumaric Acids) by Commercial Enzyme Preparations. *Journal of the Science of Food and Agriculture* 79 (3): 435—439. doi:10.1002/(sici)1097-0010(19990301)79:3<435::aid-jsfa272>3.0.co;2-s.
- Bartolomé, B., C. Gómez-Cordovés, A.I. Sancho, N. Díez, P. Ferreira, J. Soliveri, & J.L. Copa-Patiño. 2003. Growth and Release of Hydroxycinnamic Acids from Brewer's Spent Grain

by Streptomyces Avermitilis CECT 3339. *Enzyme and Microbial Technology* 32 (1): 140–144. doi:10.1016/S0141-0229(02)00277-6.

- Bernaert, H., & I.D. Ruysscher. 2016. Process of Producing Cocoa Shell Powder.
- Biliaderis, C. G., & M.S. Izydorczyk, eds. 2007. *Functional Food Carbohydrates*. Functional Foods and Nutraceuticals Series. Boca Raton: CRC Press.
- Bitzer, Z. T., S.L. Glisan, M.R. Dorenkott, K.M. Goodrich, L. Ye, S.F. O'Keefe, J.D. Lambert, & A.P. Neilson. 2015. Cocoa Procyanidins with Different Degrees of Polymerization Possess Distinct Activities in Models of Colonic Inflammation. *The Journal of Nutritional Biochemistry* 26 (8): 827–831. doi:10.1016/j.jnutbio.2015.02.007.
- Bohn, T. 2014. Dietary Factors Affecting Polyphenol Bioavailability. *Nutrition Reviews* 72 (7): 429–452. doi:10.1111/nure.12114.
- Bonvehí, J.S., & F.V. Coll. 1999. "Protein Quality Assessment in Cocoa Husk." Food Research International 32 (3): 201–208. doi:10.1016/S0963-9969(99)00088-5.
- Borrelli, R.C., F. Esposito, A. Napolitano, A. Ritieni, & V. Fogliano. 2004. "Characterization of a New Potential Functional Ingredient: Coffee Silverskin." *Journal of Agricultural and Food Chemistry* 52 (5): 1338–1343. doi:10.1021/jf034974x.
- Boušová, I., J. Martin, L. Jahodář, J. Dušek, V. Palička, & J. Dršata. 2005. Evaluation of in Vitro Effects of Natural Substances of Plant Origin Using a Model of Protein Glycoxidation. *Journal of Pharmaceutical and Biomedical Analysis* 37 (5): 957–962. doi:10.1016/j.jpba.2004.10.014.
- Budryn, G., E. Nebesny, D. Rachwal-Rosiak, & J. Oracz. 2013. Fatty Acids, Essential Amino Acids, and Chlorogenic Acids Profiles, in Vitro Protein Digestibility and Antioxidant Activity of Food Products Containing Green Coffee Extract. *International Food Research Journal* 25 (5): 2133–2144.
- Camu, N., T. D. Winter, S.K. Addo, J.S. Takrama, H. Bernaert, & L.D. Vuyst. 2008. Fermentation of Cocoa Beans: Influence of Microbial Activities and Polyphenol Concentrations on the Flavour of Chocolate: Heap Fermentation versus Chocolate Production. *Journal of the Science of Food and Agriculture* 88 (13): 2288–2297. doi:10.1002/jsfa.3349.
- Cantril, R. 2008. Phytosterols, Phytostanols and Their Esters: Chemical and Technical Assessment. In .

http://www.fao.org/fileadmin/templates/agns/pdf/jecfa/cta/69/Phytosterols.pdf.

- Carbonaro, M., G. Grant, M. Cappelloni, & A.Pusztai. 2000. Perspectives into Factors Limiting in Vivo Digestion of Legume Proteins: Antinutritional Compounds or Storage Proteins? *Journal of Agricultural and Food Chemistry* 48 (3): 742–749. doi:10.1021/jf991005m.
- Charlton, A. J., N.J. Baxter, M.L. Khan, A.J.G. Moir, E. Haslam, A.P. Davies, & M.P. Williamson. 2002. Polyphenol/Peptide Binding and Precipitation. *Journal of Agricultural* and Food Chemistry 50 (6): 1593–1601. doi:10.1021/jf010897z.
- Chen, H. 2015. Lignocellulose Biorefinery Engineering. Boston, MA: Elsevier.
- Chen, H., & X. Yan. 2005. Antioxidant Activities of Agaro-Oligosaccharides with Different Degrees of Polymerization in Cell-Based System. *Biochimica et Biophysica Acta (BBA) -General Subjects* 1722 (1): 103–111. doi:10.1016/j.bbagen.2004.11.016.
- Chen, X., Q. Liu, X. Jiang, & F. Zeng. 2005. Microwave-Assisted Extraction of Polysaccharides from Solanum Nigrum. *Journal of Central South University of Technology* 12 (5): 556– 560. doi:10.1007/s11771-005-0122-x.

- Chigorimbo-Murefu, N.T.L., S. Riva, & S.G. Burton. 2009. "Lipase-Catalysed Synthesis of Esters of Ferulic Acid with Natural Compounds and Evaluation of Their Antioxidant Properties." *Journal of Molecular Catalysis B: Enzymatic* 56 (4): 277–282. doi:10.1016/j.molcatb.2008.05.017.
- Choi, J., N. Kim, H.Y. Choi, & Y.S. Han. 2019. Effect of Cacao Bean Husk Powder on the Quality Properties of Pork Sausages. *Food Science of Animal Resources* 39 (5): 742–755. doi:10.5851/kosfa.2019.e62.
- Cirkovic Velickovic, T.D., & D.J. Stanic-Vucinic. 2018. The Role of Dietary Phenolic Compounds in Protein Digestion and Processing Technologies to Improve Their Antinutritive Properties: Phenolic Compounds in Protein Digestion. *Comprehensive Reviews in Food Science and Food Safety* 17 (1): 82–103. doi:10.1111/1541-4337.12320.
- Colivet, J., A.L. Oliveira, & R.A. Carvalho. 2016. "Influence of the Bed Height on the Kinetics of Watermelon Seed Oil Extraction with Pressurized Ethanol." *Separation and Purification Technology* 169 (September): 187–195. doi:10.1016/j.seppur.2016.06.020.
- Colquhoun, I.J., M. Ralet, J. Thibault, C.B. Faulds, & G. Williamson. 1994. Structure Identification of Feruloylated Oligosaccharides from Sugar-Beet Pulp by NMR Spectroscopy. *Carbohydrate Research* 263 (2): 243–256. doi:10.1016/0008-6215(94)00176-6.
- Costa, A.S.G., R.C. Alves, A.F. Vinha, E. Costa, C.S.G. Costa, M.A. Nunes, A.A. Almeida, A. Santos-Silva, & M.B.P.P. Oliveira. 2018. "Nutritional, Chemical and Antioxidant/pro-Oxidant Profiles of Silverskin, a Coffee Roasting by-Product." *Food Chemistry* 267 (November): 28–35. doi:10.1016/j.foodchem.2017.03.106.
- Couto, J., S. Karboune, & R. Mathew. 2010. Regioselective Synthesis of Feruloylated Glycosides Using the Feruloyl Esterases Expressed in Selected Commercial Multi-Enzymatic Preparations as Biocatalysts. *Biocatalysis and Biotransformation* 28 (4): 235– 244. doi:10.3109/10242422.2010.493209.
- Couto, J., R. St-Louis, & S. Karboune. 2011. Optimization of Feruloyl Esterase-Catalyzed Synthesis of Feruloylated Oligosaccharides by Response Surface Methodology. *Journal* of Molecular Catalysis B: Enzymatic 73 (1–4): 53–62. doi:10.1016/j.molcatb.2011.07.016.
- Davis, E.J., E. Spadoni Andreani, & S. Karboune. 2021. "Production of Extracts Composed of Pectic Oligo/Polysaccharides and Polyphenolic Compounds from Cranberry Pomace by Microwave-Assisted Extraction Process." *Food and Bioprocess Technology* 14 (4): 634– 649. doi:10.1007/s11947-021-02593-3.
- de Brito, E.S., N.H. Pezoa Garcia, M.I. Gallao, A.L. Cortelazzo, P.S. Fevereiro, & M.R. Braga. 2001. Structural and Chemical Changes in Cocoa (Theobroma Cacao L) during Fermentation Drying and Roasting. *Journal of the Science of Food and Agriculture* 81 (2): 281–288. doi:10.1002/1097-0010(20010115)81:2<281::aid-jsfa808>3.0.co;2-b.
- de Lima, R.B., T.B. dos Santos, L.G.E Vieira, M.L.L.Ferrarese, O. Ferrarese-Filho, L. Donatti, M.R.T. Boeger, & C.L.O. Petkowicz. 2014. Salt Stress Alters the Cell Wall Polysaccharides and Anatomy of Coffee (Coffea Arabica L.) Leaf Cells. *Carbohydrate Polymers* 112 (November): 686–694. doi:10.1016/j.carbpol.2014.06.042.
- de Oliveira, D.M., V. Hugo Salvador, T.R. Mota, A. Finger-Teixeira, R. de Almeida, Douglas A.A. Paixão, Amanda P. De Souza, et al. 2016. Feruloyl Esterase from Aspergillus Clavatus Improves Xylan Hydrolysis of Sugarcane Bagasse. *AIMS Bioengineering* 4 (1): 1–11. doi:10.3934/bioeng.2017.1.1.

- Delgado-Ospina, J., C.D. Di Mattia, A. Paparella, D. Mastrocola, M. Martuscelli, & C. Chaves-Lopez. 2020. "Effect of Fermentation, Drying and Roasting on Biogenic Amines and Other Biocompounds in Colombian Criollo Cocoa Beans and Shells." *Foods* 9 (4): 520. doi:10.3390/foods9040520.
- Diaz, V.M., J.J. Nakayo, & E.B. Alfaro. 2018. "Use of Coffee Grind with Cocoa Shell as the Basis for a Filter to Reduce Lead from Contaminated Water from a River, Peru." *Journal* of Energy & Environmental Sciences 2 (2): 24. doi:10.32829/eesj.v2i2.68.
- Domínguez-Pérez, L.A., L.M. Beltrán-Barrientos, A.F. González-Córdova, A. Hernández-Mendoza, & B. Vallejo-Cordoba. 2020. "Artisanal Cocoa Bean Fermentation: From Cocoa Bean Proteins to Bioactive Peptides with Potential Health Benefits." *Journal of Functional Foods* 73 (October): 104134. doi:10.1016/j.jff.2020.104134.
- Domínguez-Pérez, Leydy A., Lilia M. Beltrán-Barrientos, Aarón F. González-Córdova, Adrián Hernández-Mendoza, & Belinda Vallejo-Cordoba. 2020. "Artisanal Cocoa Bean Fermentation: From Cocoa Bean Proteins to Bioactive Peptides with Potential Health Benefits." *Journal of Functional Foods* 73 (October): 104134. doi:10.1016/j.jff.2020.104134.
- Dominiak, M., K.M. Søndergaard, J. Wichmann, S. Vidal-Melgosa, W.G.T. Willats, A.S. Meyer, & J.D. Mikkelsen. 2014. Application of Enzymes for Efficient Extraction, Modification, and Development of Functional Properties of Lime Pectin. *Food Hydrocolloids* 40 (October): 273–282. doi:10.1016/j.foodhyd.2014.03.009.
- Eggum, B. O., B. Pedersen, and I. Jacobsen. 1983. The Influence of Dietary Tea, Coffee and Cocoa on Protein and Energy Utilization of Soya-Bean Meal and Barley in Rats. *British Journal of Nutrition* 50 (2): 197–205. doi:10.1079/BJN19830089.
- El Hag, I.M.N, I.A.M Ahmed, S.O. Ahmed, M.M. Eltayeb, & E.B. Elfadil. 2013. Effect of Processing Methods on Antinutritional Factors, Protein Digestibility and Minerals Extractability of Winter Sorghum Cultivars. *Australian Journal of Basic and Applied Sciences* 7 (12): 2299–237.
- El-Saied, H.M., M. K. Morsi, & M. M. A. Amer. 1981. Composition of Cocoa Shell Fat as Related to Cocoa Butter. *Zeitschrift Für Ernährungswissenschaft* 20 (2): 145–151. doi:10.1007/BF02021260.
- Emiola, I.A., O.O. Ojebiyi, & T.O. Akande. 2011. "Performance and Organ Weights of Laying Hens Fed Diets Containing Graded Levels of Sun-Dried Cocoa Bean Shell (CBS)." *International Journal of Poultry Science* 10 (12): 987–990. doi:10.3923/ijps.2011.987.990.
- Eskilsson, C. S., & E. Björklund. 2000. Analytical-Scale Microwave-Assisted Extraction. *Journal of Chromatography A* 902 (1): 227–250. doi:10.1016/S0021-9673(00)00921-3.
- Fakhlaei, R., Rozzamri A., and N. Hussain. 2019. Composition, Color and Antioxidant Properties of Cocoa Shell at Different Roasting Temperatures. *Food Research* 4 (3): 585– 593. doi:10.26656/fr.2017.4(3).251.
- Fang, H., Y. Chen, H. Chen, S. Lin, & Y. Fang. 2012. Immunomodulatory Effects of Feruloylated Oligosaccharides from Rice Bran. *Food Chemistry* 134 (2): 836–840. doi:10.1016/j.foodchem.2012.02.190.
- Fernández-Ponce, M.T., B.R. Parjikolaei, H.N. Lari, L. Casas, C. Mantell, & E.J. Martínez de la Ossa. 2016. "Pilot-Plant Scale Extraction of Phenolic Compounds from Mango Leaves Using Different Green Techniques: Kinetic and Scale up Study." *Chemical Engineering Journal* 299 (September): 420–430. doi:10.1016/j.cej.2016.04.046.

- Fioresi, F., J. Vieillard, R. Bargougui, N. Bouazizi, P.N. Fotsing, E.D. Woumfo, N. Brun, N. Mofaddel, & F. Le Derf. 2017. "Chemical Modification of the Cocoa Shell Surface Using Diazonium Salts." *Journal of Colloid and Interface Science* 494 (May): 92–97. doi:10.1016/j.jcis.2017.01.069.
- Flint, A., A. Raben, A. Astrup, & J.J. Holst. 1998. Glucagon-like Peptide 1 Promotes Satiety and Suppresses Energy Intake in Humans. *Journal of Clinical Investigation* 101 (3): 515–520. doi:10.1172/JCI990.
- Fontes, C. M. A., R. B. Silva, & P. R. L. Lima. 2019. "Characterization and Effect of Using Bottom and Fly Ashes from Co-Combustion of Cocoa Waste as Mineral Addition in Concrete." *Waste and Biomass Valorization* 10 (1): 223–233. doi:10.1007/s12649-017-0031-x.
- Fry, S.C. 1982. Phenolic Components of the Primary Cell Wall. Feruloylated Disaccharides of d-Galactose and l-Arabinose from Spinach Polysaccharide. *Biochemical Journal* 203 (2): 493–504. doi:10.1042/bj2030493.
- Furusawa, M., Y. Narita, K. Iwai, T. Fukunaga, & O. Nakagiri. 2011. "Inhibitory Effect of a Hot Water Extract of Coffee 'Silverskin' on Hyaluronidase." *Bioscience, Biotechnology, and Biochemistry* 75 (6): 1205–1207. doi:10.1271/bbb.110106.
- Gálvez Ranilla, L., Christopher, A., Sarkar, D., Shetty, K., Chirinos, R., & Campos, D. (2017).
 Phenolic Composition and Evaluation of the Antimicrobial Activity of Free and Bound
 Phenolic Fractions from a Peruvian Purple Corn (Zea mays L.) Accession. *Journal of Food Science*, 82(12), 2968–2976. https://doi.org/10.1111/1750-3841.13973
- Gartaula, G., S. Dhital, O. Deshmukh, G. Netzel, & M.J. Gidley. 2019. Rheological Characterisation of Cell Walls from Wheat Flour and Endosperm: Effects of Diferulate Crosslink Hydrolysis. *Food Hydrocolloids* 88 (March): 265–271. doi:10.1016/j.foodhyd.2018.10.004.
- Gemechu, F.G. 2020. Embracing Nutritional Qualities, Biological Activities and Technological Properties of Coffee Byproducts in Functional Food Formulation. *Trends in Food Science* & *Technology* 104 (October): 235–261. doi:10.1016/j.tifs.2020.08.005.
- Getachew, A.T., Y.J. Cho, & B.S. Chun. 2018. Effect of Pretreatments on Isolation of Bioactive Polysaccharides from Spent Coffee Grounds Using Subcritical Water. *International Journal of Biological Macromolecules* 109 (April): 711–719. doi:10.1016/j.ijbiomac.2017.12.120.
- Giami, S.Y., M.N. Adindu, A.D. Hart, & E.O. Denenu. 2001. Effect of Heat Processing on in Vitro Protein Digestibility and Some Chemical Properties of African Breadfruit (Treculia Africana Decne) Seeds. *Plant Foods for Human Nutrition* 56 (2): 117–126. doi:10.1023/A:1011181412808.
- Gómez Hoyos, C., P. Mazo Márquez, L. Penagos Vélez, A. Serpa Guerra, A. Eceiza, L. Urbina,
 J. Velásquez-Cock, P. Gañán Rojo, L. Vélez Acosta, & R. Zuluaga. 2020. Cocoa Shell:
 An Industrial by-Product for the Preparation of Suspensions of Holocellulose Nanofibers
 and Fat. *Cellulose*, May. doi:10.1007/s10570-020-03222-6.
- Gottschalk, L.M.F., R.A. Oliveira, & E.P.S Bon. 2010. Cellulases, Xylanases, β-Glucosidase and Ferulic Acid Esterase Produced by Trichoderma and Aspergillus Act Synergistically in the Hydrolysis of Sugarcane Bagasse. *Biochemical Engineering Journal* 51 (1–2): 72–78. doi:10.1016/j.bej.2010.05.003.

- Goya, L., M. Martín, B. Sarriá, S. Ramos, R. Mateos, & L. Bravo. 2016. "Effect of Cocoa and Its Flavonoids on Biomarkers of Inflammation: Studies of Cell Culture, Animals and Humans." *Nutrients* 8 (4): 212. doi:10.3390/nu8040212.
- Grillo, G., L. Boffa, A. Binello, S. Mantegna, G. Cravotto, F. Chemat, T. Dizhbite, L. Lauberte, & G. Telysheva. 2019. Cocoa Bean Shell Waste Valorisation; Extraction from Lab to Pilot-Scale Cavitational Reactors. *Food Research International* 115 (January): 200–208. doi:10.1016/j.foodres.2018.08.057.
- Han, X., T. Shen, & H. Lou. 2007. Dietary Polyphenols and Their Biological Significance. International Journal of Molecular Sciences 8 (9): 950–988. doi:10.3390/i8090950.
- Handojo, L., Cherilisa, & A. Indarto. 2020. "Cocoa Bean Skin Waste as Potential Raw Material for Liquid Smoke Production." *Environmental Technology* 41 (8): 1044–1053. doi:10.1080/09593330.2018.1520306.
- Handojo, L., H. Triharyogi, & A. Indarto. 2019. "Cocoa Bean Shell Waste as Potential Raw Material for Dietary Fiber Powder." *International Journal of Recycling of Organic Waste in Agriculture* 8 (S1): 485–491. doi:10.1007/s40093-019-0271-9.
- Harris, P. J., & B. G. Smith. 2006. Plant Cell Walls and Cell-Wall Polysaccharides: Structures, Properties and Uses in Food Products. *International Journal of Food Science and Technology* 41 (s2): 129–143. doi:10.1111/j.1365-2621.2006.01470.x.
- Hayat, K., S. Hussain, S. Abbas, U. Farooq, B. Ding, S. Xia, C. Jia, X. Zhang, & W. Xia. 2009. Optimized Microwave-Assisted Extraction of Phenolic Acids from Citrus Mandarin Peels and Evaluation of Antioxidant Activity in Vitro. *Separation and Purification Technology* 70 (1): 63–70. doi:10.1016/j.seppur.2009.08.012.
- He, F., S. Zhang, & X. Liu. 2015. "Immobilization of Feruloyl Esterases on Magnetic Nanoparticles and Its Potential in Production of Ferulic Acid." *Journal of Bioscience and Bioengineering* 120 (3): 330–334. doi:10.1016/j.jbiosc.2015.01.006.
- Hernández-Hernández, C., A. Morales-Sillero, M.A. Fernández-Prior, J. Fernández-Bolaños, M. de la Paz Aguilera-Herrera, & G. Rodríguez-Gutiérrez. 2019. "Extra Virgin Olive Oil Jam Enriched with Cocoa Bean Husk Extract Rich in Theobromine and Phenols." *LWT* 111 (August): 278–283. doi:10.1016/j.lwt.2019.05.027.
- Hirabayashi, T., H. Ochiai, S. Sakai, K. Nakajima, & K. Terasawa. 1995. Inhibitory Effect of Ferulic Acid and Isoferulic Acid on Murine Interleukin-8 Production in Response to Influenza Virus Infections in Vitro and in Vivo. *Planta Medica* 61 (03): 221–226. doi:10.1055/s-2006-958060.
- Huang, R.Q., & S.Y. Ou. 2012. "Effects of Oxalic Acid-Autoclaving Treatment and Surfactants on Preparation Feruloylatedated Oligosaccharides from Maize Bran." *Journal of Chinese Institute of Food Science and Technology*, 05.
- International Cocoa Organization. 2017. *Cocoa Market Review August 2017*. Cote d'Ivoire: International Cocoa Organization (ICCO).
- Iriondo-DeHond, A., M.B. Rios, T. Herrera, A. Rodriguez-Bertos, F. Nuñez, M.I. San Andres, S. Sanchez-Fortun, & M.D. del Castillo. 2019. "Coffee Silverskin Extract: Nutritional Value, Safety and Effect on Key Biological Functions." *Nutrients* 11 (11): 2693. doi:10.3390/nu11112693.
- Ishii, T. 1997. Structure and Functions of Feruloylated Polysaccharides. *Plant Science* 127 (2): 111–127. doi:10.1016/S0168-9452(97)00130-1.
- Ismail, A., & C.L. Yee. 2006. "Antioxidative Effects of Extracts of Cocoa Shell, Roselle Seeds and a Combination of Both Extracts on the Susceptibility of Cooked Beef to Lipid

Oxidation." *Journal of Food Technology* 4 (1): 10–15.

http://docsdrive.com/pdfs/medwelljournals/jftech/2006/10-15.pdf.

- Jacques, R.A., C. Dariva, J.V. de Oliveira, & E.B. Caramão. 2008. "Pressurized Liquid Extraction of Mate Tea Leaves." *Analytica Chimica Acta* 625 (1): 70–76. doi:10.1016/j.aca.2008.07.002.
- Jansman, A.J.M., A.A. Frohlich, & R. R. Marquardt. 1994. Production of Proline-Rich Proteins by the Parotid Glands of Rats Is Enhanced by Feeding Diets Containing Tannins from Faba Beans (Vicia Faba L.). *The Journal of Nutrition* 124 (2): 249–258. doi:10.1093/jn/124.2.249.
- Jiménez-Zamora, A., S. Pastoriza, & J.A. Rufián-Henares. 2015. "Revalorization of Coffee By-Products. Prebiotic, Antimicrobial and Antioxidant Properties." *LWT - Food Science and Technology* 61 (1): 12–18. doi:10.1016/j.lwt.2014.11.031.
- Jokić, S., N. Nastić, S. Vidović, I. Flanjak, K. Aladić, & J. Vladić. 2020. An Approach to Value Cocoa Bean By-Product Based on Subcritical Water Extraction and Spray Drying Using Different Carriers. *Sustainability* 12 (6): 2174. doi:10.3390/su12062174.
- Jozinović, A., J. Panak Balentić, D. Ačkar, J. Babić, B. Pajin, B. Miličević, S. Guberac, A. Vrdoljak, & D. Šubarić. 2019. "Cocoa Husk Application in the Enrichment of Extruded Snack Products." *Journal of Food Processing and Preservation* 43 (2): e13866. doi:10.1111/jfpp.13866.
- Kaderides, K., L. Papaoikonomou, M. Serafim, & A.M. Goula. 2019. Microwave-Assisted Extraction of Phenolics from Pomegranate Peels: Optimization, Kinetics, and Comparison with Ultrasounds Extraction. *Chemical Engineering and Processing -Process Intensification* 137 (March): 1–11. doi:10.1016/j.cep.2019.01.006.
- Karboune, S., & N. Khodaei. 2016. "Structures, Isolation and Health-Promoting Properties of Pectic Polysaccharides from Cell Wall-Rich Food by-Products: A Source of Functional Ingredients." *Current Opinion in Food Science* 8 (April): 50–55. doi:10.1016/j.cofs.2016.02.003.
- Katapodis, P., M. Vardakou, E. Kalogeris, D. Kekos, B.J. Macris, & P. Christakopoulos. 2003. Enzymic Production of a Feruloylated Oligosaccharide with Antioxidant Activity from Wheat Flour Arabinoxylan. *European Journal of Nutrition* 42 (1): 55–60. doi:10.1007/s00394-003-0400-z.
- Khodaei, N., & S. Karboune. 2018. "Optimization of Enzymatic Production of Prebiotic Galacto/Galacto(Arabino)-Oligosaccharides and Oligomers from Potato Rhamnogalacturonan I." *Carbohydrate Polymers* 181 (February): 1153–1159. doi:10.1016/j.carbpol.2017.11.011.
- Khodaei, N., S. Karboune, & V. Orsat. 2016. Microwave-Assisted Alkaline Extraction of Galactan-Rich Rhamnogalacturonan I from Potato Cell Wall by-Product. *Food Chemistry* 190 (January): 495–505. doi:10.1016/j.foodchem.2015.05.082.
- Kilpeläinen, P. O., S. S. Hautala, O. O. Byman, L. J. Tanner, R. I. Korpinen, M. K. Lillandt, A. V. Pranovich, V. H. Kitunen, S. M. Willför, & H. S. Ilvesniemi. 2014. Pressurized Hot Water Flow-through Extraction System Scale up from the Laboratory to the Pilot Scale. *Green Chem.* 16 (6): 3186–3194. doi:10.1039/C4GC00274A.
- Kim, H., & P. G. Keeney. 1984. (-)-Epicatechin Content in Fermented and Unfermented Cocoa Beans. *Journal of Food Science* 49 (4): 1090–1092. doi:10.1111/j.1365-2621.1984.tb10400.x.

- Knapp, A.W, & K.H. Coward. 1935. The Vitamin D Activity of Cacao Shell. I. The Effect of the Fermenting and Drying of Cacao on the Vitamin D Potency of Cacao Shell. II. The Origin of Vitamin D in Cacao Shell. 29: 2728–2735.
- Kondo, T., T. Ohshita, & T. Kyuma. 1992. Comparison of Characteristics of Soluble Lignins from Untreated and Ammonia-Treated Wheat Straw. *Animal Feed Science and Technology* 39 (3–4): 253–263. doi:10.1016/0377-8401(92)90045-8.
- Kumari, N., A. Grimbs, R.N. D'Souza, S.K. Verma, M. Corno, N. Kuhnert, & M.S. Ullrich. 2018. "Origin and Varietal Based Proteomic and Peptidomic Fingerprinting of Theobroma Cacao in Non-Fermented and Fermented Cocoa Beans." *Food Research International* 111 (September): 137–147. doi:10.1016/j.foodres.2018.05.010.
- Kylli, P., P. Nousiainen, P. Biely, J. Sipilä, M. Tenkanen, & M. Heinonen. 2008. Antioxidant Potential of Hydroxycinnamic Acid Glycoside Esters. *Journal of Agricultural and Food Chemistry* 56 (12): 4797–4805. doi:10.1021/jf800317v.
- Le Bourvellec, C., B. Bouchet, & C.M.G.C. Renard. 2005. Non-Covalent Interaction between Procyanidins and Apple Cell Wall Material. Part III: Study on Model Polysaccharides. *Biochimica et Biophysica Acta (BBA) - General Subjects* 1725 (1): 10–18. doi:10.1016/j.bbagen.2005.06.004.
- Le Bourvellec, C., & C.M.G.C. Renard. 2005. Non-Covalent Interaction between Procyanidins and Apple Cell Wall Material. Part II: Quantification and Impact of Cell Wall Drying. *Biochimica et Biophysica Acta (BBA) - General Subjects* 1725 (1): 1–9. doi:10.1016/j.bbagen.2005.06.003.
- Lecumberri, E., R. Mateos, M. Izquierdo-Pulido, P. Rupérez, L. Goya, & L. Bravo. 2007. Dietary Fibre Composition, Antioxidant Capacity and Physico-Chemical Properties of a Fibre-Rich Product from Cocoa (Theobroma Cacao L.). *Food Chemistry* 104 (3): 948– 954. doi:10.1016/j.foodchem.2006.12.054.
- Leong, T., M. Ashokkumar, & S. Kentish. 2011. "The Fundamentals of Power Ultrasound a Review." *Acoustics Australia* 39 (2): 43–52. https://minervaaccess.unimelb.edu.au/bitstream/handle/11343/123494/Leong_ASA.pdf?sequence=1&is Allowed=y.
- Lessa, O.A., N.S. Reis, S.G.F. Leite, M.L.E. Gutarra, A.O. Souza, S.A. Gualberto, J.R. Oliveira, E. Aguiar-Oliveira, & M. Franco. 2018. Effect of the Solid State Fermentation of Cocoa Shell on the Secondary Metabolites, Antioxidant Activity, and Fatty Acids. *Food Science* and Biotechnology 27 (1): 107–113. doi:10.1007/s10068-017-0196-x.
- Levigne, S., M.C. Ralet, & J.F. Thibault. 2002. "Characterisation of Pectins Extracted from Fresh Sugar Beet under Different Conditions Using an Experimental Design." *Carbohydrate Polymers* 49 (2): 145–153. doi:10.1016/S0144-8617(01)00314-9.
- Li, K.Y., P. Lai, S. Lu, Y.T. Fang, & H.H. Chen. 2008. Optimization of Acid Hydrolysis Conditions for Feruloylated Oligosaccharides from Rice Bran through Response Surface Methodolgy. *Journal of Agricultural and Food Chemistry* 56 (19): 8975–8978. doi:10.1021/jf801736z.
- Li, M., Z. Jia, G. Wan, S. Wang, & D. Min. 2020. Enhancing Isolation of P-Coumaric and Ferulic Acids from Sugarcane Bagasse by Sequential Hydrolysis. *Chemical Papers* 74 (2): 499–507. doi:10.1007/s11696-019-00890-y.
- Li, Y., G.K. Skouroumounis, G.M. Elsey, & D.K. Taylor. 2011. Microwave-Assistance Provides Very Rapid and Efficient Extraction of Grape Seed Polyphenols. *Food Chemistry* 129 (2): 570–576. doi:10.1016/j.foodchem.2011.04.068.

- Lin, Q., S. Ou, & Q. Wen. 2014. In Vitro Antioxidant Activity of Feruloyl Arabinose Isolated from Maize Bran by Acid Hydrolysis. *Journal of Food Science and Technology* 51 (7): 1356–1362. doi:10.1007/s13197-012-0643-x.
- Lin, Z., J. Fischer, & L. Wicker. 2016. Intermolecular Binding of Blueberry Pectin-Rich Fractions and Anthocyanin. *Food Chemistry* 194 (March): 986–993. doi:10.1016/j.foodchem.2015.08.113.
- Lohani, U.C., K. Muthukumarappan, & G.H. Meletharayil. 2016. "Application of Hydrodynamic Cavitation to Improve Antioxidant Activity in Sorghum Flour and Apple Pomace." Food and Bioproducts Processing 100 (October): 335–343. doi:10.1016/j.fbp.2016.08.005.
- Luengo, E., I. Álvarez, & J. Raso. 2013. Improving the Pressing Extraction of Polyphenols of Orange Peel by Pulsed Electric Fields. *Innovative Food Science & Emerging Technologies* 17 (January): 79–84. doi:10.1016/j.ifset.2012.10.005.
- Magistrelli, D., R. Zanchi, L. Malagutti, G. Galassi, E. Canzi, & F. Rosi. 2016. "Effects of Cocoa Husk Feeding on the Composition of Swine Intestinal Microbiota." *Journal of Agricultural and Food Chemistry* 64 (10): 2046–2052. doi:10.1021/acs.jafc.5b05732.
- Mancini, G., S. Papirio, P.N.L. Lens, & G. Esposito. 2016. "Effect of N-Methylmorpholine- N-Oxide Pretreatment on Biogas Production from Rice Straw, Cocoa Shell, and Hazelnut Skin." *Environmental Engineering Science* 33 (11): 843–850. doi:10.1089/ees.2016.0138.
- Mandal, S.M., D. Chakraborty, & S. Dey. 2010. "Phenolic Acids Act as Signaling Molecules in Plant-Microbe Symbioses." *Plant Signaling & Behavior* 5 (4): 359–368. doi:10.4161/psb.5.4.10871.
- Marseglia, A., L. Dellafiora, B. Prandi, V. Lolli, S. Sforza, P. Cozzini, T. Tedeschi, G. Galaverna, & A. Caligiani. 2019. Simulated Gastrointestinal Digestion of Cocoa: Detection of Resistant Peptides and In Silico/In Vitro Prediction of Their Ace Inhibitory Activity. *Nutrients* 11 (5): 985. doi:10.3390/nu11050985.
- Marseglia, A., S. Sforza, A. Faccini, M. Bencivenni, G. Palla, & A. Caligiani. 2014. "Extraction, Identification and Semi-Quantification of Oligopeptides in Cocoa Beans." *Food Research International* 63 (September): 382–389. doi:10.1016/j.foodres.2014.03.046.
- Martín, M.A., & S. Ramos. 2017. "Health Beneficial Effects of Cocoa Phenolic Compounds: A Mini-Review." Current Opinion in Food Science 14 (April): 20–25. doi:10.1016/j.cofs.2016.12.002.
- Martín-Cabrejas, M.A., C. Valiente, R.M. Esteban, E. Mollá, & K. Waldron. 1994. Cocoa Hull: A Potential Source of Dietary Fibre. *Journal of the Science of Food and Agriculture* 66 (3): 307–311. doi:10.1002/jsfa.2740660307.
- Martínez, R., P. Torres, M.A. Meneses, J.G. Figueroa, J.A. Pérez-Álvarez, & M. Viuda-Martos. 2012. Chemical, Technological and in Vitro Antioxidant Properties of Cocoa (Theobroma Cacao L.) Co-Products. *Food Research International* 49 (1): 39–45. doi:10.1016/j.foodres.2012.08.005.
- Martínez-Abad, A., A.C. Ruthes, & F. Vilaplana. 2016. Enzymatic-Assisted Extraction and Modification of Lignocellulosic Plant Polysaccharides for Packaging Applications. *Journal of Applied Polymer Science* 133 (2): n/a-n/a. doi:10.1002/app.42523.
- Martínez-Cervera, S., A. Salvador, B. Muguerza, L. Moulay, & S.M. Fiszman. 2011. Cocoa Fibre and Its Application as a Fat Replacer in Chocolate Muffins. *LWT - Food Science* and Technology 44 (3): 729–736. doi:10.1016/j.lwt.2010.06.035.

- Martorell, P., E. Bataller, S. Llopis, N. Gonzalez, B. Álvarez, F. Montón, P. Ortiz, D. Ramón, & S. Genovés. 2013. "A Cocoa Peptide Protects Caenorhabditis Elegans from Oxidative Stress and β-Amyloid Peptide Toxicity." Edited by Prasun Datta. *PLoS ONE* 8 (5): e63283. doi:10.1371/journal.pone.0063283.
- Mateos-Aparicio, I., A. Redondo-Cuenca, & M.J. Villanueva-Suárez. 2010. Isolation and Characterisation of Cell Wall Polysaccharides from Legume By-Products: Okara (Soymilk Residue), Pea Pod and Broad Bean Pod. *Food Chemistry* 122 (1): 339–345. doi:10.1016/j.foodchem.2010.02.042.
- Mazzutti, S., L.G.G Rodrigues, N. Mezzomo, V. Venturi, & S.R.S. Ferreira. 2018. "Integrated Green-Based Processes Using Supercritical CO2 and Pressurized Ethanol Applied to Recover Antioxidant Compouds from Cocoa (Theobroma Cacao) Bean Hulls." *The Journal of Supercritical Fluids* 135 (May): 52–59. doi:10.1016/j.supflu.2017.12.039.
- Md Yusof, A., S. Abd Gani, U. Zaidan, M. Halmi, & B. Zainudin. 2019. "Optimization of an Ultrasound-Assisted Extraction Condition for Flavonoid Compounds from Cocoa Shells (Theobroma Cacao) Using Response Surface Methodology." *Molecules* 24 (4): 711. doi:10.3390/molecules24040711.
- Medina-Meza, I.G., & G.V. Barbosa-Cánovas. 2015. Assisted Extraction of Bioactive Compounds from Plum and Grape Peels by Ultrasonics and Pulsed Electric Fields. *Journal of Food Engineering* 166 (December): 268–275. doi:10.1016/j.jfoodeng.2015.06.012.
- Mellinas, A.C., A. Jiménez, & M.C. Garrigós. 2020. Optimization of Microwave-Assisted Extraction of Cocoa Bean Shell Waste and Evaluation of Its Antioxidant, Physicochemical and Functional Properties. *LWT* 127 (June): 109361. doi:10.1016/j.lwt.2020.109361.
- Mioso, R., F.J. Toledo Marante, & I. Herrera Bravo de Laguna. 2015. Penicillium Roqueforti: A Multifunctional Cell Factory of High Value-Added Molecules. *Journal of Applied Microbiology* 118 (4): 781–791. doi:10.1111/jam.12706.
- Mirzadeh, M., M.R. Arianejad, & L. Khedmat. 2020. Antioxidant, Antiradical, and Antimicrobial Activities of Polysaccharides Obtained by Microwave-Assisted Extraction Method: A Review. *Carbohydrate Polymers* 229 (February): 115421. doi:10.1016/j.carbpol.2019.115421.
- Motamayor, J. C., Lachenaud, P., da Silva e Mota, J. W., Loor, R., Kuhn, D. N., Brown, J. S., & Schnell, R. J. (2008). Geographic and Genetic Population Differentiation of the Amazonian Chocolate Tree (Theobroma cacao L). *PLoS ONE*, 3(10), e3311. https://doi.org/10.1371/journal.pone.0003311
- Müller-Maatsch, J., M. Bencivenni, A. Caligiani, T. Tedeschi, G. Bruggeman, M. Bosch, J. Petrusan, B. Van Droogenbroeck, K. Elst, & S. Sforza. 2016. Pectin Content and Composition from Different Food Waste Streams. *Food Chemistry* 201 (June): 37–45. doi:10.1016/j.foodchem.2016.01.012.
- Muñoz-Almagro, N., L. Valadez-Carmona, J.A. Mendiola, E. Ibáñez, & M. Villamiel. 2019.
 Structural Characterisation of Pectin Obtained from Cacao Pod Husk. Comparison of Conventional and Subcritical Water Extraction. *Carbohydrate Polymers* 217 (August): 69–78. doi:10.1016/j.carbpol.2019.04.040.
- Mussatto, S.I., G. Dragone, & I.C. Roberto. 2007. Ferulic and P-Coumaric Acids Extraction by Alkaline Hydrolysis of Brewer's Spent Grain. *Industrial Crops and Products* 25 (2): 231–237. doi:10.1016/j.indcrop.2006.11.001.

- Mussatto, S.I., E.M.S. Machado, S. Martins, & J.A. Teixeira. 2011. Production, Composition, and Application of Coffee and Its Industrial Residues. *Food and Bioprocess Technology* 4 (5): 661–672. doi:10.1007/s11947-011-0565-z.
- Mustafa, A., & C. Turner. 2011. "Pressurized Liquid Extraction as a Green Approach in Food and Herbal Plants Extraction: A Review." *Analytica Chimica Acta* 703 (1): 8–18. doi:10.1016/j.aca.2011.07.018.
- Noh, J., J. Kim, J.S. Kim, Y.S. Chung, S.T. Chang, & J. Park. 2018. "Microencapsulation by Pectin for Multi-Components Carriers Bearing Both Hydrophobic and Hydrophilic Active Agents." *Carbohydrate Polymers* 182 (February): 172–179. doi:10.1016/j.carbpol.2017.11.026.
- Nsor-Atindana, J., F. Zhong, K.J. Mothibe, M.L. Bangoura, & C. Lagnika. 2012. "Quantification of Total Polyphenolic Content and Antimicrobial Activity of Cocoa (Theobroma Cacao L.) Bean Shells." *Pakistan Journal of Nutrition* 11 (7): 672–677. doi:10.3923/pjn.2012.672.677.
- Ohta, T., S. Yamasaki, Y. Egashira, & H. Sanada. 1994. Antioxidative Activity of Corn Bran Hemicellulose Fragments. *Journal of Agricultural and Food Chemistry* 42 (3): 653–656. doi:10.1021/jf00039a010.
- Okiyama, D.C.G., S.L.B. Navarro, & C.E.C Rodrigues. 2017. "Cocoa Shell and Its Compounds: Applications in the Food Industry." *Trends in Food Science & Technology* 63 (May): 103–112. doi:10.1016/j.tifs.2017.03.007.
- Okiyama, D.C.G., I.D. Soares, M.S. Cuevas, E.J. Crevelin, L.A.B. Moraes, M.P. Melo, A.L. Oliveira, & C.E.C Rodrigues. 2018. Pressurized Liquid Extraction of Flavanols and Alkaloids from Cocoa Bean Shell Using Ethanol as Solvent. *Food Research International* 114 (December): 20–29. doi:10.1016/j.foodres.2018.07.055.
- Olabisi., A.I., A.N. Adam, & O.M. Okechukwu. 2016. "Development and Assessment of Composite Brake Pad Using Pulverized Cocoa Beans Shells Filler." *International Journal of Materials Science and Applications* 5 (2): 66. doi:10.11648/j.ijmsa.20160502.16.
- Olubamiwa, O., S.M. Ikyo, B.A. Adebowale, A.B. Omojola, & R.A. Hamzat. 2006. "Effect of Boiling Time on the Utilization of Cocoa Bean Shell in Laying Hen Feeds." *International Journal of Poultry Science* 5 (12): 1137–1139. doi:10.3923/ijps.2006.1137.1139.
- Olumide, M., A.O. Akinsoyinu, & R. A. Hamzat. 2014. "Egg Quality Characteristics of Layers Fed Raw, Fermented, and Enzyme-Treated Cocoa Bean Shell-Based Diets." In , 15:289– 296. Pacific Journal of Science and Technology.
- Oracz, J., E. Nebesny, & D. Żyżelewicz. 2014. Effect of Roasting Conditions on the Fat, Tocopherol, and Phytosterol Content and Antioxidant Capacity of the Lipid Fraction from Cocoa Beans of Different Theobroma Cacao L. Cultivars: Roast Effects on the Bioactive Compounds of Cocoa Beans. *European Journal of Lipid Science and Technology* 116 (8): 1002–1014. doi:10.1002/ejlt.201300474.
- Ou, J., J. Huang, Y. Song, S. Yao, X. Peng, M. Wang, & S. Ou. 2016. Feruloylated Oligosaccharides from Maize Bran Modulated the Gut Microbiota in Rats. *Plant Foods* for Human Nutrition 71 (2): 123–128. doi:10.1007/s11130-016-0547-4.
- Ou, J., & Z. Sun. 2014. Feruloylated Oligosaccharides: Structure, Metabolism and Function. *Journal of Functional Foods* 7 (March): 90–100. doi:10.1016/j.jff.2013.09.028.

- Ou, S., & K. Kwok. 2004. Ferulic Acid: Pharmaceutical Functions, Preparation and Applications in Foods. *Journal of the Science of Food and Agriculture* 84 (11): 1261–1269. doi:10.1002/jsfa.1873.
- Öztürk, E., & G. Ova. 2018. "Evaluation of Cocoa Bean Hulls as a Fat Replacer On Functional Cake Production." *Turkish Journal of Agriculture - Food Science and Technology* 6 (8): 1043. doi:10.24925/turjaf.v6i8.1043-1050.1934.
- Padayachee, A., G. Netzel, M. Netzel, L. Day, D. Mikkelsen, & M.J. Gidley. 2013. Lack of Release of Bound Anthocyanins and Phenolic Acids from Carrot Plant Cell Walls and Model Composites during Simulated Gastric and Small Intestinal Digestion. *Food & Function* 4 (6): 906. doi:10.1039/c3fo60091b.
- Pan, G.X., J.L. Bolton, & G.J. Leary. 1998. Determination of Ferulic and P-Coumaric Acids in Wheat Straw and the Amounts Released by Mild Acid and Alkaline Peroxide Treatment. *Journal of Agricultural and Food Chemistry* 46 (12): 5283–5288. doi:10.1021/jf980608f.
- Panak Balentić, J., D. Ačkar, S. Jokić, A. Jozinović, J. Babić, B. Miličević, D. Šubarić, and N. Pavlović. 2018. Cocoa Shell: A By-Product with Great Potential for Wide Application. *Molecules* 23 (6): 1404. doi:10.3390/molecules23061404.
- Papadopoulou, E.L., U.C. Paul, T.N. Tran, G. Suarato, L. Ceseracciu, S. Marras, R. d'Arcy, & A. Athanassiou. 2019. "Sustainable Active Food Packaging from Poly(Lactic Acid) and Cocoa Bean Shells." ACS Applied Materials & Interfaces 11 (34): 31317–31327. doi:10.1021/acsami.9b09755.
- Papillo, V.A., M. Locatelli, F. Travaglia, M. Bordiga, C. Garino, J.D. Coïsson, & M. Arlorio. 2019. "Cocoa Hulls Polyphenols Stabilized by Microencapsulation as Functional Ingredient for Bakery Applications." *Food Research International* 115 (January): 511– 518. doi:10.1016/j.foodres.2018.10.004.
- Pavlović, M.D., A.V. Buntić, S.S. Šiler-Marinković, & S.I. Dimitrijević-Branković. 2013. Ethanol Influenced Fast Microwave-Assisted Extraction for Natural Antioxidants Obtaining from Spent Filter Coffee. *Separation and Purification Technology* 118 (October): 503–510. doi:10.1016/j.seppur.2013.07.035.
- Pavlović, N., M. Jakovljević, M. Miškulin, M. Molnar, D. Ačkar, & S. Jokić. 2019. "Green Extraction Techniques of Bioactive Components from Cocoa Shell." *Croatian Journal of Food Science and Technology* 11 (1): 11–20. doi:10.17508/CJFST.2019.11.1.02.
- Pèrez Sira, E., ed. 2015. *Chocolate: Cocoa Byproducts Technology, Rheology, Styling, and Nutrition*. Food Science and Technology. New York: Nova Publishers.
- Pérez-Cadenas, M., M. Plaza-Recobert, G. Trautwein, & J. Alcañiz-Monge. 2018. "Development of Tailored Mesoporosity in Carbonised Cocoa Bean Husk." *Microporous and Mesoporous Materials* 256 (January): 128–139. doi:10.1016/j.micromeso.2017.08.005.
- Plaza-Recobert, M., G. Trautwein, M. Pérez-Cadenas, & J. Alcañiz-Monge. 2017. "Preparation of Binderless Activated Carbon Monoliths from Cocoa Bean Husk." *Microporous and Mesoporous Materials* 243 (May): 28–38. doi:10.1016/j.micromeso.2017.02.015.
- Poojary, M.M., S. Roohinejad, F.J. Barba, M. Koubaa, E. Puértolas, A.R. Jambrak, R. Greiner, & I. Oey. 2017. Application of Pulsed Electric Field Treatment for Food Waste Recovery Operations. In *Handbook of Electroporation*, edited by D. Miklavčič, 2573–2590. Cham: Springer International Publishing. doi:10.1007/978-3-319-32886-7 185.
- Preece, K.E., M. Hooshyar, A.J. Krijgsman, P.J. Fryer, & N.J. Zuidam. 2017. "Intensification of Protein Extraction from Soybean Processing Materials Using Hydrodynamic Cavitation."

Innovative Food Science & Emerging Technologies 41 (June): 47–55. doi:10.1016/j.ifset.2017.01.002.

- Preza, A.M., M.E. Jaramillo, A.M. Puebla, J.C. Mateos, R. Hernández, & E. Lugo. 2010. Antitumor Activity against Murine Lymphoma L5178Y Model of Proteins from Cacao (Theobroma Cacao L.) Seeds in Relation with in Vitro Antioxidant Activity. *BMC Complementary and Alternative Medicine* 10 (1): 61. doi:10.1186/1472-6882-10-61.
- Proestos, C., & M. Komaitis. 2008. Application of Microwave-Assisted Extraction to the Fast Extraction of Plant Phenolic Compounds. *LWT - Food Science and Technology* 41 (4): 652–659. doi:10.1016/j.lwt.2007.04.013.
- Puértolas, E., & F.J. Barba. 2016. Electrotechnologies Applied to Valorization of By-Products from Food Industry: Main Findings, Energy and Economic Cost of Their Industrialization. *Food and Bioproducts Processing* 100 (October): 172–184. doi:10.1016/j.fbp.2016.06.020.
- Puglia, D., F. Dominici, M. Badalotti, C. Santulli, & J.M. Kenny. 2016. "Tensile, Thermal and Morphological Characterization of Cocoa Bean Shells (CBS)/Polycaprolactone-Based Composites." *Journal of Renewable Materials* 4 (3): 199–205. doi:10.7569/JRM.2016.634102.
- Pushparaj, F.S., & A. Urooj. 2011. Influence of Processing on Dietary Fiber, Tannin and in Vitro Protein Digestibility of Pearl Millet. *Food and Nutrition Sciences* 02 (08): 895–900. doi:10.4236/fns.2011.28122.
- Quijano-Aviles, M., G. Franco-Agurto, K. Surez-Quirumbay, A. Barragan-Lucas, & P.
 Manzano-Santana. 2016. "Linear Programming Formulation of a Dairy Drink Made of Cocoa, Coffee and Orange By-Products." *Emirates Journal of Food and Agriculture* 28 (8): 554. doi:10.9755/ejfa.2015-09-750.
- Ralet, M., J. Thibault, C.B. Faulds, & G. Williamson. 1994. Isolation and Purification of Feruloylated Oligosaccharides from Cell Walls of Sugar-Beet Pulp. *Carbohydrate Research* 263 (2): 227–241. doi:10.1016/0008-6215(94)00175-8.
- Redgwell, R.J., & C.E. Hansen. 2000. Isolation and Characterisation of Cell Wall Polysaccharides from Cocoa (Theobroma Cacao L.) Beans. *Planta* 210 (5): 823–830. doi:10.1007/s004250050685.
- Redgwell, R.J., V. Trovato, S. Merinat, D. Curti, S. Hediger, & A. Manez. 2003. Dietary Fibre in Cocoa Shell: Characterisation of Component Polysaccharides. *Food Chemistry* 81 (1): 103–112. doi:10.1016/S0308-8146(02)00385-0.
- Reisinger, M., Ö. Tirpanalan, F. Huber, W. Kneifel, & S. Novalin. 2014. Investigations on a Wheat Bran Biorefinery Involving Organosolv Fractionation and Enzymatic Treatment. *Bioresource Technology* 170 (October): 53–61. doi:10.1016/j.biortech.2014.07.068.
- Rinaldi, L., Z. Wu, S. Giovando, M. Bracco, D. Crudo, V. Bosco, & G. Cravotto. 2017.
 "Oxidative Polymerization of Waste Cooking Oil with Air under Hydrodynamic Cavitation." *Green Processing and Synthesis* 6 (4). doi:10.1515/gps-2016-0142.
- Rinaldi, M., P. Littardi, M. Paciulli, A. Caligiani, & E. Chiavaro. 2020. "Effect of Cocoa Bean Shells Granulometries on Qualitative Properties of Gluten-Free Bread during Storage." *European Food Research and Technology* 246 (8): 1583–1590. doi:10.1007/s00217-020-03513-z.
- Rodriguez-Jasso, R.M., S.I. Mussatto, L. Pastrana, C.N. Aguilar, & J.A. Teixeira. 2011. Microwave-Assisted Extraction of Sulfated Polysaccharides (Fucoidan) from Brown Seaweed. *Carbohydrate Polymers* 86 (3): 1137–1144. doi:10.1016/j.carbpol.2011.06.006.

Roelofsen, P.A. 1958. Fermentation, Drying, and Storage of Cacao Beans. In Advances in Food Research, 8:225–296. Elsevier. doi:10.1016/S0065-2628(08)60021-X.

- Rojo-Poveda, O., L. Barbosa-Pereira, L. Mateus-Reguengo, M. Bertolino, C. Stévigny, & G. Zeppa. 2019. "Effects of Particle Size and Extraction Methods on Cocoa Bean Shell Functional Beverage." *Nutrients* 11 (4): 867. doi:10.3390/nu11040867.
- Rojo-Poveda, O., L. Barbosa-Pereira, G. Zeppa, & C. Stévigny. 2020. Cocoa Bean Shell—A By-Product with Nutritional Properties and Biofunctional Potential. *Nutrients* 12 (4): 1123. doi:10.3390/nu12041123.

Romancyzk, L.J., & C. McClelland. 2004. "Extraction of Sterols from Cocoa Hulls."

- Rossin, D., L. Barbosa-Pereira, N. Iaia, G. Testa, B. Sottero, G. Poli, G. Zeppa, & F. Biasi. 2019. A Dietary Mixture of Oxysterols Induces In Vitro Intestinal Inflammation through TLR2/4 Activation: The Protective Effect of Cocoa Bean Shells. *Antioxidants* 8 (6): 151. doi:10.3390/antiox8060151.
- Ryan, C.M., W. Khoo, A.C. Stewart, S.F. O'Keefe, J.D. Lambert, & A.P. Neilson. 2017. Flavanol Concentrations Do Not Predict Dipeptidyl Peptidase-IV Inhibitory Activities of Four Cocoas with Different Processing Histories. *Food & Function* 8 (2): 746–756. doi:10.1039/C6FO01730D.
- Sakai, S., H. Kawamata, T. Kogure, N. Mantani, K. Terasawa, M. Umatake, & H. Ochiai. 1999. Inhibitory Effect of Ferulic Acid and Isoferulic Acid on the Production of Macrophage Inflammatory Protein-2 in Response to Respiratory Syncytial Virus Infection in RAW264.7 Cells. *Mediators of Inflammation* 8 (3): 173–175. doi:10.1080/09629359990513.
- Sánchez, D., L. Moulay, B. Muguerza, M. Quiñones, M. Miguel, & A. Aleixandre. 2010. Effect of a Soluble Cocoa Fiber-Enriched Diet in Zucker Fatty Rats. *Journal of Medicinal Food* 13 (3): 621–628. doi:10.1089/jmf.2009.0127.
- Sánchez-Machado, D. I., J. López-Hernández, P. Paseiro-Losada, & J. López-Cervantes. 2004. An HPLC Method for the Quantification of Sterols in Edible Seaweeds. *Biomedical Chromatography* 18 (3): 183–190. doi:10.1002/bmc.316.
- Santos, T.C., I.S. Cavalcanti, R.C.F. Bonomo, N.B. Santana, & M. Franco. 2011. "Optimization of Productions of Cellulolytic Enzymes by Aspergillus Niger Using Residue of Mango a Substrate." *Ciência Rural* 41 (12): 2210–2216. doi:10.1590/S0103-84782011005000145.
- Santos, T.C., A.R. de Brito, R.C.F. Bonomo, A.J.V. Pires, E. Aguiar-Oliveira, T.P. Silva, & M. Franco. 2017. "Statistical Optimization of Culture Conditions and Characterization for Ligninolytic Enzymes Produced from Rhizopus Sp. Using Prickly Palm Cactus Husk." *Chemical Engineering Communications* 204 (1): 55–63. doi:10.1080/00986445.2016.1230851.
- Sarmadi, B., F. Aminuddin, M. Hamid, N. Saari, A. Abdul-Hamid, & A. Ismail. 2012. "Hypoglycemic Effects of Cocoa (Theobroma Cacao L.) Autolysates." *Food Chemistry* 134 (2): 905–911. doi:10.1016/j.foodchem.2012.02.202.
- Sarmadi, B., A. Ismail, & M. Hamid. 2011. "Antioxidant and Angiotensin Converting Enzyme (ACE) Inhibitory Activities of Cocoa (Theobroma Cacao L.) Autolysates." *Food Research International* 44 (1): 290–296. doi:10.1016/j.foodres.2010.10.017.
- Saulnier, L., J. Vigouroux, & J. Thibault. 1995. Isolation and Partial Characterization of Feruloylated Oligosaccharides from Maize Bran. *Carbohydrate Research* 272 (2): 241– 253. doi:10.1016/0008-6215(95)00053-V.

- Schwan, R.F., & A.E. Wheals. 2004. The Microbiology of Cocoa Fermentation and Its Role in Chocolate Quality. *Critical Reviews in Food Science and Nutrition* 44 (4): 205–221. doi:10.1080/10408690490464104.
- Shen, W.Z., X.X. Wen, S.C Cao, S.Y. Ou, Y.H. Wu, & G.S. Liu. 2012. Pilot Study on the Antioxidation Activity in Vitro and Mechanism of Feruloyl Oligosaccharides Extracted from Maize. Acta Nutrimenta Sinica 34: 483–487.
- Silva, F.A.M., F. Borges, C. Guimarães, J.L.F.C. Lima, C. Matos, & S. Reis. 2000. "Phenolic Acids and Derivatives: Studies on the Relationship among Structure, Radical Scavenging Activity, and Physicochemical Parameters." *Journal of Agricultural and Food Chemistry* 48 (6): 2122–2126. doi:10.1021/jf9913110.
- Silván, J.M., S.H. Assar, C. Srey, M. Dolores del Castillo, & J.M. Ames. 2011. Control of the Maillard Reaction by Ferulic Acid. *Food Chemistry* 128 (1): 208–213. doi:10.1016/j.foodchem.2011.03.047.
- Simonsen, H.T., M.S. Nielsen, N.J. Christensen, U. Christensen, Thomas V. La Cour, M.S. Motawia, B.P.M. Jespersen, S.B. Engelsen, & B.L. Møller. 2009. Molecular Interactions between Barley and Oat β-Glucans and Phenolic Derivatives. *Journal of Agricultural and Food Chemistry* 57 (5): 2056–2064. doi:10.1021/jf802057v.
- Snelders, J., Olaerts, H., Dornez, E., Van de Wiele, T., Aura, A. M., Vanhaecke, L., Delcour, J. A., & Courtin, C. M. (2014). Structural features and feruloylation modulate the fermentability and evolution of antioxidant properties of arabinoxylanoligosaccharides during in vitro fermentation by human gut derived microbiota. *Journal of Functional Foods*, 10, 1–12. https://doi.org/10.1016/j.jff.2014.05.011
- Soares, I.D., D.C.G. Okiyama, & C.E.C. Rodrigues. 2020. Simultaneous Green Extraction of Fat and Bioactive Compounds of Cocoa Shell and Protein Fraction Functionalities Evaluation. *Food Research International* 137 (November): 109622. doi:10.1016/j.foodres.2020.109622.
- Soeharsono, M.A, & A. Cahyono. 2017. "The Use of Cocoa Bean Waste as a Supplement in Male Bali Cattle Feeding." In *Proceedings of International Seminar on Livestock Production and Veterinary Technology*. Indonesian Center for Animal Research and Development (ICARD). doi:10.14334/Proc.Intsem.LPVT-2016-p.45-49.
- Sotelo, A., & R.G. Alvarez. 1991. "Chemical Composition of Wild Theobroma Species and Their Comparison to the Cacao Bean." *Journal of Agricultural and Food Chemistry* 39 (11): 1940–1943. doi:10.1021/jf00011a009.
- Spadoni Andreani, E., & S. Karboune. 2020. Comparison of Enzymatic and Microwave-assisted Alkaline Extraction Approaches for the Generation of Oligosaccharides from American Cranberry (Vaccinium Macrocarpon) Pomace. *Journal of Food Science* 85 (8): 2443– 2451. doi:10.1111/1750-3841.15352.
- Spadoni Andreani, E., M. Li, J. Ronholm, & S. Karboune. 2021. "Feruloylation of Polysaccharides from Cranberry and Characterization of Their Prebiotic Properties." *Food Bioscience* 42 (August): 101071. doi:10.1016/j.fbio.2021.101071.
- Srogi, K. 2006. A Review: Application of Microwave Techniques for Environmental Analytical Chemistry. *Analytical Letters* 39 (7): 1261–1288. doi:10.1080/00032710600666289.
- Stanley, T.H., C.B. Van Buiten, S.A. Baker, R.J. Elias, R.C. Anantheswaran, & J.D. Lambert. 2018. Impact of Roasting on the Flavan-3-Ol Composition, Sensory-Related Chemistry, and in Vitro Pancreatic Lipase Inhibitory Activity of Cocoa Beans. *Food Chemistry* 255 (July): 414–420. doi:10.1016/j.foodchem.2018.02.036.

- Stojadinovic, M., J. Radosavljevic, J. Ognjenovic, J. Vesic, I. Prodic, D. Stanic-Vucinic, & T. Cirkovic Velickovic. 2013. Binding Affinity between Dietary Polyphenols and β-Lactoglobulin Negatively Correlates with the Protein Susceptibility to Digestion and Total Antioxidant Activity of Complexes Formed. *Food Chemistry* 136 (3–4): 1263–1271. doi:10.1016/j.foodchem.2012.09.040.
- Świeca, M., U. Gawlik-Dziki, D. Dziki, B. Baraniak, & J. Czyż. 2013. The Influence of Protein– Flavonoid Interactions on Protein Digestibility in Vitro and the Antioxidant Quality of Breads Enriched with Onion Skin. *Food Chemistry* 141 (1): 451–458. doi:10.1016/j.foodchem.2013.03.048.
- Szwajgier, D., J. Pielecki, & Z. Targoński. 2005. The Release of Ferulic Acid and Feruloylated Oligosaccharides During Wort and Beer Production. *Journal of the Institute of Brewing* 111 (4): 372–379. doi:10.1002/j.2050-0416.2005.tb00222.x.
- Takam, B., E. Acayanka, G.Y. Kamgang, M.T. Pedekwang, & S. Laminsi. 2017. "Enhancement of Sorption Capacity of Cocoa Shell Biomass Modified with Non-Thermal Plasma for Removal of Both Cationic and Anionic Dyes from Aqueous Solution." *Environmental Science and Pollution Research* 24 (20): 16958–16970. doi:10.1007/s11356-017-9328-3.
- Tamayo-Cabezas, J., & S. Karboune. 2020. "Optimizing Immobilization and Stabilization of Feruloyl Esterase from Humicola Insolens and Its Application for the Feruloylation of Oligosaccharides." *Process Biochemistry* 98 (November): 11–20. doi:10.1016/j.procbio.2020.07.009.
- Tamayo-Cabezas, J., & S. Karboune. 2019. Immobilized Feruloyl Esterase from Humicola Insolens Catalyzes the Synthesis of Feruloylated Oligosaccharides. *Process Biochemistry* 79 (April): 81–90. doi:10.1016/j.procbio.2018.12.013.
- Tenkanen, M., M. Makkonen, M. Perttula, L. Viikari, & A. Teleman. 1997. Action of Trichoderma Reesei Mannanase on Galactoglucomannan in Pine Kraft Pulp. *Journal of Biotechnology* 57 (1–3): 191–204. doi:10.1016/S0168-1656(97)00099-0.
- Thörn, C., H. Gustafsson, & L. Olsson. 2011. "Immobilization of Feruloyl Esterases in Mesoporous Materials Leads to Improved Transesterification Yield." *Journal of Molecular Catalysis B: Enzymatic* 72 (1–2): 57–64. doi:10.1016/j.molcatb.2011.05.002.
- Tomšik, A., B. Pavlić, J. Vladić, M. Cindrić, Pavle Jovanov, M. Sakač, A. Mandić, & S. Vidović. 2017. Subcritical Water Extraction of Wild Garlic (Allium Ursinum L.) and Process Optimization by Response Surface Methodology. *The Journal of Supercritical Fluids* 128 (October): 79–88. doi:10.1016/j.supflu.2017.05.012.
- Topakas, E., C. Vafiadi, H. Stamatis, & P. Christakopoulos. 2005. Sporotrichum Thermophile Type C Feruloyl Esterase (StFaeC): Purification, Characterization, and Its Use for Phenolic Acid (Sugar) Ester Synthesis. *Enzyme and Microbial Technology* 36 (5–6): 729–736. doi:10.1016/j.enzmictec.2004.12.020.
- Torre, P., B. Aliakbarian, B. Rivas, J.M. Domínguez, & A. Converti. 2008. Release of Ferulic Acid from Corn Cobs by Alkaline Hydrolysis. *Biochemical Engineering Journal* 40 (3): 500–506. doi:10.1016/j.bej.2008.02.005.
- Toschi, T.G., V. Cardenia, G. Bonaga, M. Mandrioli, & M.T. Rodriguez-Estrada. 2014. "Coffee Silverskin: Characterization, Possible Uses, and Safety Aspects." *Journal of Agricultural* and Food Chemistry 62 (44): 10836–10844. doi:10.1021/jf503200z.
- Tovar-Pérez, E.G., A. Lugo-Radillo, & A. Aguilera-Aguirre. 2019. "Amaranth Grain as a Potential Source of Biologically Active Peptides: A Review of Their Identification,

Production, Bioactivity, and Characterization." *Food Reviews International* 35 (3): 221–245. doi:10.1080/87559129.2018.1514625.

- Tran, T.N., J.A. Heredia-Guerrero, B.T. Mai, L. Ceseracciu, L. Marini, A. Athanassiou, & I.S. Bayer. 2017. "Bioelastomers Based on Cocoa Shell Waste with Antioxidant Ability." *Advanced Sustainable Systems* 1 (7): 1700002. doi:10.1002/adsu.201700002.
- Utami, R., R. Armunanto, S. Rahardjo, & Supriyanto. 2016. Effects of Cocoa Bean (Theobroma Cacao L.) Fermentation on Phenolic Content, Antioxidant Activity and Functional Group of Cocoa Bean Shell. *Pakistan Journal of Nutrition* 15 (10): 948–953. doi:10.3923/pjn.2016.948.953.
- Vafiadi, C., E. Topakas, E. Bakx, H. Schols, & P. Christakopoulos. 2007. Structural Characterisation by ESI-MS of Feruloylated Arabino-Oligosaccharides Synthesised by Chemoenzymatic Esterification. *Molecules* 12 (7): 1367–1375. doi:10.3390/12071367.
- Vafiadi, C., E. Topakas, & P. Christakopoulos. 2006. Regioselective Esterase-Catalyzed Feruloylation of l-Arabinobiose. *Carbohydrate Research* 341 (12): 1992–1997. doi:10.1016/j.carres.2006.05.022.
- Vardakou, M., P. Katapodis, E. Topakas, D. Kekos, B.J. Macris, & P. Christakopoulos. 2004. "Synergy between Enzymes Involved in the Degradation of Insoluble Wheat Flour Arabinoxylan." *Innovative Food Science & Emerging Technologies* 5 (1): 107–112. doi:10.1016/S1466-8564(03)00044-4.
- Várnai, A., T.H.F. Costa, C.B. Faulds, A.M.F. Milagres, M. Siika-aho, & A. Ferraz. 2014. Effects of Enzymatic Removal of Plant Cell Wall Acylation (Acetylation, p-Coumaroylation, and Feruloylation) on Accessibility of Cellulose and Xylan in Natural (Non-Pretreated) Sugar Cane Fractions. *Biotechnology for Biofuels* 7 (1): 153. doi:10.1186/s13068-014-0153-3.
- Várnai, A., L. Huikko, J. Pere, M. Siika-aho, & L. Viikari. 2011. Synergistic Action of Xylanase and Mannanase Improves the Total Hydrolysis of Softwood. *Bioresource Technology* 102 (19): 9096–9104. doi:10.1016/j.biortech.2011.06.059.
- Vlassara, H., & M.R. Palace. 2002. Diabetes and Advanced Glycation Endproducts. *Journal of Internal Medicine* 251 (2): 87–101. doi:10.1046/j.1365-2796.2002.00932.x.
- Vojvodić, A., D. Komes, I. Vovk, A. Belščak-Cvitanović, & A. Bušić. 2016. Compositional Evaluation of Selected Agro-Industrial Wastes as Valuable Sources for the Recovery of Complex Carbohydrates. *Food Research International* 89 (November): 565–573. doi:10.1016/j.foodres.2016.07.023.
- Wang, B., Wang, LY., Dong, L., Özkan N., Shu-Jun, L., & Zhi-Huai, M. 2009. Rheological Properties of Waxy Maize Starch and Xanthan Gum Mixtures in the Presence of Sucrose. *ScienceDirect* 77 (3): 472–481. doi:https://doi.org/10.1016/j.carbpol.2009.01.017.
- Wang, J., B. Sun, Y. Cao, H. Song, & Y. Tian. 2008. Inhibitory Effect of Wheat Bran Feruloyl Oligosaccharides on Oxidative DNA Damage in Human Lymphocytes. *Food Chemistry* 109 (1): 129–136. doi:10.1016/j.foodchem.2007.12.031.
- Wang, Z., Li, S., Ge, S., & Lin, S. (2020). Review of Distribution, Extraction Methods, and Health Benefits of Bound Phenolics in Food Plants. *Journal of Agricultural and Food Chemistry*, 68(11), 3330–3343. https://doi.org/10.1021/acs.jafc.9b06574
- Watrelot, A.A., C. Le Bourvellec, A. Imberty, & C.M.G.C. Renard. 2013. Interactions between Pectic Compounds and Procyanidins Are Influenced by Methylation Degree and Chain Length. *Biomacromolecules* 14 (3): 709–718. doi:10.1021/bm301796y.
- Wu, J., C. Hsieh, H. Wang, & H. Chen. 2009. Inhibitory Effects of Guava (Psidium Guajava L.) Leaf Extracts and Its Active Compounds on the Glycation Process of Protein. *Food Chemistry* 113 (1): 78–84. doi:10.1016/j.foodchem.2008.07.025.
- Wu, L., S. Lee, & T. Endo. 2013. Effect of Dimethyl Sulfoxide on Ionic Liquid 1-Ethyl-3-Methylimidazolium Acetate Pretreatment of Eucalyptus Wood for Enzymatic Hydrolysis. *Bioresource Technology* 140 (July): 90–96. doi:10.1016/j.biortech.2013.04.072.
- Xia, E., B. Wang, X. Xu, L. Zhu, Y. Song, & H. Li. 2011. Microwave-Assisted Extraction of Oleanolic Acid and Ursolic Acid from Ligustrum Lucidum Ait. *International Journal of Molecular Sciences* 12 (8): 5319–5329. doi:10.3390/jjms12085319.
- Xu, F., R. Sun, J. Sun, C. Liu, B. He, & J. Fan. 2005. Determination of Cell Wall Ferulic and P-Coumaric Acids in Sugarcane Bagasse. *Analytica Chimica Acta* 552 (1–2): 207–217. doi:10.1016/j.aca.2005.07.037.
- Yajima, A., H. Owada, S. Kobayashi, N. Komatsu, K. Takehara, M. Ito, K. Matsuda, et al. 2016. "Cacao Bean Husk: An Applicable Bedding Material in Dairy Free-Stall Barns." *Asian-Australasian Journal of Animal Sciences* 30 (7): 1048–1053. doi:10.5713/ajas.16.0877.
- Yamada, H., & H. Kiyohara. 2007. Immunomodulating Activity of Plant Polysaccharides Structures. *Comprehensive Glycoscience* 4. Elsevier: 663–693. https://ci.nii.ac.jp/naid/10029656425/en/.
- Yao, S. W., Wen, X. X., Huang, R. Q., He, R. R., Ou, S. Y., Shen, W. Z., Huang, C. H., & Peng, X. C. (2014). Protection of Feruloylated Oligosaccharides from Corn Bran against Oxidative Stress in PC 12 Cells. *Journal of Agricultural and Food Chemistry*, 62(3), 668–674. https://doi.org/10.1021/jf404841c
- Yoon, H.J., M.E. Moon, H.S. Park, S.Y. Im, & Y.H. Kim. 2007. Chitosan Oligosaccharide (COS) Inhibits LPS-Induced Inflammatory Effects in RAW 264.7 Macrophage Cells. *Biochemical and Biophysical Research Communications* 358 (3): 954–959. doi:10.1016/j.bbrc.2007.05.042.
- Yuan, X., J. Wang, H. Yao, & F. Chen. 2005. Free Radical-Scavenging Capacity and Inhibitory Activity on Rat Erythrocyte Hemolysis of Feruloyl Oligosaccharides from Wheat Bran Insoluble Dietary Fiber. LWT - Food Science and Technology 38 (8): 877–883. doi:10.1016/j.lwt.2004.09.012.
- Zhang, J., S.Y. Ou, & N. Zhang. 2005. The Scavenging Capacity of Free Radicals by Enzymes-Hydrolyzed Wheat Bran in Vitro. *Acta Nutrimenta Sinica* 27: 25–29.
- Zhu, F. 2018. Interactions between Cell Wall Polysaccharides and Polyphenols. *Critical Reviews in Food Science and Nutrition* 58 (11): 1808–1831. doi:10.1080/10408398.2017.1287659.
- Zia-Ur-Rehman, & W.H. Shah. 2001. Tannin Contents and Protein Digestibility of Black Grams (Vigna Mungo) after Soaking and Cooking. *Plant Foods for Human Nutrition* 56 (3): 265–273. doi:10.1023/A:1011162823925.